STAFF REPORTS

NO. 1126 OCTOBER 2024

REVISED FEBRUARY 2025

Firm Wage Setting and On-the-Job Search Limit Wage-Price Spirals

Justin Bloesch | Seung Joo Lee | Jacob P. Weber

Firm Wage Setting and On-the-Job Search Limit Wage-Price Spirals

Justin Bloesch, Seung Joo Lee, and Jacob P. Weber *Federal Reserve Bank of New York Staff Reports*, no. 1126 October 2024; revised February 2025 https://doi.org/10.59576/sr.1126

Abstract

We argue that if firms set wages and workers search on the job, then pass-through from prices to wages is weak, limiting wage-price spirals. We derive a tractable general equilibrium model with firm wage setting and on-the-job search. Our wage Phillips curve matches U.S. empirical evidence that quits predict nominal wage growth. To isolate pass-through mechanisms, we theoretically examine a shock that raises consumer prices but not the marginal product of labor. In response, real wages fall at both workers' current job and their outside options, limiting firms' incentive to raise wages and preventing propagation into further price increases.

JEL classification: E31, E52 Key words: monopsony, inflation, cost-of-living shocks, on-the-job search

This paper presents preliminary findings and is being distributed to economists and other interested readers solely to stimulate discussion and elicit comments. The views expressed in this paper are those of the author(s) and do not necessarily reflect the position of the Federal Reserve Bank of New York or the Federal Reserve System. Any errors or omissions are the responsibility of the author(s).

To view the authors' disclosure statements, visit https://www.newyorkfed.org/research/staff_reports/sr1126.html.

Weber: Federal Reserve Bank of New York (email: jake.weber@ny.frb.org). Bloesch: Cornell University (email: jb2722@cornell.edu). Lee: University of Oxford (email: seung.lee@sbs.ox.ac.uk). This paper was previously circulated under the title "Do Cost-of-Living Shocks Pass Through to Wages?" The authors thank Daron Acemoglu, Keshav Dogra, Ryan Chahrour, Andrea Ferrero, John Grigsby, Marianna Kudlyak, Arvind Krishnamurthy, Walker Ray, Iván Werning, and seminar participants at the Reserve Bank of Australia, Federal Reserve Bank of New York, West Coast Search and Matching Conference, Oxford, UCL, Chinese University of Hong Kong, Hong Kong University of Science and Technology, and the 2024 SED, CEBRA, and Midwest Macro Annual Meetings for helpful comments and discussion.

1 Introduction

In the economic recovery following the COVID pandemic, economies throughout the world experienced both rapid price inflation and nominal wage growth, raising concerns of a wage-price spiral. While there is substantial evidence that wage costs are passed through into prices,¹ there is little consensus regarding how much, and by what mechanisms, wages respond to higher prices. Mainstream macroeconomic models used to understand the determinants of nominal wage growth assume that unions representing workers set wages and imply that higher prices will pass-through to wages. However, union membership has declined dramatically in many advanced economies, and evidence suggests that wage posting is the most common, if not dominant, method of wage determination in the United States.² In light of this evidence and the recent experience of inflation, we ask: *when firms set both prices and wages, through what mechanisms do workers' wages respond to shocks to higher consumer prices, and how large is this response?*

To answer this, we derive a two-sector macroeconomic environment where workers search on the job, and firms set both prices and wages. We study the response of wages to a "cost-ofliving" shock, i.e., a shock that raises the cost of households' consumption bundle without affecting the marginal product of labor. Formally, we assume that workers consume two goods: a laborintensive services bundle (e.g., haircuts) and an endowment good (e.g., unprepared food or energy). Negative shocks to the quantity of the endowment good raise the price of workers' consumption basket without affecting the marginal product of labor.³ While such shocks are not necessarily representative of all inflationary shocks, this setup allows us to isolate *in general equilibrium* the pass-through of prices to wages that is often described in narratives of how wage-price spirals occur.

Next, we outline the labor block of the model. We assume that workers search on the job, and firms set a common wage for all workers subject to nominal rigidities in the form of standard, convex adjustment costs. When setting wages, a firm considers that paying a higher wage decreases costly turnover: a firm fills costly vacancies more quickly and discourages its workers from quitting by paying a higher wage. With on-the-job search, wages are determined in equilibrium primarily by firms competing for already-employed workers, as the threat of workers quitting into unemployment is low. We analytically derive the wage Phillips curve in this environment, and we show how it can be written as a simple relationship between nominal wage growth and log deviations in the quit rate and the unemployment rate alone. Calibrated to match U.S. data on worker

¹Heise et al. (2022) show that pass-through of wage costs to services prices is high, while pass-through in goodsproducing industries is lower. Harasztosi and Lindner (2019) similarly find nearly complete pass-through of wages to prices in non-tradable industries.

²See Hall and Krueger (2012); Lachowska et al. (2022); Di Addario et al. (2023).

³In our baseline model with Cobb-Douglass preferences, the marginal *revenue* product of labor is also unchanged.

flows, our model's wage Phillips curve states that the quit rate has the most explanatory power for wage growth, while the unemployment rate, the forcing variable in standard sticky-wage models such as Galí (2011), has almost no weight. Estimating this wage Phillips curve in reduced form on U.S. data, we find empirically that quits dominates unemployment in predicting wage growth, providing a validation of the model.

We then consider how a transitory cost-of-living shock to the (endowment) goods sector passes through into wages in the services sector and propagates into generalized inflation. A higher cost of living raises firms' optimal wage only if it affects turnover costs, e.g., by making vacancies harder to fill or workers more likely to quit. In our model, a higher cost of living has no effect on the probability of a vacancy being filled or a worker quitting: while the increase in the price level erodes a worker's real wage, it also erodes both the real wage offered at other firms and the value of consumption obtained in unemployment. This means that inflation induced by the cost-of-living shock will have little effect on a firm's vacancy filling or quit rates, and hence little effect on the optimal wage. Indeed, in our benchmark model with log utility, there is zero pass-through from the cost-of-living shock to wages—unlike in models with neoclassical labor supply or where unions set wages.⁴

Lastly, we consider an extension of the model where increases in the cost of living erode the real wage but do not similarly erode the value of unemployment, motivated by the idea that the value of unemployment lies in increased leisure or home production. While this extension generates some pass-through into wages and an increase in the long-run price level, we show that on-the-job search quantitatively limits this pass-through. Note that if the consumption value of unemployment is insulated from higher consumer prices while real wages are not, a cost-of-living shock lowers the value of taking a job, making unemployed workers harder to recruit and incentivizing firms to set higher wages. Without on-the-job search, this decrease in labor supply meaningfully increases wages, as firms have to pay higher wages to attract unemployed workers. However, with realistic rates of on-the-job search, firm wage setting is dominated by competition for already-employed workers, whose propensity to accept outside job offers is unaffected by the cost-of-living shock. As a consequence, the cost-of-living shock has trivial effects on firms' optimal wage, leaving the cumulative growth in the price level similarly muted. Thus firm wage setting and on-the-job search greatly limit propagation of cost-of-living shocks into generalized inflation.

Our results suggest that in economies like the United States where few workers operate under collective bargaining agreements with cost-of-living adjustments, and where firms' wage setting

⁴More formally, we show that this is because our setup eliminates wealth effects resulting from the shock to the endowment good, which can cause wages to rise in the neoclassical or union wage setting model even in the log utility case where income and substitution effects cancel. Further, because cumulative growth in nominal wages and prices will be the same in the long run, the non-response of wages to a cost-of-living shock also means there is zero propagation from the initial cost-of-living shock to the long-run price level.

decisions reflect competition for already-employed rather than for unemployed workers, the ability for workers to reclaim real wages in response to a supply shock that raises their cost of living is limited. There is thus little scope for supply-shock induced wage-price spirals fueled by workers' ability to command higher nominal wages in response to higher nominal prices.

Recent studies have explored supply shocks and the response of wages in equilibrium. We differ from Lorenzoni and Werning (2023a,b) where workers set wages via unions and Gagliardone and Gertler (2023) where workers bargain and wages are rigid in real terms. Unlike these works and other papers which study oil or other shocks which affect the marginal product of labor, we study a shock which only affects workers' cost of living and focus on understanding whether the pass-through of cost-of-living shocks to wages amplifies inflationary shocks in the modern U.S. economy. Given this focus, we assume there is no *ad hoc* real wage rigidity, which mechanically generates pass-through from cost-of-living changes to wages, noting that there is little evidence to suggest this type of indexation is widely used in the United States at present.⁵ Similarly, while there is a long tradition of tractably incorporating nominal wage rigidity in New Keynesian models by assuming workers are unionized following Erceg et al. (2000), we assume that workers are not unionized, as only 11.3% of U.S. workers were unionized as of 2022 (Shierholz et al., 2023); we also show that doing so is important for our results, as assuming unions set wages does imply pass-through from prices to wages in response to a cost-of-living shock. Our findings accord with empirical results in Bernanke and Blanchard (2024) that "catch-up" of wages to unexpected price increases was close to zero for most advanced economies during the COVID period.

A recent literature explores how workers' beliefs or search behavior respond to inflationary episodes in the presence of nominal wage rigidity (Pilossoph and Ryngaert, 2024; Pilossoph et al., 2023; Afrouzi et al., 2024). An important property of the models in this literature is that the distribution of workers' outside options improves in nominal terms as inflation occurs, either because the distribution of offered real wages is exogenously assumed to be fixed, or because the marginal revenue product of labor exogenously rises with inflation. In our general equilibrium setting with two consumption goods, we can disentangle the roles of higher consumer prices versus a higher marginal revenue product of labor on nominal wage growth. This allows us to isolate the *price-to-wage channel* of wage-price spirals, which is distinct from monetary shocks that affect both the price of consumption goods and the marginal revenue product of labor.

While stylized, our model is consistent with a range of recent microeconomic evidence on how wages are determined. Our model captures the result in Jäger et al. (2020) that wages are insensitive to the flow value of unemployment benefits, even for workers who were hired directly

⁵Evidence on the use of cost-of-living adjustments (COLAs) comes from studies of large union contracts, which now cover only a small share of U.S. employment. Even within unionized workers, the share covered by contracts with COLAs has shrunk dramatically since the 1970s. See Christiano et al. (2016), footnote 4.

from unemployment. This feature arises for two reasons. First, in our model calibrated to U.S. data, the value of unemployment is significantly below the value of employment, so even sizable changes in the flow value of unemployment benefits does not make unemployment a credible outside option. Second, because firms post wages rather than bargain, all workers are paid the same regardless of their previous employment status. This common wage policy is supported by the finding in Di Addario et al. (2023) that workers' prior employer has small effects on workers' current wages in a large majority of occupations, as well as results in Hall and Krueger (2012) and Lachowska et al. (2022) that most workers in the United States face wage posting rather than bargaining. Lastly, our model features finite elasticities of hiring and separation rates with respect to firms' wage policies (i.e., when a firm raises its wages, workers join the firm more quickly and leave the firm more slowly), which has been extensively documented in the monopsony literature (e.g., Datta, 2023; Bassier et al., 2022). We also note that our results for pass-through are qualitatively consistent with survey evidence in Hajdini et al. (2023) who find that U.S. workers believe that the pass-through from aggregate inflation to their income is low.

There is a large macro-labor literature that embeds search frameworks and labor market frictions in macroeconomic models to study implications for the business cycle. de la Barrera i Bardalet (2023) develops a similar model of labor market monopsony with on-the-job search and finds that increased monopsony power flattens the wage Phillips curve. Moscarini and Postel-Vinay (2023) study a model where workers search on the job and bargain when they receive an outside offer, resulting in a wage Phillips curve in which the distribution of wages and misallocation of workers matters for wage growth. Instead, we assume wage-posting and study a simpler setting without misallocation, where workers and firms are homogeneous in their productivity, and study the implications for pass-through from cost-of-living shocks to prices.

While prior research has modeled the relationship between job-to-job mobility and wage growth,⁶ our setting provides a tractability advantage: if firms are ex-ante identical and adjust prices and wages subject to pricing frictions à la Rotemberg (1982), then our model features a symmetric equilibrium with a single wage alongside endogenous worker flows between firms (and unemployment). This single-wage outcome is compatible with workers' on-the-job search due to the presence of idiosyncratic, worker-specific preference shocks over workplaces, so that workers will sometimes choose to switch jobs even when firms offer identical wages. The result is a tractable labor block with job-to-job mobility that can be easily integrated into the standard New Keynesian modeling framework widely used both in the macroeconomics literature (Christiano et al., 2005; Smets and Wouters, 2007) and for policy analysis at central banks. Additionally, our model has

⁶For example, Moscarini and Postel-Vinay (2016a) extend the wage posting model of Burdett and Mortensen (1998) into a dynamic setting, and Birinci et al. (2022) assume a three-party bargaining protocol to determine wages. Faccini and Melosi (2023) study the relationship of job-to-job mobility and price inflation.

proven useful for policymakers' reduced-form empirical analysis of the U.S. wage Phillips curve (Heise et al., 2024a,b; Williams, 2024) by providing a *tractable* structural explanation for the strong correlation between wage growth and certain measures of labor market tightness (namely, quits and vacancies per effective searcher).

Layout Section 2 presents stylized facts from U.S. data, demonstrating the tight correlation between quits and wage inflation that motivates our model's assumptions of wage posting and onthe-job search, as well as a weak relationship between price and wage inflation. Section 3 presents our benchmark New Keynesian model with on-the-job search and firms' wage posting. Section 4 demonstrates that our wage-posting model with on-the-job search implies little scope for passthrough from prices to wages: specifically, Section 4.1 demonstrates this result quantitatively in our model, and also shows that monetary policy shocks cause wages and prices to co-move. Section 4.2 works through an extension of our baseline model, in which our cost-of-living shock makes employment less desirable and causes firms to raise wages, and shows that on-the-job search greatly dampens this channel quantitatively. Section 5 provides an analytic comparison of our model to neoclassical labor supply models and union wage setting models commonly used in the literature, in which changes in prices do pass through to wages. Section 6 concludes.⁷

2 Stylized Facts on the Wage Phillips Curve

Before proceeding to our formal framework, we present two stylized facts about the wage Phillips curve. First, nominal wage growth, measured by the employment cost index, and price inflation are weakly related at high frequencies. Second, nominal wage growth is tightly correlated with the quits rate.

Figure 1 plots the time series of four quarter growth in the employment cost index of wages and salaries for private industries and four quarter growth in the consumer price index, beginning in the fourth quarter of 1990 when the employment cost index data is available. Outside of the post-COVID period, nominal wage growth and price inflation are weakly correlated. Divergence between price inflation and wage growth is particularly visible in 2011 and 2015, when inflation rose and subsequently fell, while nominal wage growth was changing only gradually.

Figure 2 plots the relationship between the four quarter moving average of the quit rate, which is measured as quits per hundred employees from the Job Openings and Labor Turnover Survey (JOLTS), and the four quarter growth in the employment cost index. This figure shows an extension of the result documented by Faberman and Justiniano (2015) that quits are highly related to growth

⁷Online Appendices and Supplementary Appendices provide additional related results.



Figure 1: Wage Growth and Headline Price Inflation

Notes: Between 1990-2019, nominal wage growth and price inflation were weakly correlated. Nominal wage growth and price inflation both surged during the COVID pandemic and recovery.



Figure 2: Wage Growth and Quits

Notes: There is a strong correlation between quits, measured here as the four quarter moving average of quits per hundred employees from the Job Openings and Labor Turnover Survey (JOLTS), and four quarter growth in the employment cost index, wages and salaries for private industries.

in the employment cost index, and is related to results documented by Moscarini and Postel-Vinay (2017) that nominal wage growth is well-predicted by job-to-job transitions.⁸ The fact that a large share of quits are in fact job-to-job transitions motivates the inclusion of on-the-job search in our model; we will show later that including a realistic quantity of on-the-job search (i.e., by calibrating our model to U.S. data) has important implications for the pass-through of cost-of-living shocks to nominal wages.

We formalize these claims about the correlations of these variables using OLS estimates of the empirical wage Phillips curve which include measures of wage inflation, quits, and price inflation over time. We also include the unemployment rate and the unemployment gap (the difference between the unemployment rate and the natural rate of unemployment), which are the traditional labor market indicators for estimating the wage Phillips curve (Galí, 2011).⁹ Formally, in quarter t, letting Q_t denote quits, U_t denote the unemployment rate, and π_t^w and π_t denote quarterly nominal wage and price inflation, respectively, we estimate the following regression:¹⁰

$$\pi_t^w = \beta_0 + \beta_Q \ln Q_t + \beta_U \ln U_t + \beta_\pi \pi_{t-1} + \varepsilon_t.$$
(1)

Table 1 reports the results: the empirical estimate $\hat{\beta}_Q$ is much larger than $\hat{\beta}_U$.¹¹ Indeed, $\hat{\beta}_U$ is not generally significant at conventional levels and is not of the expected sign once we include quits (see Column (3)). These results are robust to the inclusion of the COVID pandemic and recent recovery. In the following sections, we will develop a model that is capable of matching the empirical correlations in equation (1): specifically, we will write down a structural wage Phillips curve of the similar form of (1), where we will have both a large positive value of $\hat{\beta}_Q$ but small (and potentially positive) value of $\hat{\beta}_U$ when including quits and unemployment in the wage Phillips curve.

⁸The empirical measure of quits include various labor market transitions: job-to-job transitions without a period of non-employment, job-to-job transitions with a period of non-employment, and voluntary quits into non-employment. Qiu (2022) shows finds that 3% of workers transition from employment to non-participation each month (most of which appear voluntary) and Elsby et al. (2010) find that only 16% of workers who quit enter a period of unemployment.

⁹For the quits data, from 2010Q3-2023, we use the private sector quit rate JTS1000QUR from the St. Louis Federal Reserve FRED database, aggregated by averaging at the quarterly level. Prior to 2000, we use the quarterly private sector quits rate from Davis et al. (2012). Between 2001Q1 and 2010Q2, we use the average of these two series. Similarly for the employment cost index, we use the employment cost index wages and salaries series for private industry workers ECIWAG from FRED beginning in 2005. Prior to 2001, we use the SIC industry basis of the employment cost index for private industry wages and salaries, series ECS20002I, from the Bureau of Labor Statistics. From 2001-2005, we take the average of these two wage series.

¹⁰We compute the inflation terms as log differences: letting W_t be the nominal wage and P_t be the price level, $\pi_t^w \equiv 100 \times (\ln W_t \quad \ln W_{t-1})$ and $\pi_{t-1} \equiv 100 \times (\ln P_{t-1} \quad \ln P_{t-2})$.

¹¹Column 1 should be interpreted as "for a 100% increase in the unemployment rate (i.e., double the unemployment), there is an .485% point decrease in quarterly gross wage growth". Likewise, Column 2 implies that 100% increase in the quits rate (i.e., double the quits) would result in 0.9737% point increase in quarterly wage growth. The standard deviation of log quits and log unemployment are 0.17 and 0.28, respectively, from 1990Q2-2024Q1.

	(1)	(2)	(3)	(4)	(5)
Variable	ECI	ECI	ECI	ECI	ECI
$\ln U_t$	-0.4850***	-0.0388	0.1375		
	(0.0727)	(0.0986)	(0.1054)		
$\ln Q_t$		0.9737***	1.1175***	0.9815***	0.9995***
		(0.1553)	(0.1875)	(0.1657)	(0.1223)
$\ln U_t \ln U_t^*$				-0.0325	
				(0.0967)	
$\ln U_{t-1}$					-0.0226
					(0.0748)
π_{t-1}	0.1411***	0.0882***	0.0365	0.0880***	0.0877***
	(0.0480)	(0.0331)	(0.0321)	(0.0329)	(0.0328)
Observations	136	136	119	136	136
Sample End	2024Q1	2024Q1	2019Q4	2024Q1	2024Q1
Standard errors in parentheses					

Table 1: Time Series Regression of Wage Growth on Labor Market Variables, 1990Q2-2024Q1

Standard errors in parentheses

*** p<0.01, ** p<0.05, * p<0.1

Notes: Results from a quarterly regression of wage growth measured using the Employment Cost Index (ECI) on unemployment, the quit rate, and lagged price inflation, as specified in equation (1). While Column 1 shows that a regression of wage growth on unemployment alone yields a familiar negative sign, including quits reduces the significance to below conventional levels as seen in Column 2. Columns 3 and 4 demonstrate that this result is robust to dropping the COVID pandemic and recovery, and also to measuring unemployment in log deviations from it's natural rate as estimated by the CBO, $\ln(U_t) - \ln(U_t^*)$. The final column demonstrates that using lagged U_t doesn't alter the results. Standard errors are Newey-West with 4 lags.

Table 1 also reports the coefficient of lagged price inflation on wage growth $\hat{\beta}_{\pi}$. The coefficient on inflation is fairly small, with elasticities all below 0.15. Comparing Columns (1) and (2), we see that the coefficient on price inflation falls when adding quits, suggesting that the unemployment rate is failing to pick up fluctuations in both wage and price inflation that the quit rate absorbs. Column (3) shows that prior to COVID, there was no significant relationship between price inflation and wage growth after conditioning on quits. Including the unemployment gap or lagging unemployment in Columns (4) and (5), respectively, does not meaningfully change the results.

In Section 4.2.2, we will use this wage Phillips curve evidence to compare the empirical wage Phillips curve with our model-implied wage Phillips curve.

3 Model

This section builds a model where firms post wages and workers search on the job, and calibrates that model to U.S. data. In laying out the model, we first describe the problem of a firm posting wages in the presence of recruiting costs and where workers search on-the-job. When deciding whether to raise wages, the firm trades off between a higher wage bill and lower turnover costs. Higher wages lower turnover costs by increasing the probability that the firm recruits a particular searcher, regardless of whether that searcher is already employed or unemployed (the recruiting rate), while also lowering the probability that incumbent workers leave (the separation rate). Because the firm's problem does not depend directly on the price level in partial equilibrium, an increase in workers' cost of living can only affect wages through its effects on these recruiting or separation rates.

We then describe the solution to the worker's problem, which determines firms' recruiting and separation rates. Since a change in the price level affects the real wages offered by all firms proportionally, changes in the price level can relatively improve workers' outside option, and raise wages, only if changes in the price level make unemployment relatively more attractive. If this is the case, this can lead to pass-through from cost of living to wages, as firms must now offer a higher wage to retain the same number of workers as before. However, these considerations are quantitatively small when (i) most workers already vastly prefer a job to unemployment irrelevant when considering whether to accept a new job offer. Moreover, this channel need not exist at all if changes in the price level do not affect the desirability of unemployment, as in our baseline model described below.

Structure There are two goods in the economy: an endowment good X_t and services Y_t . They are combined into an aggregate consumption good, C_t , according to the CES function

$$C_{t} = \left(\alpha_{Y}^{\frac{1}{\eta}}Y_{t}^{\frac{\eta-1}{\eta}} + \alpha_{X}^{\frac{1}{\eta}}X_{t}^{\frac{\eta-1}{\eta}}\right)^{\frac{\eta}{\eta-1}},$$
(2)

with corresponding aggregate price index¹²

$$P_{t} = \left(\alpha_{Y} P_{y,t}^{1 \eta} + \alpha_{X} P_{x,t}^{1 \eta}\right)^{\frac{1}{1 \eta}}.$$
(3)

Workers are hired by firms to produce services Y_t , so that their real wage is determined by the nominal wage offered in that sector divided by the aggregate price level P_t . The total amount of

¹²We assume $\alpha_X + \alpha_Y = 1$ with $\alpha_X > 0$ and $\alpha_Y > 0$ as usual.

endowment good $X_t = 1$ is given, and X_t is flexibly and competitively priced.

Cost-of-Living Shock Our "pure" cost of living shock is a decline in the endowment good X_t which raises its price, $P_{x,t}$, and hence the price level P_t in (3). This is a pure cost-of-living shock in the sense that it raises the cost of living for workers without affecting their marginal products, unlike an oil shock, for example, which may affect both. The point of considering such a shock is not to downplay the role or importance of oil shocks to many modern economies, but to highlight how these shocks propagate and consider whether a "wage-price spiral" amplifies their effects on the price level.

Firm's Wage-Posting Problem We now turn to the determination of the nominal wage. We assume that perfectly-competitive retailers bundle service types j according to a standard Dixit-Stiglitz production function with an associated ideal price index:

$$\begin{split} Y_t &= \left(\int \left(Y_t^j \right)^{\frac{\epsilon}{\epsilon}} dj \right)^{\frac{\epsilon}{\epsilon-1}}, \\ P_{y,t} &= \left(\int \left(P_{y,t}^j \right)^{1-\epsilon} dj \right)^{\frac{1}{1-\epsilon}}, \end{split}$$

yielding product demand for variety *j*:

$$\frac{Y_t^j}{Y_t} = \left(\frac{P_{y,t}^j}{P_{y,t}}\right)^{-\epsilon}.$$
(4)

The firm j produces only with labor according to $Y_t^j = A_t^j N_{jt}$. Firm j sets nominal wages W_{jt} each period, which is assumed to be the same for all workers in the firm, including new hires. Workers separate from firm j with probability $S_t(W_{jt}|\{W_{kt}\}_{k\neq j})$ each period, with $S'_t(W_{jt}|\{W_{kt}\}_{k\neq j}) < 0$: firms retain a higher share of workers each period by paying a higher wage, given other firms' wages. The firm can recruit workers by posting vacancies V_{jt} , and the probability that a vacancy successfully results in a hire is $R_t(W_{jt}|\{W_{kt}\}_{k\neq j})$, with $R'_t(W_{jt}|\{W_{kt}\}_{k\neq j}) > 0$. How retention and separation functions $R_t(W_{jt}|\{W_{kt}\}_{k\neq j})$ and $S_t(W_{jt}|\{W_{kt}\}_{k\neq j})$ depend on wages set by other service firms will be derived after we describe households' and workers' problems in Section 3.2.

The firm pays a convex, per-vacancy hiring cost, $c \left(\frac{V_{jt}}{N_{j,t-1}}\right)^{\chi} W_t$, to post V_t vacancies, where W_t is the aggregate wage, c > 0 and $\chi \ge 0$. Finally, the firm is also subject to price and wage adjustment frictions à la Rotemberg (1982). Given this, each firm j maximizes the present discounted

value of profits, solving

$$\max_{\substack{\{P_{y,t}^{j}\},\{Y_{t}^{j}\},\\\{N_{jt}\},\{W_{jt}\},\{V_{t}^{j}\}}} \sum_{t=0}^{\infty} \left(\frac{1}{1+\rho}\right)^{t} \left(P_{y,t}^{j}Y_{t}^{j} - W_{jt}N_{jt} - c\left(\frac{V_{jt}}{N_{j,t-1}}\right)^{\chi}V_{jt}W_{t} - \frac{\psi}{2}\left(\frac{P_{y,t}^{j}}{P_{y,t-1}^{j}} - 1\right)^{2}Y_{t}^{j}P_{y,t}^{j} \\
- \frac{\psi^{w}}{2}\left(\frac{W_{jt}}{W_{j,t-1}} - 1\right)^{2}W_{jt}N_{jt}\right)$$
(5)

subject to the law of motion for employment

$$N_{jt} = (1 \quad S_t(W_{jt}))N_{j,t-1} + V_{jt}R_t(W_{jt})$$
(6)

and the product demand equation (4). From inspecting equations (5) and (6), we can observe that the service sector firm chooses the wage (and other choice variables) taking as given the choices of other service sector firms (embodied in the price index and aggregate output of the service sector), parameters, and the separation and recruiting rates $S_t(\cdot)$ and $R_t(\cdot)$. Note that since the vacancyposting cost is denominated in labor (i.e., priced by the aggregate wage W_t), the aggregate price level P_t does not appear directly in (5). Thus, in partial equilibrium, the only way that changes in the price level can impact the firm's wage setting decision is through changes in $S_t(\cdot)$ and $R_t(\cdot)$, which will be determined by the solution to workers' optimization problem described in Section 3.2.

3.1 Symmetric Equilibrium

To make this relationship between the separation and recruiting rates and the firm's choice of wage clearer, we derive a wage Phillips curve from the firm's first order conditions, assuming for the moment that a symmetric equilibrium, where all firms offer the same aggregate wage W_t , exists. Under this assumption, the wage Phillips curve expresses nominal wage growth as exclusively a function of aggregate, endogenous labor market variables: vacancies, employment, recruiting and separation rates, and recruiting and separation elasticities, again with no direct role for aggregate price index P_t .

Denote $\varepsilon_{R,W}$ and $\varepsilon_{S,W}$ as the elasticities of the recruiting function $R_t(W_{jt})$ and the separation function $S_t(W_{jt})$ with respect to the wage W_{jt} . Then in any symmetric equilibrium where $W_{jt} = W_t$, $N_{jt} = N_t$, $V_{jt} = V_t$, $P_t^j = P_t$, and $Y_t^j = Y_t$, the wage Phillips curve characterizing nominal wage growth $\Pi_t^w \equiv \frac{W_t}{W_{t-1}}$ is given by:

$$\psi^{w} (\Pi_{t}^{w} = 1) \Pi_{t}^{w} + 1 = c(1+\chi) \left(\frac{V_{t}}{N_{t-1}}\right)^{\chi} \left[\frac{V_{t}}{N_{t}}\varepsilon_{R,W_{t}} + (-\varepsilon_{S,W_{t}})\frac{N_{t-1}}{N_{t}}\frac{S_{t}(W_{t})}{R_{t}(W_{t})}\right] + \frac{1}{1+\rho}\psi^{w} (\Pi_{t+1}^{w} = 1) (\Pi_{t+1}^{w})^{2}\frac{N_{t+1}}{N_{t}}.$$
(7)

Thus, in any symmetric equilibrium where firms solve an optimization problem of the form of (5), the wage Phillips curve will be a function of the current and expected future paths of the job vacancy rate V_t , employment N_t , the recruiting and separation rates $R_t(W_t)$ and $S_t(W_t)$, and their elasticities, denoted $\varepsilon_{R,W_t} > 0$ and $\varepsilon_{S,W_t} < 0$ following conventions in the monopsony literature; see e.g. Bloesch et al. (2024).¹³

This wage Phillips curve (7) captures how competition for workers affects firms' Interpretation optimal wage growth. The first term $(\frac{V_t}{N_{t-1}})^{\chi}$ captures the convex cost of posting vacancies: since firms must post vacancies to attract workers, higher marginal vacancy posting costs raises the value of both recruiting a worker the firm has matched with as well as retaining existing workers. If getting a worker in the door is more valuable, then firms will want to pay higher wages. The next term, within brackets, includes the recruiting elasticity term ε_{R,W_t} , which captures how sensitive the probability of hiring a matched worker is to the wage. If this recruiting elasticity is elevated, then workers' acceptance probability will be more sensitive to the wage, increasing the incentive at the margin for a firm to raise its wage. This ε_{R,W_t} is multiplied by the number of vacancies V_t . Next is the separation elasticity term ε_{S,W_t} . This elasticity is negative, so the negative of the separation elasticity is positive. A more steeply negative separation elasticity means that workers' likelihood of quitting is more sensitive to wages, so the more negative this value is, the greater the incentive to raise wages at the margin. Lastly, we have the separation rate $S_t(W_t)$ and recruiting rate $R_t(W_t)$. A higher separation rate indicates that workers have more opportunities to quit, increasing pressure for firms to raise wages. Analogously, when the recruiting rate $R_t(W_t)$ is higher, workers are easier to hire, lowering the pressure for firms to raise wages.

The next section describes the household and workers' optimization problems, which determine the recruiting and separation functions faced by firms. Having done so, we can then loglinearize and simplify (7) above to evaluate the model's ability to match the empirical wage Phillips curve discussed in Section 2. We do this in Sections 3.3 and 4, after discussing the model's implications for pass-through and wage-price spirals.

¹³Appendix B.2 derives the firm's first-order conditions in (5), including the price Phillips curve and wage Phillips curve (7).

3.2 Households and Workers

This section derives the household and worker block of the model. We deviate from the standard assumption in the New Keynesian literature of perfect consumption insurance within the household by assuming that households only imperfectly insure the consumption of workers who are unemployed, consistent with evidence that unemployed workers consume less than employed workers (see e.g., Chodorow-Reich and Karabarbounis (2016)). We assume that workers themselves choose whether to take a particular job offer, making employment decisions based on relative wages and consumption levels, in addition to idiosyncratic firm-specific preference shocks. Workers' mobility decisions aggregate up into the firms' recruiting and separation functions. Households smooth aggregate consumption within the household over time, yielding a standard Euler equation, making the labor block easy to integrate into a standard New Keynesian setting.

Frictional Markets Workers and firms match according to random search in a frictional market. As mentioned above, each firm j posts V_{jt} vacancies, and aggregate vacancies are $V_t = \int V_{jt} dj$. At the beginning of each period, a share of workers $s \in [0, 1]$ exogenously separate and enter unemployment Also each period, employed workers can search on the job with some constant, exogenous probability $\lambda_{EE} \in [0, 1]$, and unemployed workers always search.¹⁴

The unemployment rate is defined as U_t , so the total mass of searchers S_t becomes $S_t = \lambda_{EE}(1 \quad U_{t-1}) + U_{t-1}$, which includes on-the-job searchers from this period and unemployed workers from last period. Matching is random and follows a constant returns to scale matching function $M_t(V_t, S_t)$, given by

$$M_t(V_t, \mathcal{S}_t) = \frac{\mathcal{S}_t V_t}{(\mathcal{S}_t^{\nu} + V_t^{\nu})^{\frac{1}{\nu}}},$$

with $\nu = 2$ following the literature. Labor market tightness is $\theta_t = \frac{V_t}{S_t}$. The job finding rate for workers, $f(\theta_t) = \frac{M_t}{S_t}$, is increasing in tightness θ_t , and the probability that a vacancy is matched with a worker $g(\theta_t) = \frac{M_t}{V_t}$ is a decreasing function of tightness θ . The share of searchers who are employed is $\phi_{E,t} = \frac{\lambda_{EE}(1 \ U_t \ 1)}{S_t}$, and the share of searchers who are unemployed is $\phi_{U,t} = 1 \quad \phi_{E,t} = \frac{U_t \ 1}{S_t}$. The job finding rate for workers $f(\theta_t)$ and vacancy-filling rate $g(\theta_t)$ are given by

$$f(\theta_t) = \frac{\theta_t}{(1 + \theta_t^{\nu})^{\frac{1}{\nu}}}, \ g(\theta_t) = \frac{1}{(1 + \theta_t^{\nu})^{\frac{1}{\nu}}}.$$

Households A representative household has a unit mass $i \in [0, 1]$ of members who can work. Households seek to maximize the discounted present value of their members' utility, which is log

¹⁴This simplifying assumption ignores the possibility that on-the-job search intensity increases with the price level (Hajdini et al., 2023; Pilossoph and Ryngaert, 2024); Appendix A relaxes this assumption and finds that pass-through remains small even if workers search more intensely as prices rise.

in consumption; see Section 4.3 for discussion of the more general CRRA utility case. Without loss of generality, assume that unemployed household members must each have the same consumption level, $C_t^{u.15}$ Then letting $C_t(i, j(i))$ denote the consumption of worker *i* in state j(i), where j(i) indicates the firm *i* is employed at, the household's objective function becomes

$$\sum_{t=0}^{\infty} \left(\frac{1}{1+\rho}\right)^t \left[U_t \ln(C_t^u) + \int_0^{1-U_t} \ln\left(C_t(i,j(i))\right) di \right].$$

The household is allowed to choose C_t^u (effectively, an unemployment benefit) and also a linear tax/subsidy on employed workers, who consume their income each period:

$$C_t(i, j(i)) = \tau_t \frac{W_{j(i)t}}{P_t}$$

subject to the following budget constraint: letting D_t be nominal dividend payments from services firms (who profit from monopoly and monopsony power) and perfectly competitive goods firms (who receive the endowment X_t and sell it, rebating the proceeds to households), B_t be nominal bond holdings in zero net supply paying nominal interest rate i_t , and finally letting $\overline{W}_t \equiv \frac{1}{1 \quad U_t} \int_0^1 U_t W_{j(i)t} di$ be the average wage of employed workers, the budget constraint is

$$U_t C_t^u = \frac{D_t}{P_t} - \frac{B_t}{P_t} + \frac{(1+i_{t-1})B_{t-1}}{P_t} + (1-\tau_t)(1-U_t)\frac{W_t}{P_t}.$$
(8)

To make further progress in delivering a tractable model with households' standard consumption Euler equation, we impose an *ad hoc* consumption sharing rule within the household requiring that unemployed workers' consumption must be a *constant* fraction of employed workers' average consumption:

$$\frac{\bar{C}_t^e}{C_t^u} = \xi,\tag{9}$$

where $\xi \ge 1$ and $\bar{C}_t^e \equiv \frac{1}{1 \ U_t} \int_0^{1 \ U_t} C(i, j(i)) di$ is the average consumption of employed. This rule allows us to capture the fact that the ratio of unemployed and employed consumption is relatively constant over the business cycle (Chodorow-Reich and Karabarbounis, 2016). Moreover, it can be thought of as the result of a household facing an incentive-insurance trade-off: by insuring unemployed workers less and making unemployment relatively worse (higher ξ), the household provides greater incentive for its members to take jobs, at the expense of taking consumption away from unemployed workers with higher marginal utility of consumption. Note that in Section 4.2, we will study an extension of this model in which the household implements a different unemployment

¹⁵This is not restrictive, as given our other assumptions the household will always choose to equalize consumption across unemployed agents due to diminishing marginal utility of consumption.

insurance scheme where the consumption ratio is not held constant.

3.2.1 Symmetric Equilibrium Features a Standard Euler Equation

In a symmetric equilibrium where all firms set the same wage, and so $W_{jt} = W_t$, the household's problem under constraints (8) and (9) simplifies to choosing aggregate consumption C_t and bond holdings B_t to maximize

$$\sum_{t=0}^{\infty} \left(\frac{1}{1+\rho}\right)^t \ln\left(\frac{C_t}{(1-U_t)\xi + U_t}\right)$$

subject to the simplified budget constraint

$$C_t = \frac{D_t}{P_t} - \frac{B_t}{P_t} + \frac{(1+i_{t-1,t})B_{t-1}}{P_t} + (1 - U_t)W_t.$$

Optimization yields the standard consumption Euler equation with log utility given by:

$$C_t^{-1} = \frac{1}{1+\rho} \frac{1+i_{t,t+1}}{\Pi_{t+1}} C_{t+1}^{-1}, \tag{10}$$

where $\Pi_{t+1} \equiv \frac{P_{t+1}}{P_t}$.

Workers Workers get utility from consumption and an idiosyncratic preference draw ι . ι represents how much workers like their current job at firm j, which is redrawn every period and is i.i.d. Workers draw a similar preference shock each period during unemployment (note that the household does not take the idiosyncratic preference shocks into account when solving the problem described above). Workers are myopic¹⁶ and consider their utility only one period at a time, which for worker i in state j is given by¹⁷

$$\mathcal{V}_t(i,j) = \ln\left(C_t(i,j)\right) + \iota_{ijt}.$$

These assumptions of myopic workers and log utility will greatly help with tractability, and we discuss how relaxing these assumptions have small effects on our results in Section 4.3.

Workers are allowed to search on the job with probability λ^{EE} , and conditional on searching,

¹⁶Section 4.3 and Supplementary Appendix **F** which analyzes the case where workers are forward looking, reaching the conclusion that doing so adds considerably to the complexity of the model and burdens discussion without changing the dynamics of the model's response to cost-of-living shocks.

¹⁷The absence of utility from leisure, which may be greater in unemployment, is a simplifying assumption: we can introduce leisure without changing the results if the elasticity of substitution between leisure and consumption is one. See Appendix B.1 for further discussion on how assuming a different elasticity affects the results.

are matched with a vacancy with probability $f(\theta)$. Workers are allowed to consider unemployment with probability λ^{EU} . Consider a worker *i* currently employed at firm *j* who successfully matches with firm *k*'s vacancy. She will move to firm *k* only if $\mathcal{V}_t(i, k) \ge \mathcal{V}_t(i, j)$. Let us define $s_{jk}(W_{jt}, W_{kt})$ as the probability that the worker is poached from firm *j* to firm *k*, conditional on matching with firm *k*'s vacancy.

We assume that ι follows a Type-1 extreme value distribution with scale parameter γ^{-1} for tractability. Following the consumption sharing rule in (8) and (9), $s_{jk}(W_{jt}, W_{kt})$ is given by

$$s_{jk}\left(W_{jt}, W_{kt}\right) = \frac{\left(\tau_t \frac{W_{kt}}{P_t}\right)^{\gamma}}{\left(\tau_t \frac{W_{kt}}{P_t}\right)^{\gamma} + \left(\tau_t \frac{W_{jt}}{P_t}\right)^{\gamma}} = \frac{W_{kt}^{\gamma}}{W_{kt}^{\gamma} + W_{jt}^{\gamma}},\tag{11}$$

which is decreasing in W_{jt} : if firm j pays a higher wage, workers are less likely to be poached. Notice also that due to log utility, the probability a worker switches jobs is only a function of the relative *nominal* wage. The worker takes as given the internal tax rate set by the household τ_t and the price level P_t , both of which are unchanged regardless of which job the worker chooses.

Now consider a worker who is deciding whether to quit into unemployment. Let the average wage of employed workers in worker *i*'s household be \overline{W}_t , which determines consumption in unemployment through $C_t^u = \frac{\overline{C}_t^e}{\xi} = \tau_t \frac{\overline{W}_t}{\xi P_t}$.¹⁸ Since a worker *i* who is currently employed at firm *j* quits into unemployment only if $\mathcal{V}(i, j) \leq \mathcal{V}(i, \text{ unemployed})$, thus the probability that a worker voluntarily quits into unemployment $s_{ju}(W_{jt})$ is given by

$$s_{ju}(W_{jt}) = \frac{\left(\frac{1}{\xi}\tau\frac{\bar{W}_t}{P_t}\right)^{\gamma}}{\left(\frac{1}{\xi}\tau\frac{\bar{W}_t}{P_t}\right)^{\gamma} + \left(\tau\frac{W_{jt}}{P_t}\right)^{\gamma}} = \frac{\left(\frac{\bar{W}_t}{\xi}\right)^{\gamma}}{\left(\frac{\bar{W}_t}{\xi}\right)^{\gamma} + W_{jt}^{\gamma}},\tag{12}$$

which is decreasing in W_{jt} but does not depend on the price level P_t . This result of independence from the price level comes from the combined assumptions of the consumption sharing rule and log utility over consumption. Later in Section 4.3, we relax this log utility assumption and introduce constant relative risk aversion (CRRA) preferences of workers.

These individual transition probabilities aggregate up into the firm's separation rate $S(W_{jt})$: each period, a share of workers $s \in (0, 1)$ exogenously separate while the remainder (1 - s)endogenously separate if they receive an opportunity that they prefer to their current job (either another job, or the chance to exit to unemployment). Recalling that $f(\theta_t)\lambda_{EE}$ denotes the probability that a particular employed worker is allowed to search on the job and matches to another firm, and that λ_{EU} denote the probabilities that an employed worker is allowed to consider quitting

¹⁸Note that in the symmetric equilibrium in the next section, the average wage earned by workers in each household \bar{W}_t will equal the aggregate wage W_t .

into unemployment, the separation rate is written as

$$S_{t}(W_{jt}) \equiv S_{t}(W_{jt}|\{W_{kt}\}_{k\neq j}) = s + (1 \ s) \left[\lambda_{EE} f(\theta_{t}) \int s_{jk}(W_{jt}, W_{kt}) z(W_{kt}) dk + \lambda_{EU} s_{ju}(W_{jt}) \right],$$
(13)

where $z(W_{kt})$ is an endogenous density function of outside posted wages. Note that $S_t(\cdot)$ is a decreasing function of W_{jt} , i.e. $S'_t(W_{jt}) < 0$, since all of it's components are decreasing in W_{jt} ; in other words, the firm's separation rate falls as the wage rises.

Analogously to the individual separation probabilities, there are probabilities that a matched worker is recruited into the firm conditional on whether the worker is employed or unemployed. Consider a worker employed at firm k that encounters firm j's vacancy. The probability that firm j successfully poaches the worker $r_{kj}(W_{kt}, W_{jt})$ is:

$$r_{kj}(W_{kt}, W_{jt}) = \frac{\left(\tau_t \frac{W_{jt}}{P_t}\right)^{\gamma}}{\left(\tau_t \frac{W_{kt}}{P_t}\right)^{\gamma} + \left(\tau_t \frac{W_{jt}}{P_t}\right)^{\gamma}} = \frac{W_{jt}^{\gamma}}{W_{kt}^{\gamma} + W_{jt}^{\gamma}},\tag{14}$$

which is increasing in W_{jt} and is a function of relative wages.

Now consider an unemployed worker who is matched with firm j's vacancy. The probability that the worker takes the job with firm j is defined as $r_{uj}(W_{jt})$ and is equal to

$$r_{uj}(W_{jt}) = \frac{\left(\tau_t \frac{W_{jt}}{P_t}\right)^{\gamma}}{\left(\frac{1}{\xi}\tau_t \frac{\bar{W}_t}{P_t}\right)^{\gamma} + \left(\tau_t \frac{W_{jt}}{P_t}\right)^{\gamma}} = \frac{W_{jt}^{\gamma}}{\left(\frac{\bar{W}_t}{\xi}\right)^{\gamma} + W_{jt}^{\gamma}},\tag{15}$$

which is increasing in W_{jt} .

We can use (14) and (15) to write firm j's recruiting rate, defined as the share of vacancies that successfully result in hiring a worker. Recalling that $g(\theta_t)$ denotes the probability that a vacancy is matched with a worker, and that $\phi_{E,t}$ and $\phi_{U,t}$ denote the share of searchers who are employed and unemployed, respectively, we can write the recruiting rate as:

$$R_t(W_{jt}) \equiv R_t(W_{jt}|\{W_{kt}\}_{k \neq j}) = g(\theta_t) \left[\phi_{E,t} \int_k r_{kj}(W_{kt}, W_{jt}) \omega(W_{kt}) dk + \phi_{U,t} r_{uj}(W_{jt}) \right].$$
(16)

where $\omega(W_{kt})$ is the distribution of wages that workers are currently employed at. The recruiting rate $R_t(W_{jt})$ is an increasing function because all of its components r_{kj} and r_{uj} are also increasing in W_{jt} . In other words, a higher wage improves the firm's odds of recruiting workers through its vacancies.

3.2.2 Symmetric Equilibrium Features Simple Separation and Recruiting Functions

In a symmetric equilibrium where all the firms set the same wage, i.e., $W_{jt} = W_t$ for $\forall j$, both $S(\cdot)$ and $R(\cdot)$ becomes functions of tightness θ_t and simplify from (13) and (16) to

$$S_t = s + \begin{pmatrix} 1 & s \end{pmatrix} \left(\lambda_{EE} f(\theta_t) \frac{1}{2} + \lambda_{EU} \left(\frac{1}{1 + \xi^{\gamma}} \right) \right)$$
(17)

$$R_t = g(\theta_t) \left(\phi_{E,t} \frac{1}{2} + \phi_{U,t} \left(\frac{\xi^{\gamma}}{1 + \xi^{\gamma}} \right) \right).$$
(18)

where the aggregate wage W_t does not appear on the right hand side of (17) and (18). Therefore, in a symmetric equilibrium, both separation and recruiting rates S_t and R_t become independent of aggregate wage W_t . This is because all the competing firms set the same wage level, and the relative desirability of employment over unemployment is independent of the wage due to household's consumption sharing rule (9).

The reason for the absence of the price level P_t in the separation rate formula (17) is similar: with log utility from consumption, the price level is irrelevant to a worker considering choosing between two different nominal wage offers.¹⁹ Additionally, the price level is irrelevant for workers considering choosing between working and *unemployment*. This is due to the combination of log utility of workers and the households' consumption sharing rule (9) which fixes the relative consumption of employed and unemployed workers at ξ and appears in equations (17) and (18) above: the higher real consumption ratio ξ is on average, the more likely unemployed workers are to prefer the state of employment to that of unemployment, so S_t decreases with ξ and R_t increases with ξ all else equal. Note that we can relax our assumption that the consumption ratio ξ is fixed without changing the results for pass-through: the key here is that the price level, P_t , has no effect on both S_t and R_t in equilibrium. Fixing unemployment benefits at some nominal level, for example, would still result in the relative attractiveness of employment and unemployment being insensitive to the price level by the same logic that applies to employed workers choosing between nominal wages at two different jobs.

As we will show in Section 4.1, there is no pass-through at all in our baseline case where the relative consumption of employed and unemployed workers is fixed at ξ . However, the assumption that the relative desirability of unemployment (formally, the probability of preferring a job offer at aggregate wage W_t to the unemployment state) is constant and completely independent of the aggregate price level is strong; this would not be true if, for example, we had assumed that the representative household insures unemployed households by guaranteeing them some constant,

 $^{^{19}}$ In Section 4.3 and Supplementary Appendix G, we show that while a more general CRRA utility does reintroduce the price level into worker mobility decisions, reasonable levels of risk aversion only minimally affect our results quantitatively.

real unemployment benefit b (i.e., if unemployment benefits are perfectly indexed to inflation), or if we had assumed that workers derive some utility from leisure, as well as consumption, and that leisure utility is systematically higher while unemployed. We discuss the former case in Section 4.2; Appendix B.1 discusses the worker's problem with leisure, which has similar implications but requires more burdensome notation.²⁰

3.3 Equilibrium Selection

We can close the model with a simple Taylor rule, with a potential policy shock $\varepsilon_{i,t}$:

$$1 + i_t = \Pi_{Y,t}^{\phi_\Pi} (1 + \varepsilon_{i,t}) \tag{19}$$

where $\Pi_{Y,t} \equiv \frac{P_{y,t}}{P_{y,t-1}}$ is inflation in services prices and with $\phi_{\Pi} = 2$, and solve for a symmetric equilibrium. Our symmetric equilibrium consists of sequences of all endogenous prices and quantities satisfying that: (i) firms choose identical sequences such that $W_{jt} = W_t$, $N_{jt} = N_t$, $V_{jt} = V_t$, $P_{y,t}^j = P_{y,t}$, (ii) workers and households maximize utility, (iii) firms maximize profits, (iv) product markets clear, and (v) labor market flows add up.

We linearize these necessary conditions in a symmetric equilibrium around a non-stochastic steady state, and solve for the unique solution. While there is a unique, symmetric equilibrium (for our given parameter values) we cannot rule out and leave unexplored the possibility of non-symmetric equilibria where *ex ante* identical firms choose different wages. The fact that we have one wage in equilibrium, while still having worker flows between unemployment and various firms due to idiosyncratic shocks, buys us a highly tractable dynamic model with on-the-job search.

Implications for Wage Growth Specifically, consider the firm's wage Phillips curve (7). In a symmetric equilibrium, the separation and recruiting rates and elasticities appearing in (7) depend only on θ_t and $\phi_{E,t}$, the share of searchers who are already on the job.²¹ Thus, we can show that in equilibrium the wage Phillips curve up to a first order is

$$\check{\Pi}_t^w = \beta_\theta \check{\theta}_t + \beta_U \check{U}_{t-1} + \frac{1}{1+\rho} \check{\Pi}_{t+1}^w$$
(20)

²⁰This is because we must take a stance on the elasticity of substitution between leisure and consumption; Appendix **B.1** demonstrates that if leisure and consumption have an elasticity of substitution of one, then changes in the price level have no effect on the relative desirability of employment for a given nominal wage, as in the benchmark case with fixed consumption ratios.

²¹This can be seen by observing the separation and recruiting rate expressions (17) and (18) above; the same can be shown for their elasticities. The derivation of (20) is provided in Appendix B.3.

where the "check" (\check{x}) variables denote log deviations from steady state. The appearance of θ_t , rather than $\frac{V_t}{U_t}$, in the wage Phillips curve reflects the presence of on-the-job searchers: intuitively, since unemployed workers are not the only job searchers, labor market tightness θ is not $\frac{V}{U}$ but $\theta \equiv \frac{V}{S}$. In the model, when θ_t is high, workers are harder to both recruit and retain, putting pressure on firms to raise wages (i.e., $\beta_{\theta} > 0$).

The appearance of unemployment U_{t-1} in (20) alongside θ_t reflects two features of our model. First, note that our benchmark calibration features unit vacancy posting costs that are convex in $\frac{V}{N}$, so that when U is low, firms are larger (N is big), making hiring via vacancies easier, implying *downward* pressure on wages. This yields a slightly *positive* β_U in our benchmark calibration ($\chi = 1$); assuming a linear cost ($\chi = 0$) in V actually yields slightly negative β_U .

The reason that β_U becomes negative under $\chi = 0$ arises from the second feature: the *composition* of searchers matters for wage growth beyond the relative number of jobs and searchers. Because our model's calibration implies that unemployed workers accept job offers with high probability, their job-taking decision is not very sensitive to the offered wage, in contrast to the decision by employed workers. Thus when U_{t-1} is high, and relatively more searchers are unemployed, optimizing firms prefer to acquire workers by posting vacancies, rather than raising wages. However, this effect is quantitatively small in the calibrated benchmark model ($\beta_U \approx 0$): even if unemployed workers' job-taking decision is much less wage-sensitive than employed workers' decision, changes in unemployment U_{t-1} do not change the equilibrium wage much given θ_t . Consistent with this intuition, Appendix B.3.1 shows that setting $\chi = 0$ and shutting down on-the-job search $\lambda_{EE} = 0$ yields a wage Phillips curve which depends on the ratio $\frac{V}{U}$ alone; intuitively, with only unemployed searchers, there is no role for composition effects, and labor market tightness $\frac{V}{U}$ summarizes wage pressures. Appendix D discusses this issue in further detail.

Finally, note that using the tight relationship between tightness θ_t and quits Q_t , which are the endogenous component of separations ($Q_t = S_t - s$), shown in equation (17), we can write wage inflation as:

$$\check{\Pi}_{t}^{w} = \beta_{Q}\check{Q}_{t} + \beta_{U}\check{U}_{t-1} + \frac{1}{1+\rho}\check{\Pi}_{t+1}^{w}.$$
(21)

Because workers receive more job offers and thus quit more frequently when labor market tightness is high, the model predicts that either quits Q_t or θ_t can summarize wage pressures—along with unemployment.

In summary, given that $\beta_U \approx 0$, this simple wage Phillips curve reveals that current and future expected path of labor market tightness $\theta_t = V_t/(U_{t-1} + \lambda_{EE}(1 - U_{t-1}))$ or quits Q_t is effectively a *sufficient statistic* for wage growth. Neither conventional demand shocks (i.e., monetary policy shocks ϵ_t) nor conventional supply shocks (both aggregate technology shocks and "cost of living shocks") appear in the wage Phillips curve; see Appendices B.2 and B.3 for a derivation (which allows for aggregate TFP shocks). These shocks will only affect wage growth in equilibrium through their effects on labor market tightness or quits rate.

3.4 Calibration

We calibrate the model to match labor market flows and the sensitivity of recruiting and separation rates to firm wage policies. For labor market flows, we calibrate the model to match U.S. data during the period 2015-2019 to capture the approximately full-employment conditions that existed prior to the COVID shock. Data on the unemployment rate and separation rate come from the Bureau of Labor Statistics (BLS) and the Job Openings and Labor Turnover Survey (JOLTS). For the sensitivity of recruiting and separation rates to wages, we choose the inverse scale parameter for workers' idiosyncratic preferences γ to match the recruiting and separation elasticities estimated in Bassier et al. (2022).

We set the consumption ratio $\xi = 2$, which is higher than in Chodorow-Reich and Karabarbounis (2016) but closer to what maximizes steady-state utility for the household in our setting; the results are largely insensitive to changing this parameter. We otherwise choose standard values for most parameters; Table 2 lists the model's calibrated parameters, some of which are chosen to target other moments in U.S. data given in Table 3.

Wage Phillips Curve Table 4 reports the model implied coefficients on β_Q and β_U from equation (21) given the calibration in Table 2, giving $\beta_Q = 2.48$ and $\beta_U = 0.09$. We compare these modelimplied coefficients to the regression results in Column (5) of Table 1. Consistent with empirical evidence, our structural wage Phillips curve puts a large weight on quits, and the coefficient on unemployment becomes nearly zero. In Appendix E, we revisit this result and compare the modelimplied and empirical wage Phillips curves in more detail.

We conclude by noting that the model is broadly consistent with the the wage Phillips curve describing the U.S. labor market. In the next section, we show that the model will also imply that there is very little scope for pass-through from a cost-of-living shock to wages and wage-price spirals in the United States.

4 Pass-Through from Prices to Wages and Wage-Price Spirals

This section studies the effects of a cost-of-living shock, defined as the following thought experiment: what happens to nominal wages when an unanticipated, temporary, negative shock to the endowment good X raises the price level at t = 0? We also demonstrate that the model generates

Parameter	Value	Meaning	Reason		
λ_{EE}	.14	OTJ search probability	Match EE rates		
λ_{EU}	.30	Opportunity to quit probability	Match voluntary EU rate, Qiu (2022)		
ξ	2	Consumption ratio: C_t^e/C_t^u	See Notes below		
s	.01	Exogenous separation rate	Match JOLTS monthly separation Rate		
γ	6	Scale parameter ¹ of i.i.d. preferences	Match $\varepsilon_{R,W}$ $\varepsilon_{S,W}$		
ϵ	10	Elasticity of substitution of services			
ψ	100	Services price adjustment cost			
ψ^w	100	Wage adjustment cost			
η	1	Services/endowment good EOS			
α_X	.2	Endowment good share in CES Utility			
χ	1	Convexity in vacancy posting costs	Bloesch et al. (2024)		
c	30	Hiring cost shifter	Targeting U		
ho	.004	Discount Rate	Monthly model		

Table 2: Parameters in the Monthly Benchmark New Keynesian Model

Table 3: Selected Moments in Model Steady State and Data

Targeted Moment	Meaning	Model	Data	Source
U	Unemployment rate	.044	.044	BLS
S	Monthly separation rate	.036	.036	JOLTS
$\varepsilon_{R,W}$ $\varepsilon_{S,W}$	Recruiting minus separation elasticities	4.4	4.2	Bassier et al. (2022)

Notes: We calibrate the model to match labor market flows of the U.S. economy during 2015-2019 to capture the approximately full employment conditions that existed prior to the COVID shock. Data on the unemployment rate and separation rate come from the Bureau of Labor Statistics (BLS) and the Job Openings and Labor Turnover Survey (JOLTS). We set $\xi = 2$, higher than in Chodorow-Reich and Karabarbounis (2016) but closer to what maximizes steady-state utility for the household in our setting; the results are largely insensitive to changing this parameter.

positive co-movement between the quits rate and nominal wage growth in response to demand shocks, e.g., monetary policy shocks.

Section 4.1 demonstrates quantitatively that in Section 3's baseline model cost-of-living shocks have "zero" effect on wages. We also show that monetary policy shocks cause quits and wages to co-move, as in the data. Section 4.2 presents quantitative results for pass-through in an extension to the model in which price level increases make unemployment more attractive for a given nominal wage, generating a positive response of wages to higher cost of living. However, we show that on-the-job search dramatically dampens the pass-through of cost-of-living shocks to wages. This is because incorporating on-the-job search makes wages less sensitive to the value of unemployment, consistent with micro-evidence that wages are unresponsive to the value of unemployment benefits

Source	β_Q	β_U		
Baseline Model: $\chi = 1$	2.48	0.09		
Baseline Model: $\chi = 0$	2.13	-0.11		
OLS using ECI 1990-Present	1.00***	-0.02		
	(0.12)	(0.07)		
Standard errors in parentheses (Newey-West; 4 lags)				
*** p<0.01, ** p<0.05, * p<0.1				

Table 4: Structural Wage Phillips Curve Coefficients vs. OLS Coefficients

Notes: The models' structural wage Phillips curve is broadly in line with the OLS estimates from U.S. data, putting much more weight on quits than unemployment. The table compares the model's calibrated structural wage Phillips curve with OLS estimates from Column (5) of Table 1.

in Jäger et al. (2020). Section 4.3 briefly discusses further extensions including endogenous onthe-job search probability, CRRA utility, and forward-looking workers, arguing that none of these extensions affects the conclusion from Section 4.2 that pass-through of cost-of-living shocks to wages is quantitatively small.

4.1 Response of Wages to Cost-of-living and Monetary Policy Shocks: Baseline Model

This section analyzes the response of the baseline model described in Section 3 in response to both cost-of-living shocks (which do not move wages) and monetary policy shocks (which do move wages). We use the calibration presented in Table 2.

4.1.1 Cost-of-Living Shocks

We subject the economy to a 10% negative quantity shock of the endowment good X_t from $X_t = 1$. Given the assumption of a unit elasticity of substitution between services and goods in final aggregation, i.e., $\eta = 1$, this implies a 10% relative price shock to good X_t and an increase in the overall price index P_t . Additionally, we assume that the monetary authority leaves nominal interest rates pegged:²² given the household's Euler equation (10) and the fact that $\eta = 1$ implies constant expenditure shares across services Y_t and endowments X_t , this experiment effectively holds aggregate demand for services constant, leaving the labor market unaffected.

²²In this case, monetary pegging effectively satisfies the Taylor rule (19), as services price inflation $\Pi_{Y,t}$ remains unchanged in response to our cost-of-living shocks. Pegged monetary policy in this environment stabilizes N_t , unless firms charge a higher services price $P_{y,t}$ in response to a cost-of-living shock, which is not the case here. As seen in our price Phillips curve (B.11) in Appendix B.2, services price inflation depends on nominal wage growth and the same set of labor market variables that appear in the wage Phillips curve. Thus the cost-of-living shock does not affect service price inflation because it does not affect the equilibrium wage or those labor market variables.



Figure 3: Impulse Response to a 10% Negative Shock to Supply of Endowment Good

Notes: This figure presents the effects of a decreased supply of the endowment good X_t , which we call a pure cost-of-living shock, under a nominal interest rate peg. Given the assumption of a unit elasticity between services and goods in final aggregation, this implies a 10% relative price shock to good X_t and an increase in the overall price index P_t . Given the household's Euler equation and constant expenditure shares, the nominal interest rate peg experiment effectively holds aggregate demand for services constant. Since the shock does not affect the relative attractiveness of unemployment and working, the recruiting and separation elasticities faced by firms are also unchanged as discussed in Section 3.2.2: the result is no change in vacancy posting, no change in tightness, and no change in the *nominal* wage, which causes *real* wages to fall as shown in the last panel.

Figure 3 shows the results. In response to the cost-of-living shock, headline inflation Π_t increases, but nominal wage growth Π_t^w and labor market tightness θ_t are unchanged. This zero result occurs for two reasons. First, because of the household's consumption sharing rule, higher cost of living lowers real consumption by the same proportion for both employed and unemployed members. Because workers' utility is log in consumption, the relative desirability of employment and unemployment are unchanged. Therefore, both the probability that an employed worker quits into unemployment and the probability that an unemployed worker accepts a job are unchanged. Second, because monetary policy stabilizes employment N_t and therefore services output Y_t , firm demand for labor is unchanged in response to the cost of living shock. With labor demand and the probabilities that workers flow across labor market states unchanged, firms can maintain employment levels by keeping vacancies constant, which subsequently holds tightness θ_t constant. Because all of the real labor market variables are constant, firms' optimal wage is unchanged.

This zero result can be seen in our nonlinear wage Phillips curve (7) in Section 3.1, which writes wages exclusively as a function of labor market variables: vacancies, employment, recruiting and separation rates, and their elasticities. If none of the labor market variables are affected by the higher price level, then there will be no effect on nominal wages. The net result from the cost-of-living shock is that real wages fall on impact, and revert when the shock has passed. Furthermore, because wages and labor market tightness are unchanged, services inflation Π^Y is also zero.²³

4.1.2 Monetary Policy Shock

To show that the model is capable of generating a positive relationship between quits and wages, as in the data, we also analyze the effect of an expansionary monetary policy shock on wages and quits. Specifically, we subject the economy to a one period, 1 percentage point decrease in nominal interest rates, with a monthly persistence of 0.8. On impact, nominal wage growth and the quits rate rise together, as seen in Figure 4. Recall that the quits rate is defined in the model as separations less exogenous separations: $Q_t = S_t$ s. Lower nominal interest rates raise demand for services consumption, which increases demand for labor. Firms then post more vacancies, increasing opportunities for workers to find other jobs, which raises job-to-job quits, while also increasing competition among firms for workers, leading to higher wages. This result demonstrates that our model can rationalize co-movement between quits rate and wage growth, documented in Figure 2 and the literature (Faberman and Justiniano, 2015; Moscarini and Postel-Vinay, 2017), through demand shocks like monetary policy shocks.

4.2 Extension: Inflation-Indexed Unemployment Insurance (UI)

This section revisits the cost-of-living shock experiment of Section 4.1.1 while relaxing the assumption that the relative desirability of unemployment and employment is held fixed by the household's consumption sharing rule (9), allowing the relative desirability of unemployment to rise along with the price level. We then show how on-the-job search mutes the pass-through from prices to wages in this variant of the model.

4.2.1 Separation and Recruiting Rates

We now assume that households no longer fix the ratio of consumption between employed and unemployed workers, but instead provide unemployed workers some inflation-indexed quantity

²³We derive the price Phillips curve for services in Appendix B.2. Real marginal costs in the service sector depend on the product wage (W/P^Y) and labor market variables. Since in response to a cost-of-living shock wages, employment, and labor market tightness are unchanged, then service firms' optimal price is unchanged, and services inflation is zero.



Figure 4: Expansionary 1% Decrease in the Policy Rate

Notes: This figure plots the effects of a 1% point decrease in nominal interest rates in our baseline model. Both nominal wage growth and quits rise as lower nominal interest rates raise demand for consumption, which increases demand for labor. Firms post more vacancies, increasing opportunities for workers to find other jobs, which increases the quits rate, while also increasing competition for workers, which raises wages. This result demonstrates that the model can rationalize co-movements between quits and nominal wage growth, documented in Faberman and Justiniano (2015), through demand shocks like monetary policy shocks.

of consumption, *b*. This is a convenient way to represent various model modifications, such as home production of final consumption goods, or utility from leisure if consumption and leisure are substitutes.²⁴ Our assumption of inflation-indexed unemployment benefits is appealing because it allows for the least change in notation from our baseline model.

Accordingly, for a given nominal wage, an increase in the price level now raises the relative consumption of unemployed agents, making unemployment more desirable. To see this, note that the probability that a worker separates from employment to unemployment is now given by

$$s_{ju}\left(W_{jt}|P_{t}\right) = \frac{b^{\gamma}}{\left(\frac{W_{jt}}{P_{t}}\right)^{\gamma} + b^{\gamma}}$$

The separation rate s_{ju} from employment to unemployment now depends on the price level: at a given nominal wage, higher prices makes unemployment more attractive. Similarly, the new

²⁴See Appendix B.1 for further discussion on the similarities between this approach and assuming that the value of unemployment is additional leisure.

recruiting function from unemployment is

$$r_{uj}\left(W_{jt}|P_t\right) = \frac{\left(\frac{W_{jt}}{P_t}\right)^{\gamma}}{\left(\frac{W_{jt}}{P_t}\right)^{\gamma} + b^{\gamma}},$$

where now we see that a higher price level makes recruiting from unemployment more difficult at a given nominal wage, by the same logic.

In a symmetric equilibrium where $W_{jt} = W_t$ for $\forall j$, the separation and recruiting rates become

$$S_t = s + \begin{pmatrix} 1 & s \end{pmatrix} \left(\lambda_{EE} f(\theta_t) \frac{1}{2} + \lambda_{EU} \frac{b^{\gamma}}{\left(\frac{W_t}{P_t}\right)^{\gamma} + b^{\gamma}} \right), \tag{22}$$

and

$$R_t = g(\theta_t) \left(\phi_{E,t} \frac{1}{2} + \phi_{U,t} \frac{\left(\frac{W_t}{P_t}\right)^{\gamma}}{\left(\frac{W_t}{P_t}\right)^{\gamma} + b^{\gamma}} \right).$$
(23)

Unlike the benchmark case represented by (17) and (18), the price level P_t affects the recruiting and separation rates via the probability of quitting into unemployment and the probability of successfully recruiting unemployed workers. Now a higher price level decreases the real wage but does not decrease the value of unemployment, weakening workers' incentive to accept or stay in a job.

All the other model equations (i.e. the first-order conditions of firms and the household) remain unchanged; Appendix B.4 shows how to derive an Euler equation in this setting which is identical to that used above, given appropriate assumptions on the representative household's optimization problem with a redistribution plan. We calibrate the model with a choice for unemployment benefit b instead of ξ ; we set b = 0.4 which results in a steady-state consumption ratio for employed to unemployed agents of 2, so that this moment is the same at the steady state as in the baseline model with $\xi = 2.^{25}$

4.2.2 Real Wage "Catch-up" in the Wage Phillips Curve

In Section 3.3, we derived the log-linearized wage Phillips curve for the baseline model and reported the coefficients on deviations of log quits and the log unemployment rate from the calibrated

²⁵Recall from the discussion in Table 2 that the consumption ratio $\xi = 2$ is higher than in Chodorow-Reich and Karabarbounis (2016) but closer to what maximizes the steady-state utility of households. While results in the baseline model are insensitive to changing ξ , modifying the model to allow for pass-through from cost of living shocks to wages as we do here implies changes in *b* matter: lowering *b* might raise or reduce the pass-through of cost of living to wages depending on λ_{EE} , but does not affect the result that on-the-job search λ_{EE} mutes this pass-through. Quantitatively, changes in *b* do not affect the level of pass-through much, as we will discuss in depth in Section 4.2.3.

model. In the extension with inflation-adjusted unemployment benefits, the price level affects labor market transitions, and so the real wage level will matter for the growth rate of nominal wages. The log-linearized wage Phillips curve now becomes²⁶

$$\check{\Pi}_{t}^{w} = \beta_{Q}\check{Q}_{t} + \beta_{U}\check{U}_{t-1} + \beta_{\tilde{w}}\check{\tilde{w}}_{t} + \frac{1}{1+\rho}\check{\Pi}_{t+1}^{w},$$
(24)

where $\tilde{w}_t \equiv \sum_{s=0}^{t-1} \Pi_s^w = \sum_{s=0}^{t} \Pi_s$ is a function of last period's real wage and current inflation, and $\beta_{\tilde{w}}$ indicates the strength with which nominal wages "catch-up" to price inflation.

Table 5 reports the coefficients from equation (24) from our calibrated model compared against the estimate coefficients of a regression of the same form as (24). As before, the model generates higher weight on quits than on unemployment in driving nominal wage growth. The estimated wage Phillips curve has a coefficient of $\beta_{\bar{w}} = .021$, which suggest that if real wages are 10% below steady state, nominal wage growth becomes 0.2% faster on a quarterly basis. Interestingly, the model implied coefficient on the real wage term is positive. This is because with inflation-adjusted unemployment benefits, higher prices will induce some employment-to-unemployment quits, and so higher inflation will be captured by the higher quits rate. Depending on the parameterization, this sign may be either positive or negative. Overall, however, the model still captures the high weight on quits in the wage Phillips curve and small weights on unemployment and price inflation.

 Table 5: Structural Wage Phillips Curve Coefficients vs. OLS Coefficients

Source	β_Q	β_U	$eta_{ ilde{w}}$
Real Unemployment Benefit Model ($\chi = 1$)	2.48	0.09	.0426
OLS using ECI 1990-Present	1.11***	-0.04	021***
	(0.16)	(0.07)	(.007)
Standard errors in parentheses (Newey-West; 4 lags)			

*** p<0.01, ** p<0.05, * p<0.1

Notes: This table reports the model implied coefficients for the wage Phillips curve (24). Both the model-implied coefficients and estimate coefficients put a high weight on quits, with the coefficient on the unemployment rate and real wage level close to zero.

Appendix E revisits this result in depth and compares the model-implied and empirical wage Phillips curves.

4.2.3 Pass-Through Results and the Role of On-the-Job Search λ_{EE}

We now show that on-the-job search quantitatively mutes pass-through of higher cost of living to wages when unemployment benefits are indexed to inflation. In this setting, a higher cost of

²⁶Derivations of (24) can be found in Supplementary Appendix I.

living diminishes the desirability of employment relative to unemployment, making unemployed workers harder to recruit and employed workers more likely to quit into unemployment, prompting firms to raise wages and generating pass-through of cost-of-living shocks to wages. In this setting increasing the frequency of on-the-job search dampens this pass-through of cost-of-living shocks to wages for two reasons.

First, as can be seen in equations (22) and (23), when on-the-job search is higher, firms face greater threat of separations to on-the-job search, and the share of searchers who will be on-the-job searchers increases. Since a higher cost of living lowers real wages at all jobs evenly, the probability that an on-the-job searcher accepts a given job offer is unaffected by the price level. If a greater share of separations and hires are job-to-job, where the price level does not matter, then mechanically the relative value of unemployment will matter less for wage setting, and cost-ofliving shocks will have less of an effect on optimal wages. Second, the presence of on-the-job search increases competition for labor, pushing wages further above the value of unemployment and making the changes in the relative value of unemployment less important for firm wage setting. For calibrations that deliver empirically realistic unemployment rates and job-to-job flows, equilibrium wages are far above the value of unemployment, and so most unemployed workers accept jobs (in our calibration, 98.5% of the time), and employed workers quit into unemployment very infrequently. In this case, marginal changes in the relative value of unemployment due to unemployment benefits being inflation-indexed have small effects on the probability that an unemployed worker accepts a job or an employed worker voluntarily quits. As a consequence, marginal changes in a firm's own wage primarily affects job-to-job hires and separations, and so firms are primarily concerned with competition for already-employed workers when setting wages. Since the price level has no direct effect on rate of job-to-job mobility, and firms mostly care about jobto-job mobility when setting wages, then the price level and the relative value of unemployment will have little effect on wages.

To illustrate this, we perform two exercises. First, we calculate the steady-state value of wages for different values of real unemployment benefits b and on-the-job search probabilities λ_{EE} . With even a modest amount of on-the-job search, we generate the empirical finding in Jäger et al. (2020) that the value of unemployment benefits has little effect on the equilibrium wage level. Figure 5 illustrates these results. As $\lambda_{EE} \rightarrow 0$, changing the real unemployment benefit b has large effects on the equilibrium real wage, as seen by examining the gaps between the blue solid line and dashed red line, for example. At our value of $\lambda = .14$ calibrated to U.S. data, we observe that the same changes in b yield almost no change in the equilibrium real wage offered by firms, as in the data. Thus, beyond the fact that incorporating on-the-job search is important to capture the fact that quits are mostly job-to-job quits rather than quits into nonemployment, on-the-job search helps capture the near irrelevance of unemployment benefits for the wage. Figure 5: On-the-Job Search Mutes the Effect of Changing Unemployment Benefits on Equilibrium Real Wages



Notes: In the model with fixed real unemployment benefits described in Section 4.2, eliminating on-the-job search by sending $\lambda_{EE} \rightarrow 0$ means that changes in unemployment benefit *b* have large effects on the equilibrium real wage (denominated in the price of aggregate consumption), seen by examining the gaps between the blue solid line and dashed red line. This is because without on-the-job search, firms set wages mostly considering the problem of recruiting unemployed workers, which makes the level of *b* important in their wage-setting problem. At our value of $\lambda = .14$ calibrated to U.S. data, where firms mostly recruit from other firms, we see that the same changes in *b* have almost no change in the equilibrium real wage offered, as in the data: the three lines lie on top of each other at this point.

In the second exercise, we again consider the effect of a cost-of-living shock on wages and prices under the same interest rate peg, showing how different frequencies of on-the-job search λ_{EE} affect the results.²⁷ Figure 6 presents the impulse response function of nominal wage growth, cumulative services price inflation, and cumulative headline inflation to a cost-of-living shock in the log-linearized model. The solid blue lines indicate the response of these variables under our standard calibration when $\lambda_{EE} = 0.14$, while dashed red lines show the response if we nearly shut down on-the-job search and set $\lambda_{EE} = 0.001$. As expected, the response of wages on impact from the cost-of-living shock in panel (a) of Figure 6 is positive, as the value of employment falls by

²⁷Notice in equations (22) and (23) that there is an isomorphism between P_t and b: a higher price level lowers the relative desirability of employment as does a change in the level of unemployment benefits b. Based on our findings in Figure 5 that the unemployment benefit has little effect on firms' optimal wage, we analogously can infer that with on-the-job search, changes in the price level will similarly have little effect on firms' optimal wage.



Figure 6: Response of Wage Growth and Inflation to Cost-of-Living Shock

Notes: This figure presents the path of nominal wage growth and cumulative price inflation from a decreased supply of the endowment good X_t under an interest rate peg, in a variant of the baseline model where increased cost of living lowers the desirability of employment as described in Section 4.2. While there is now some pass-through from the cost-of-living shock to wages, on-the-job search significantly dampens this result, as seen by comparing the results in calibrations where the on-the-job search probability, λ_{EE} is calibrated to match U.S. data (the solid blue line) with a calibration where workers are almost never allowed to search on the job (the dashed red line).

more than the value of unemployment, decreasing the willingness of unemployed workers to take jobs and encouraging firms to pay higher wages. However, the wage response decreases by nearly a factor of 10 in the presence of a realistic amount of on-the-job search. Given a cost-of-living shock that raises consumer prices by around 2%, the pass-through in the model with low levels of on-the-job search ($\lambda_{EE} = .001$) is 0.05, while a realistic amount of on-the-job search ($\lambda_{EE} = .14$) delivers pass-through from consumer prices to wages of less than 0.01.

Panel (b) of Figure 6 plots the cumulative response of services prices. On impact, services

prices fall, because firms post fewer vacancies (i.e., labor market tightness θ actually falls on impact). The reason wages and vacancies move in opposite directions is that the higher price level makes it harder to recruit workers out of unemployment, in contrast to the case in our baseline model. This makes vacancies less useful than wages as a recruiting tool; optimizing firms thus reduce the number of vacancies posted while raising wages as they attempt to retain workers (though note that an interest rate peg no longer holds N_t fixed in this extension of the model). Due to convexity in vacancy costs, fewer vacancies lowers the marginal cost of production, and so prices fall. However, in the second period after the shock as firms re-hire, marginal cost recovers, with the additional cost that wages grew in period 1 but never reverted to their initial nominal level. By assumption, the ratios of nominal variables return in the long run to initial steady state values, and since nominal wages cumulatively grew, the price level of services is permanently higher in the long run. Compared to the standard calibration with $\lambda_{EE} = .14$, the model that nearly shuts down on-the-job search with $\lambda_{EE} = 0.01$ has nearly 10 times greater cumulative increase in the services price level. A similar argument about the cumulative effect on the aggregate price level in panel (c) of Figure 6 can be made, as the relative price of the endowment good to wages returns to its initial level, and so cumulative headline inflation will also be equal to cumulative nominal wage growth.

While decomposing the shocks during COVID is beyond the scope of this paper, the timing displayed in Figure 6 accords well with the timing of inflation as described in Şahin (2023), where goods inflation rose first with supply shocks, and services inflation rose later. However, given our result that cost-of-living shocks pass through at most weakly into wages and propagates to services prices, our model would not explain the rise in services inflation during COVID as the result of propagation of the initial supply shock. Instead, our model would point to initial negative supply shocks that increase the cost of living relative to wages, and later demand shocks (or some other shock outside the model) driving both nominal wage growth and services inflation.

4.3 Robustness: Endogenous λ_{EE} , CRRA Utility, and Forward-Looking Workers

Throughout Section 4, we showed that pass-through of cost-of-living shocks to wages through a particular channel, namely inflation-indexed unemployment insurance, is quantitatively small. However, inflation-indexed unemployment insurance is not the only way to achieve positive passthrough of higher cost of living to wages. Here we briefly summarize two alternative model extensions, arguing that other plausible mechanisms for pass-through also deliver quantitatively small response of wages to higher cost of living. We also find that our results of small pass-through are robust to workers being forward looking, rather than myopic.

First, we consider an extension to the baseline model where the probability λ_{EE} at which work-

ers search on-the-job rises when real wages fall, considering evidence by Pilossoph and Ryngaert (2024), who find that inflation raises the rate at which workers search for job opportunities, and Pilossoph et al. (2023), who study how an endogenous change in search probability affects workers' wages in partial equilibrium. In Appendix A, we incorporate a simple extension where on-the-job search intensity λ_{EE} rises when real wages fall. This is a simplified way to capture that workers' on-the-job search effort may increase when real wages fall. We find that in general equilibrium, the pass-through of cost-of-living shocks to wages remains small, as greater on-the-job search intensity lowers labor market tightness, offsetting the positive wage impulse in partial equilibrium from greater on-the-job search.²⁸

Second, we generalize workers' preferences to allow for constant relative risk aversion (CRRA) utility in consumption, and we consider cases where workers are more risk averse than under log utility. This makes workers' mobility decision more sensitive to relative wage differences between two job offers when the cost of living is higher, encouraging firms to pay higher wages. We show in Supplementary Appendix G that this effect is quantitatively very small: under reasonable risk aversion levels, wages rise by less than 0.1% in response to shock that raises the aggregate price level by 2%.

Lastly, we analyze how cost-of-living shocks pass through to wages if workers are forward-looking. In this case, workers make decisions based on idiosyncratic utility draws and the expected future path of wages promised by the firm. Even when workers are forward looking, we similarly find that pass-through of cost-living-shocks to wages is quantitatively small. Details such as how we address the firm's commitment problem when faced with time-inconsistency²⁹ can be found in Supplementary Appendix F.

5 Comparison with Other Wage Setting Models

In this section, we show how our framework with wage posting and on-the-job search differs from two common labor blocks in macroeconomic models: the standard neoclassical labor supply model and the sticky wage model with differentiated labor and union wage setting as in Erceg et al. (2000). In all three models, pass-through of cost-of-living shocks ultimately depends on workers' labor supply responses, based on how the cost-of-living shock affects the relatively desirability of work and non-work (work hours and leisure in the neoclassical and union models, employment and

²⁸Birinci et al. (2022) discuss the same offsetting force, but in a setting with outside offer matching.

²⁹In particular, firms have an incentive to pay lower wages initially, promising higher wages in the future to attract workers. Since firms have those incentives in every period, we face a classic time-inconsistency problem. We provide a novel, tractable way to circumvent this issue by borrowing the *timeless* approach from the literature on optimal monetary policy under commitment. Recently, Rudanko (2023) considers this time-inconsistency problem in a model of firm wage setting and directed search.

unemployment in our model). In practice for the neoclassical and union wage setting models, it will matter greatly how much non-labor income is generated from the cost-of-living shock: if non-labor income goes up, this generates a wealth effect that decreases household labor supply, which results in higher nominal wage growth. In our setting, because unemployment is rarely a desirable outside option, wages are primarily determined by competition for already employed workers, so changes in the relative value of employment versus unemployment has little effect on wages.

Pass-Through with Neoclassical Labor Supply and Flexible Wages Consider an alternative model where firms still have sticky prices and a similar production environment (so the consumption and price aggregators (2) and (3) are unchanged), but the labor supply block is neoclassical. Households maximize

$$\sum_{t=0}^{\infty} \beta^t \left(\ln C_t \quad \frac{1}{1+\frac{1}{\nu}} N_t^{1+\frac{1}{\nu}} \right)$$

given the usual budget constraint

$$C_{t} = \frac{D_{t}}{P_{t}} - \frac{B_{t}}{P_{t}} + \frac{(1 + i_{t-1,t})B_{t-1}}{P_{t}} + \frac{W_{t}}{P_{t}}N_{t},$$

so that labor N_t is hired in a spot market and there is no unemployment. Under the assumption that the central bank stabilizes employment $N_t = N$ in response to a negative shock to the endowment good X_t at time t = 0, wages rise only if $\eta < 1$, i.e., the elasticity of substitution between X_t and Y_t is relatively weak. The sign and magnitude of the response in wages depends on the strength of *income* and *substitution* effects, governed by the elasticity of intertemporal substitution of consumption utility, and *wealth* effects when workers are endowed both with leisure and the good X_t . Because utility of consumption is log, income and substitution effects on labor supply from a lower real wage cancel out. Therefore wealth effects will determine the labor supply response of households.

The direction of those wealth effects are governed by η , the degree of substitution between X_t and Y_t . When $\eta < 1$, a cost of living shock as described above generates positive wealth effects, as prices for the endowment good rise more than one-for-one with the decline in quantity. This makes the household want to supply less labor, so firms must raise wages if N_t is to be stabilized at its pre-shock level.³⁰ If $\eta > 1$, the opposite logic will hold: workers will want to work more, and wages will actually fall in response to the shock. Thus, even in a perfectly competitive labor market, workers' wages can respond to a cost-of-living shock even when the shock has no direct effect on workers' productivity.³¹

 $^{^{30}}$ See Appendix C.1 for a proof.

³¹Appendix C.1 also analyzes the non-log case of a general consumption utility with the elasticity of intertemporal substitution σ possibly different from 1 in this neoclassical labor supply model.
Pass-Through with Differentiated Labor and Union Wage Setting Given the consumption and price aggregators (2) and (3), now let us assume that households now supply multiple types of labor; unions set wages for each type to maximize household utility subject to downward-sloping labor demand for each type from a "labor packer" which packages each labor type $\{N_t(i)\}$ into aggregate labor $N_t = \left(\int_0^1 N_t(i)^{\frac{1+\nu}{\nu}} di\right)^{\frac{\nu}{1+\nu}}$ which is purchased at wage W_t by services firms. Wages are sticky as in Erceg et al. (2000) and Galí et al. (2012) as unions only occasionally receive the chance to reset their wage. The aggregate household maximizes

$$\sum_{t=0}^{\infty} \beta^t \left(\ln C_t \quad \int_0^1 \frac{1}{1 + \frac{1}{\nu}} N_t(i)^{1 + \frac{1}{\nu}} di \right),$$

given the same budget constraint above. We can again prove that in response to a shock to X_t at t = 0, and assuming monetary policy stabilizes employment, there is pass-through only if $\eta < 1$. We prove this result in Appendix C.2. The similar result as in neoclassical labor block arises because income, substitution, and wealth effects operate in a similar way, with the only difference being that wages are sticky and unions mark up wages above the marginal rate of substitution.

Pass-through in Our Setting with Wage Posting and On-the-Job Search In the two common alternative models discussed in this section (a neoclassical labor block and a union wage setting model with differentiated labor), the response of wages to a cost-of-living shock is ultimately a function of workers' labor supply response to higher cost of living. The importance of workers' labor supply response is also present in our setting of wage posting and on-the-job search: in our baseline model, by construction we eliminate the role of wealth effects by fixing the ratio of consumption by employed and unemployed household members, and so workers' labor supply is unchanged. Thus, assuming monetary policy stabilizes employment N_t , we have no pass-through, unlike in the neoclassical model or union wage-setting model.³²

In our extension of inflation-adjusted unemployment benefits, workers are more likely to quit into unemployment and less likely to accept job offers from unemployment when the cost of living goes up: the discrete choice analogue to a decrease in labor supply. Our setting ultimately delivers *quantitatively* low pass-through because on-the-job search makes workers' extensive margin labor supply response nearly irrelevant in firms' wage setting decision, as illustrated in Section 4.2.3.

³²See Appendix C.3 for a proof.

6 Conclusion

This paper develops a tractable New Keynesian model where firms set both prices and wages, subject to nominal rigidities in price and wage setting, and workers search on the job. Calibrated to match U.S. data on worker flows, the model implies that the quits rate is the dominant predictor of nominal wage growth in the wage Phillips curve, consistent with recent U.S. data. We analyze the propagation of cost-of-living shocks in the model economy. Because firms set wages to avoid costly turnover, such shocks pass through to wages only to the extent that higher cost of living improves worker's outside options, such as competing jobs or unemployment, relative to their current job. As higher cost of living lowers real wages at all jobs evenly, and unemployment is rarely a credible outside option in the modern U.S. economy, we find that cost-of-living shocks have little to no effect on relative outside options and therefore wages.

While stylized, our model is consistent with a range of recent microeconomic evidence on how wages are determined, including the result in Jäger et al. (2020) that wages are insensitive to the flow value of unemployment benefits, and direct evidence on the preponderance of wage posting (Hall and Krueger, 2012; Lachowska et al., 2022; Di Addario et al., 2023). Admittedly, our simple model does abstract from the fact that there are a minority of unionized workers in the United States, and workers with automatic COLAs, for whom prices would pass through into wages. However, our results suggest that in a setting such as the United States where few workers operate under collective bargaining agreements with cost-of-living adjustments, and where firms' wage setting decision reflects competition for already-employed rather than for unemployed workers, the ability for most workers to reclaim real wages in response to a supply shock that raises their cost of living is limited. We conclude that there is little scope for supply-shock induced wage-price spirals specifically fueled by workers' ability to command higher nominal wages in response to higher nominal prices.

Appendix A Extension with Variable On-the-Job Search Intensity

Our baseline model features an exogenous, constant on-the-job search probability of λ_{EE} , which we calibrate to match U.S. data. However, it is possible that employed workers may respond to a pure cost-of-living shock by searching more intensely. Both Pilossoph and Ryngaert (2024) and Hajdini et al. (2023) provide evidence from household surveys that this is indeed the case.

Motivated by these findings, we solve a version of the model where λ_{EE} is assumed to rise along with inflation according to a reduced form, *ad hoc* relationship calibrated to match the results in Pilossoph and Ryngaert (2024). Specifically, we assume:

$$\lambda_{EE,t} = \lambda_{EE,0} \left(\frac{W_t}{P_t}\right)^{-m}$$

where $\lambda_{EE,0}$ is chosen to target the same steady-state value for λ_{EE} as in the baseline model, and m = 4 to match the fact that Pilossoph and Ryngaert (2024) find that in response to a one percentage point increase in inflation expectations (and thus a 1% decline in expect real wages), the probability that an employed worker searches on the job rises by 0.57 percentage points.¹ With a share of 14.9% of employed workers typically searching, this represents a (0.0057/0.149) \approx 4 percent increase in search probability, yielding an elasticity of search probability with respect to expected real wages of -4.

We then revisit the response in the model to a shock to the quantity of the endowment good X_t . Note that here, there are two contrasting effects of allowing for endogenous onthe-job search probability. In response to the inflationary shock, workers search more which induces firms to raise wages in order to retain workers (more searchers means more workers find jobs they prefer to their current jobs, due to the idiosyncratic preference shocks over workplaces). However, as separation rates rise, so do recruiting rates: with more searchers, tightness falls, and thereby firms can afford to lower wages and still recruit the same number of workers are before. Figure A.1 plots the impulse responses of headline inflation, wage growth, labor market tightness, and the separation rate to the shock to the quantity of endowment good X_t . We can observe the net effect of the shock is an extremely limited

¹See equation (3) and accompanying Table 3 of Pilossoph and Ryngaert (2024). Hajdini et al. (2023) estimate a much smaller effect in an information treatment RCT, finding that a one percentage point increase in inflation expectations raises the probability that a household respondent will "apply for a job(s) that pays more" by only about 0.11 percentage points.



Figure A.1: Impulse Response with Endogenous On-the-Job Search

Notes: This figure presents the effects of a decreased supply of the endowment good X_t under a nominal interest rate peg, i.e. the same experiment as in Figure 3, but comparing the benchmark case (solid blue line) of the main text to the case described in Section A where the job-to-job search probability increases along with the price level (dashed red line). In this second model, as on-the-job search rises, separations rise, inducing firms to raise wages to retain workers. But at the same time, tightness θ_t falls due to the increasing number of searchers, pushing firms to lower wages. The net effect is the modest increase in wages in the top right panel, so that overall there is very little pass-through from the aggregate price to wages even when the probability of on-the-job search rises in response to lower real wages. Note that the axes are in percent deviations, so the axis for wage growth is comparable to Figure 6.

pass-through from cost-of-living to wages: separations and wage growth rise, pushing firms to want to raise wages, but on the other hand tightness θ_t falls due to the increasing number of searchers, pushing firms to lower wages. In sum, wages respond positively but modestly in response to the cost of living shock.

References

Afrouzi, Hassan, Andres Blanco, Andres Drenik, and Erik Hurst, "A Theory of How Workers Keep Up with Inflation," 2024. NBER Working Paper No. 33233.

- Andolfatto, David and Serdar Birinci, "Is the Labor Market as Tight as It Seems?," Technical Report, Federal Reserve Bank of St. Louis 2022.
- Barnichon, Regis, "Building a Composite Help-Wanted Index," *Economics Letters*, 2010, *109* (3), 175–178.
- _ and Adam Hale Shapiro, "What's the Best Measure of Economic Slack?," FRBSF Economic Letter, 2022, 4, 1–05.
- **Bassier, Ihsaan, Arindrajit Dube, and Suresh Naidu**, "Monopsony in Movers: The Elasticity of Labor Supply to Firm Wage Policies," *Journal of Human Resources*, 2022, *57* (S), S50–s86.
- Benigno, Pierpaolo and Gauti B Eggertsson, "Slanted-L Phillips Curve," in "AEA Papers and Proceedings," Vol. 114 American Economic Association 2014 Broadway, Suite 305, Nashville, TN 37203 2024, pp. 84–89.
- **Bernanke, Ben and Olivier Blanchard**, "An Analysis of Pandemic-Era Inflation in 11 Economies," 2024. Hutchins Center Working Paper 91.
- Birinci, Serdar, Fatih Karahan, Yusuf Mercan, and Kurt See, "Labor Market Shocks and Monetary Policy," *FRB St. Louis Working Paper*, 2022, (2022-16).
- Blatter, Marc, Samuel Muehlemann, and Samuel Schenker, "The Costs of Hiring Skilled Workers," *European Economic Review*, 2012, *56* (1), 20–35.
- **Bloesch, Justin and Jacob Weber**, "Congestion in Onboarding Workers and Sticky R&D," *Available at SSRN 4465590*, 2023.
- _, Birthe Larsen, and Anders Yding, "Monopsony with Recruiting," 2024. Working Paper.
- Burdett, Kenneth and Dale T Mortensen, "Wage differentials, Employer Size, and Unemployment," *International Economic Review*, 1998, pp. 257–273.
- Chodorow-Reich, Gabriel and Loukas Karabarbounis, "The Cyclicality of the Opportunity Cost of Employment," *Journal of Political Economy*, 2016, *124* (6), 1563–1618.
- **Christiano, Lawrence J, Martin Eichenbaum, and Charles L Evans**, "Nominal Rigidities and the Dynamic Effects of a Shock to Monetary Policy," *Journal of political Economy*, 2005, *113* (1), 1–45.

- _, Martin S Eichenbaum, and Mathias Trabandt, "Unemployment and business cycles," *Econometrica*, 2016, 84 (4), 1523–1569.
- **Datta, Nikhil**, "The Measure of Monopsony: the Labour Supply Elasticity to the Firm and its Constituents," 2023. Working Paper.
- **Davis, Steven J, R Jason Faberman, and John Haltiwanger**, "Labor market flows in the cross section and over time," *Journal of monetary economics*, 2012, 59 (1), 1–18.
- de la Barrera i Bardalet, Marc, "Monopsony in New Keynesian Models," 2023. Working Paper.
- **Di Addario, Sabrina, Patrick Kline, Raffaele Saggio, and Mikkel Sølvsten**, "It Ain't Where You're From, It's Where You're At: Hiring Origins, Firm Heterogeneity, and Wages," *Journal of Econometrics*, 2023, *233* (2), 340–374.
- **Elsby, Michael W, Bart Hobijn, and Ayşegül Şahin**, "The Labor Market in the Great Recession," Technical Report, National Bureau of Economic Research 2010.
- Erceg, Christopher J, Dale W Henderson, and Andrew T Levin, "Optimal monetary policy with staggered wage and price contracts," *Journal of monetary Economics*, 2000, 46 (2), 281–313.
- Faberman, R Jason and Alejandro Justiniano, "Job Switching and Wage Growth," *Chicago Fed Letter*, 2015.
- **Faccini, Renato and Leonardo Melosi**, "Job-to-job Mobility and Inflation," *The Review of Economics and Statistics*, 2023, pp. 1–45.
- **Gagliardone, Luca and Mark Gertler**, "Oil Prices, Monetary Policy and Inflation Surges," 2023. NBER Working Paper 31263.
- Galí, Jordi, "The Return of the Wage Phillips Curve," *Journal of the European Economic Association*, 2011, 9 (3), 436–461.
- _, Frank Smets, and Rafael Wouters, "Unemployment in an Estimated New Keynesian model," NBER macroeconomics annual, 2012, 26 (1), 329–360.

- Hajdini, Ina, Edward S Knotek, John Leer, Mathieu Pedemonte, Robert W Rich, and Raphael Schoenle, "Low Passthrough from Inflation Expectations to Income Growth Expectations: Why People Dislike Inflation," 2023. Federal Reserve Bank of Cleveland Working Paper.
- Hall, Robert E and Alan B Krueger, "Evidence on the Incidence of Wage Posting, Wage Bargaining, and On-the-Job Search," *American Economic Journal: Macroeconomics*, 2012, 4 (4), 56–67.
- Harasztosi, Péter and Attila Lindner, "Who Pays for the Minimum Wage?," American Economic Review, 2019, 109 (8), 2693–2727.
- Heise, Sebastian, Fatih Karahan, and Ayşegül Şahin, "The Missing Inflation Puzzle: The Role of the Wage-Price Pass-Through," *Journal of Money, Credit and Banking*, 2022, 54 (S1), 7–51.
- _, Jeremy Pearce, and Jacob Weber, "A New Indicator of Labor Market Tightness for Predicting Wage Inflation," *Liberty Street Economics*, 2024.
- _, _, and _, "Wage Growth and Labor Market Tightness," *FRB of New York Staff Report*, 2024, (1128).
- Jäger, Simon, Benjamin Schoefer, Samuel Young, and Josef Zweimüller, "Wages and the Value of Nonemployment," *The Quarterly Journal of Economics*, 2020, *135* (4), 1905–1963.
- Lachowska, Marta, Alexandre Mas, Raffaele Saggio, and Stephen A Woodbury, "Wage Posting or Wage Bargaining? A Test using Dual Jobholders," *Journal of Labor Economics*, 2022, 40 (S1), S469–S493.
- **Lorenzoni, Guido and Iván Werning**, "Inflation is conflict," Technical Report, National Bureau of Economic Research 2023.
- _ and _ , "Wage-Price Spirals," Technical Report, National Bureau of Economic Research 2023.
- Michaillat, Pascal and Emmanuel Saez, "Resolving New Keynesian Anomalies with Wealth in the Utility Function," *Review of Economics and Statistics*, 2021, *103* (2), 197–215.

 $_$ and $_$, "u*= uv," Technical Report, National Bureau of Economic Research 2022.

- Moscarini, Giuseppe and Fabien Postel-Vinay, "Wage Posting and Business Cycles," American Economic Review, Papers and Proceedings, 2016, 106 (5), 208–213.
- and _, "Wage Posting and Business Cycles: a Quantitative Exploration," *Review of Economic Dynamics*, 2016, (19), 135–160.
- and __, "The Relative Power of Employment-to-Employment Reallocation and Unemployment Exits in Predicting Wage Growth," *American Economic Review*, 2017, *107* (5), 364–368.
- _ and _ , "The Job Ladder: Inflation vs. Reallocation," Technical Report, National Bureau of Economic Research 2023.
- Pilossoph, Laura and Jane M. Ryngaert, "Job Search, Wages, and Inflation," *Working Paper*, 2024.
- _, Jane M Ryngaert, and Jesse Wedewer, "Job Search, Raises, and Inflation," 2023. Working Paper.
- **Pissarides, Christopher A**, "The Unemployment Volatility Puzzle: Is Wage Stickiness the Answer?," *Econometrica*, 2009, 77 (5), 1339–1369.
- Qiu, Xincheng, "Vacant Jobs," 2022. Working Paper.
- Rotemberg, Julio J., "Sticky Prices in the United States," *Journal of Political Economy*, 1982, *90* (6), 1187–1211.
- Rudanko, Leena, "Firm Wages in a Frictional Labor Market," *American Economic Journal: Macroeconomics*, 2023, *15* (1), 517–550.
- Şahin, Ayşegül, "Comment of "Wage Price Spirals"," Brookings Papers in Economic Activity, 2023, pp. 364–368.
- Shierholz, Heidi, Margaret Poydock, and Celine McNicholas, "Unionization increased by 200,000 in 2022: Tens of millions more wanted to join a union, but couldn't," 2023.
- Smets, Frank and Rafael Wouters, "Shocks and Frictions in US Business Cycles: A Bayesian DSGE Approach," *American economic review*, 2007, *97* (3), 586–606.

Williams, John C., "All About Data," Speech 98995, Federal Reserve Bank of New York October 2024.

Online Appendix for Firm Wage Setting and On-the-Job Search Limit Wage-Price Spirals

JUSTIN BLOESCH SEUNG JOO LEE JACOB P. WEBER

Appendix B Derivations and Proofs

B.1 Worker's Problem with Utility From Leisure

This section reviews the worker's problem, deriving the probability a worker chooses a particular job j over outside offer k or unemployment. We then show that allowing for utility from leisure, as well as consumption, will not generally overturn the result that the price level does not affect the worker's optimal choice unless the elasticity of substitution between leisure and consumption is different from one.

Discrete choice with Type-1 extreme value preference draws Suppose worker *i* in state *j* (which could be working at firm *j*, for example), gets utility U(ijt) plus a draw ι_{ijt} that is distributed type-1 extreme value:

$$\mathcal{V}_t(i,j) = \mathcal{U}(ijt) + \iota_{ijt}$$

Let ι_{ijt} have scale parameter $\frac{1}{\gamma}$. Then given options two states j and k, the probability that the worker chooses j is

$$\frac{\exp(\gamma \mathcal{U}(ijt))}{\exp\left(\gamma \mathcal{U}(ijt)\right) + \exp\left(\gamma \mathcal{U}(ikt)\right)}$$

Suppose now that utility \mathcal{U} is a function of log consumption: $\mathcal{U}(ijt) = \log(C_t(i, j))$. This is the case in the main text. Then the probability of choosing j is

$$\frac{C_t(i,j)^{\gamma}}{C_t(i,j)^{\gamma} + C_t(i,k)^{\gamma}}.$$

Case with a more general utility function Consider now the more general form

$$\mathcal{V}(ijt) = \log\left(U\left(C_t(i,j), \ell_t(i,j)\right)\right) + \iota_{ijt}$$

where $\ell_t(i, j)$ is the leisure *i* gets in state *j* at time *t*, which nests the above case. For simplicity, denote utility while unemployed by $U(C_t(i, u), \ell_t(i, u))$, and while employed by $U(C_t(i, e), \ell_t(i, e))$; then the probability of an unemployed worker taking a job when matched is now:

$$\frac{1}{1 + \left(\frac{U(C_t(i,u),\ell_t(i,u))}{U(C_t(i,e),\ell_t(i,e))}\right)^{\gamma}}$$
(B.1)

Proposition 1 In partial equilibrium (i.e. holding all other equilibrium prices and quantities fixed) the probability that an unemployed worker takes a job in our general setting, (B.1), is invariant to changes in the price level P_t if and only if $\frac{\partial}{\partial P_t} \frac{U(C_t(i,e),\ell_t(i,e))}{U(C_t(i,e),\ell_t(i,e))} = 0.$

CES preference To make progress, consider the case with CES preferences:

$$U = \left(aC^{\frac{\rho-1}{\rho}} + (1 \quad a)\ell^{\frac{\rho-1}{\rho}}\right)^{\frac{\rho}{\rho-1}}$$

where ρ is the elasticity of substitution. We write $U(C, \ell) = U(\frac{I}{P}, \ell)$, imposing $C = \frac{I}{P}$ for both types, who differ only in the nominal spending I (i.e. I_e for employed and I_u for unemployed).² Noting constant returns to scale (CRS) yields

$$\frac{U\left(\frac{I_t(i,u)}{P_t},\ell_t(i,u)\right)}{U\left(\frac{I_t(i,e)}{P_t},\ell_t(i,e)\right)} = \frac{U\left(I_t(i,u),P_t\ell_t(i,u)\right)}{U\left(I_t(i,e),P_t\ell_t(i,e)\right)},$$

and using the property of CES functions: $\frac{\partial}{\partial P}U(I, P\ell) = (1 \quad a)U(\cdot)^{\frac{1}{\rho}}(P\ell)^{-\frac{1}{\rho}}l$, we can show:

$$\frac{\partial}{\partial P_{t}} \frac{U\left(\frac{I_{t}(i,u)}{P_{t}}, \ell_{t}(i,u)\right)}{U\left(\frac{I_{t}(i,e)}{P_{t}}, \ell_{t}(i,e)\right)} = \frac{(1-a)P_{t}^{-\frac{1}{\rho}}}{U(I_{t}(i,e), P_{t}\ell_{t}(i,e))} \\ \cdot \left[U(I_{t}(i,u), P_{t}\ell_{t}(i,u))^{\frac{1}{\rho}}\ell_{t}(i,u)^{1-\frac{1}{\rho}} \\ U(I_{t}(i,e), P_{t}\ell_{t}(i,e))^{\frac{1}{\rho}}\ell_{t}(i,e)^{1-\frac{1}{\rho}} \frac{U(I_{t}(i,u), P_{t}\ell_{t}(i,u))}{U(I_{t}(i,e), P_{t}\ell_{t}(i,e))}\right]$$

which becomes 0 when $\rho \rightarrow 1$, i.e., the Cobb-Douglas case. Therefore, under the unit elasticity of substitution between consumption and leisure, Proposition 1 still holds.

²Here, the result does not depend on the case where we impose a tax-and-transfer scheme to keep $I_e/I_u = C_e/C_u$ constant over the business cycle as in Section 3.2.

B.2 Firm's Problem and Derivation of the wage Phillips curve in (7)

The firm's problem is:³

$$\max_{\substack{\{P_{y,t}^{j}\},\{N_{jt}\}\\\{W_{jt}\},\{V_{j,t}\}}} \sum_{t=0}^{\infty} \left(\frac{1}{1+\rho}\right)^{t} \left(P_{y,t}^{j}Y_{t}^{j} \quad W_{jt}N_{jt} \quad c\left(\frac{V_{j,t}}{N_{j,t-1}}\right)^{\chi}V_{j,t}W_{t} \quad \frac{\psi}{2}\left(\frac{P_{y,t}^{j}}{P_{y,t-1}^{j}} \quad 1\right)^{2}Y_{t}^{j}P_{y,t}^{j} \\
\frac{\psi^{w}}{2}\left(\frac{W_{jt}}{W_{j,t-1}} \quad 1\right)^{2}W_{jt}N_{jt}\right)$$
(B.2)

subject to

 $N_{jt} = (1 \quad S_t(W_{jt}))N_{j,t-1} + R_t(W_{jt})V_{j,t}.$ (B.3)

Output is produced with labor with linear production: $Y_t^j = A_t^j N_{jt}$,⁴ and Dixit-Stiglitz demand $\frac{Y_t^j}{Y_t} = \left(\frac{P_{y,t}^j}{P_{y,t}}\right)^{\epsilon}$, hence $N_{jt} = \left(\frac{P_{y,t}^j}{P_{y,t}}\right)^{\epsilon} \frac{Y_t}{A_t^j}$ with $\epsilon > 1$. The Lagrangian is:

$$\mathcal{L} = \sum_{t=0}^{\infty} \left(\frac{1}{1+\rho}\right)^{t} \left((P_{y,t}^{j})^{1-\epsilon} (P_{y,t})^{\epsilon} Y_{t} - W_{jt} \left(\frac{P_{y,t}^{j}}{P_{y,t}}\right)^{-\epsilon} \frac{Y_{t}}{A_{t}^{j}} - c (V_{j,t})^{1+\chi} \left(\frac{P_{y,t-1}^{j}}{P_{y,t-1}}\right)^{\epsilon\chi} \left(\frac{Y_{t-1}}{A_{t-1}^{j}}\right)^{-\chi} W_{t} - \frac{\psi}{2} \left(\frac{P_{y,t}^{j}}{P_{y,t-1}^{j}} - 1\right)^{2} \left(P_{y,t}^{j}\right)^{1-\epsilon} (P_{t})^{\epsilon} Y_{t} - \frac{\psi^{w}}{2} \left(\frac{W_{jt}}{W_{j,t-1}} - 1\right)^{2} W_{jt} \left(\frac{P_{y,t}^{j}}{P_{y,t}}\right)^{-\epsilon} \frac{Y_{t}}{A_{t}^{j}} + \lambda_{t}^{j} \left[-\left(\frac{P_{y,t}^{j}}{P_{y,t}}\right)^{-\epsilon} \frac{Y_{t}}{A_{t}^{j}} + V_{j,t} R_{t}(W_{jt}) + (1 - S_{t}(W_{jt})) \left(\frac{P_{y,t-1}^{j}}{P_{y,t-1}}\right)^{-\epsilon} \frac{Y_{t-1}}{A_{t-1}^{j}} \right] \right).$$

The first order conditions are:

$$\mathcal{L}_{W_{jt}} = \left(\frac{P_{y,t}^{j}}{P_{y,t}}\right)^{\epsilon} \frac{Y_{t}}{A_{t}^{j}} + \lambda_{t}^{j} \left(V_{j,t}R_{t}'(W_{jt}) - S_{t}'(W_{jt})\left(\frac{P_{y,t-1}^{j}}{P_{y,t-1}}\right)^{\epsilon} \frac{Y_{t-1}}{A_{t-1}^{j}}\right)$$
$$- \frac{\psi^{w}}{2} \left(\frac{W_{jt}}{W_{j,t-1}} - 1\right)^{2} \left(\frac{P_{y,t}^{j}}{P_{y,t}}\right)^{\epsilon} \frac{Y_{t}}{A_{t}^{j}} - \psi^{w} \left(\frac{W_{t}^{j}}{W_{j,t-1}} - 1\right) \frac{1}{W_{j,t-1}} W_{t} N_{jt}$$
$$+ \frac{1}{1+\rho} \psi^{w} \left(\frac{W_{j,t+1}}{W_{jt}} - 1\right) \frac{W_{t+1}^{j}}{(W_{jt})^{2}} W_{j,t+1} N_{j,t+1} = 0.$$
(B.4)

³Note that here we assume that vacancy costs are denominated in labor; see Bloesch and Weber (2023) for microfoundations and Appendix C.3 for additional implications. We also use the aggregate wage W_t rather than the firm-specific wage W_{jt} to simplify the firm's wage setting problem.

⁴Instead of production function $Y_t^j = N_t^j$ assumed in Section 3, we assume a linear technology $Y_t^j = A_t^j N_t^j$ in the derivation. Later we will assume a symmetric equilibrium with $A_t^j = A_t$ for $\forall j$.

and

$$\mathcal{L}_{V_{j,t}} = c(1+\chi)(V_{j,t})^{\chi} \left(\frac{P_{y,t-1}^{j}}{P_{y,t-1}}\right)^{\epsilon\chi} \left(\frac{Y_{t-1}}{A_{t-1}^{j}}\right)^{-\chi} W_{t} + \lambda_{t}^{j} R_{t}(W_{jt}) = 0, \quad (B.5)$$

and

$$\begin{aligned} \mathcal{L}_{P_{y,t}^{j}} = & (1 \quad \epsilon) \left(\frac{P_{y,t}^{j}}{P_{y,t}}\right)^{-\epsilon} Y_{t} + \epsilon W_{jt} \left(P_{y,t}^{j}\right)^{-1} \left(\frac{P_{y,t}^{j}}{P_{y,t}}\right)^{-\epsilon} \frac{Y_{t}}{A_{t}^{j}} \\ & \frac{c\epsilon}{1+\rho} \chi \left(V_{j,t+1}\right)^{1+\chi} \left(P_{y,t}^{j}\right)^{-1} \left(\frac{P_{y,t}^{j}}{P_{y,t}}\right)^{\epsilon} \left(\frac{Y_{t}}{A_{t}^{j}}\right)^{-\chi} W_{t+1} \\ & \psi \left(\frac{P_{y,t}^{j}}{P_{y,t-1}^{j}} - 1\right) \frac{1}{P_{y,t-1}^{j}} (P_{y,t}^{j})^{1-\epsilon} P_{y,t}^{\epsilon} Y_{t} - (1-\epsilon) \frac{\psi}{2} \left(\frac{P_{y,t}^{j}}{P_{y,t-1}^{j}} - 1\right)^{2} \left(\frac{P_{y,t}^{j}}{P_{y,t}}\right)^{-\epsilon} Y_{t} \\ & + \frac{1}{1+\rho} \psi \left(\frac{P_{t+1}^{j}}{P_{t}^{j}} - 1\right) \frac{P_{t+1}^{j}}{(P_{y,t}^{j})^{2}} \left(P_{t+1}^{j}\right)^{1-\epsilon} P_{t+1}^{\epsilon} Y_{t+1} \\ & + \epsilon \frac{\psi^{w}}{2} \left(\frac{W_{jt}}{W_{j,t-1}} - 1\right)^{2} \frac{W_{jt}}{P_{y,t}} \left(\frac{P_{y,t}^{j}}{P_{y,t}}\right)^{-\epsilon} \frac{1}{X_{t}^{j}} \\ & + \lambda_{t}^{j} \epsilon \left(P_{y,t}^{j}\right)^{-1} \left(\frac{P_{y,t}^{j}}{P_{y,t}}\right)^{-\epsilon} \frac{Y_{t}}{A_{t}^{j}} - \frac{1}{1+\rho} \lambda_{t+1}^{j} \epsilon (1 - S_{t}(W_{j,t+1})) (P_{y,t}^{j})^{-1} \left(\frac{P_{y,t}^{j}}{P_{y,t}}\right)^{-\epsilon} \frac{Y_{t}}{A_{t}^{j}} = 0 \\ & (\mathbf{B.6})
\end{aligned}$$

Equilibrium We focus on one particular equilibrium where $P_{y,t}^j = P_{y,t}$, $V_{j,t} = V_t$, $W_{jt} = W_t$, $A_t^j = A_t \forall j$. Then we can summarize the above equations as follows:

FOC on Wages in (B.4) : If we define the aggregate wage inflation $\Pi_t^w = \frac{W_t}{W_{t-1}}$ and approximate with $(\Pi_t^w = 1)^2 \simeq 0$, equation (B.4) becomes

$$N_{t} + \frac{\lambda_{t}}{P_{y,t}} \left(\frac{W_{t}}{P_{y,t}}\right)^{-1} \left(V_{t}R_{t}'(W_{t})W_{t} - N_{t-1}S_{t}'(W_{t})W_{t}\right) - \psi^{w}\left(\Pi_{t}^{w} - 1\right)\Pi_{t}^{w}N_{t} + \frac{1}{1+\rho}\psi^{w}\left(\Pi_{t+1}^{w} - 1\right)\left(\Pi_{t+1}^{w}\right)^{2}N_{t+1} = 0.$$
(B.7)

This is important so that we have the real wage and real Lagrange multiplier, i.e., $\frac{\lambda_t}{P_{y,t}}$ in our equilibrium equations.

FOC on vacancies in (**B.5**) :

$$c(1+\chi)\left(\frac{V_t}{N_{t-1}}\right)^{\chi}\frac{W_t}{P_{y,t}} + \frac{\lambda_t}{P_{y,t}}R_t(W_t) = 0.$$
(B.8)

Plugging in (B.8) into (B.7) and rearranging gives:

$$\begin{split} N_{t} + \psi^{w} \left(\Pi_{t}^{w} - 1\right) \Pi_{t}^{w} N_{t} &= c(1+\chi) \left(\frac{V_{t}}{N_{t-1}}\right)^{\chi} \frac{1}{R(W_{t})} \left(V_{t} R_{t}'(W_{t}) W_{t} - N_{t-1} S_{t}'(W_{t}) W_{t}\right) \\ &+ \frac{1}{1+\rho} \psi^{w} \left(\Pi_{t+1}^{w} - 1\right) \left(\Pi_{t+1}^{w}\right)^{2} N_{t+1} \\ &= c(1+\chi) \left(\frac{V_{t}}{N_{t-1}}\right)^{\chi} \left(V_{t} \frac{R_{t}'(W_{t}) W_{t}}{R_{t}(W_{t})} - N_{t-1} \frac{S_{t}(W_{t})}{R_{t}(W_{t})} \frac{S_{t}'(W_{t}) W_{t}}{S_{t}(W_{t})} \right) \\ &+ \frac{1}{1+\rho} \psi^{w} \left(\Pi_{t+1}^{w} - 1\right) \left(\Pi_{t+1}^{w}\right)^{2} N_{t+1}. \end{split}$$

Dividing by N_t in both sides, we obtain wage Phillips curve in our model:

$$\psi^{w} (\Pi_{t}^{w} = 1) \Pi_{t}^{w} + 1 = c(1+\chi) \left(\frac{V_{t}}{N_{t-1}}\right)^{\chi} \left(\frac{V_{t}}{N_{t}}\varepsilon_{R,W_{t}} - \frac{N_{t-1}}{N_{t}}\frac{S_{t}(W_{t})}{R_{t}(W_{t})}\varepsilon_{S,W_{t}}\right) + \frac{1}{1+\rho}\psi^{w} (\Pi_{t+1}^{w} - 1) (\Pi_{t+1}^{w})^{2}\frac{N_{t+1}}{N_{t}}.$$
(B.9)

FOC on pricing in (B.6) :

$$(1 \quad \epsilon) + \epsilon \frac{W_t}{P_{y,t}} A_t^{-1} \quad \frac{1}{1+\rho} c \epsilon \chi \left(V_{t+1}\right)^{1+\chi} \frac{P_{y,t+1}}{P_{y,t}} \frac{W_{t+1}}{P_{y,t+1}} Y_t^{-1-\chi} A_t^{\chi} + \epsilon \frac{\psi^w}{2} \underbrace{\left(\prod_{t=1}^w 1\right)^2}_{\simeq 0} \frac{W_t}{P_{y,t}} N_t \\ \psi \left(\frac{P_{y,t}}{P_{y,t-1}} - 1\right) \frac{P_{y,t}}{P_{y,t-1}} \quad (1 - \epsilon) \frac{\psi}{2} \underbrace{\left(\frac{P_{y,t}}{P_{y,t-1}} - 1\right)^2}_{\simeq 0} + \frac{1}{1+\rho} \psi \left(\frac{P_{y,t+1}}{P_{y,t}} - 1\right) \left(\frac{P_{y,t+1}}{P_{y,t}}\right)^2 \frac{Y_{t+1}}{Y_t} \\ + \frac{\lambda_t}{P_{y,t}} \epsilon A_t^{-1} - \frac{1}{1+\rho} \frac{\lambda_{t+1}}{P_{y,t+1}} \frac{P_{y,t+1}}{P_{y,t}} \epsilon (1 - S_t(W_{t+1})) A_t^{-1} = 0,$$

$$(B.10)$$

where we approximate $(\frac{P_{y,t+1}}{P_{y,t}} - 1)^2 \approx 0$ as above. If we define the service inflation $\frac{P_{y,t}}{P_{y,t-1}} = \prod_{Y,t}$, (B.10) can be written as

$$\frac{\psi}{\epsilon} (\Pi_{Y,t} \quad 1)\Pi_{Y,t}Y_t + \frac{\epsilon}{\epsilon} \frac{1}{\epsilon}Y_t = \frac{W_t}{P_{y,t}}N_t + \frac{1}{1+\rho}\frac{\psi}{\epsilon} (\Pi_{Y,t+1} \quad 1)\Pi_{Y,t+1}^2Y_{t+1} + N_t \left(\frac{c\chi}{1+\rho}\left(\frac{V_{t+1}}{N_t}\right)^{1+\chi}\Pi_{Y,t+1}\frac{W_{t+1}}{P_{y,t+1}} + \frac{\lambda_t}{P_{y,t}} \quad \frac{1}{1+\rho}\frac{\lambda_{t+1}}{P_{y,t+1}}\Pi_{Y,t+1}(1 \quad S_t(W_{t+1}))\right)$$
(B.11)

which is our price Phillips curve.

B.3 Linearized Wage Phillips Curve

A Log-Linear Wage Phillips Curve We log-linearize the wage Phillips curve in (7), except we leave in the second order term, $\frac{\psi^w}{2} (\Pi_t^w - 1)^2$, which we dropped when we derive (7).

$$\frac{\psi^{w}}{2} (\Pi_{t}^{w} - 1)^{2} + \psi^{w} (\Pi_{t}^{w} - 1) \Pi_{t}^{w} + 1 = c(1 + \chi) \left(\frac{V_{t}}{N_{t-1}}\right)^{\chi} \left(\frac{V_{t}}{N_{t}}\varepsilon_{R,W_{t}} - \frac{N_{t-1}}{N_{t}}\frac{S_{t}(W_{t})}{R_{t}(W_{t})}\varepsilon_{S,W_{t}}\right) \\ + \frac{1}{1 + \rho}\psi^{w} (\Pi_{t+1}^{w} - 1) (\Pi_{t+1}^{w})^{2}\frac{N_{t+1}}{N_{t}}.$$
(B.12)

Equation (B.12) would hold even if we added other factors of production (e.g., we could have Cobb-Douglass production with capital or some other inputs including oil), and is unaffected by the presence of price rigidities (e.g., if we had flexible or price rigidity à la Rotemberg (1982), (B.12) would be the same). To ease interpretation, we rewrite this using $T_t \equiv \frac{V_t}{N_{t-1}}$ and $g_t \equiv \frac{N_t}{N_{t-1}}$:

$$0 = \frac{\psi^{w}}{2} (\Pi_{t}^{w} - 1)^{2} + \psi^{w} (\Pi_{t}^{w} - 1) \Pi_{t}^{w} + 1 - c(1 + \chi) T_{t}^{\chi} g_{t}^{-1} \left(T_{t} \varepsilon_{R,W_{t}} - \frac{S_{t}(W_{t})}{R_{t}(W_{t})} \varepsilon_{S,W_{t}} \right)$$
$$\frac{1}{1 + \rho} \psi^{w} (\Pi_{t+1}^{w} - 1) (\Pi_{t+1}^{w})^{2} g_{t+1}.$$
(B.13)

We can suppress the dependence of $S_t(\cdot)$ and $R_t(\cdot)$ on W_t (since we know that in equilibrium, S_t and R_t are not functions of the aggregate wage W_t): we rewrite (B.13) as:

$$0 = F\left(\ln(\Pi_t^w), \ln(\Pi_{t+1}^w), \ln(S_t), \ln(R_t), \ln(\varepsilon_{R,t}), \ln(\varepsilon_{S,t}), \ln(T_t), \ln(g_t), \ln(g_{t+1})\right),$$

and take a linear approximation around a zero wage-inflation steady state with variables $\ln(\Pi_t^w)$, $\ln(\Pi_{t+1}^w)$, $\ln(S_t)$, $\ln(R_t)$, $\ln(\varepsilon_{R,t})$, $\ln(\varepsilon_{S,t})$, $\ln(T_t)$, $\ln(g_t)$, and $\ln(g_{t+1})$. We first calculate derivatives of $F(\cdot)$ with respect to each variable as follows:

$$\begin{split} F_{\ln(\Pi_{t}^{w})} &= \psi^{w} \Pi_{t}^{w} \left(2(\Pi_{t}^{w} \quad 1) + \Pi_{t}^{w} \right) \\ F_{\ln(\Pi_{t+1}^{w})} &= \frac{\psi^{w} g_{t+1}}{1+\rho} \left(\Pi_{t+1}^{w} (\Pi_{t+1}^{w})^{2} + (\Pi_{t+1}^{w} \quad 1) 2(\Pi_{t+1}^{w})^{2} \right) \\ F_{\ln(S_{t})} &= c(1+\chi) T_{t}^{\chi} g_{t}^{-1} \frac{S_{t}}{R_{t}} \varepsilon_{S,t}, \quad F_{\ln(R_{t})} &= c(1+\chi) T_{t}^{\chi} g_{t}^{-1} \frac{S_{t}}{R_{t}} \varepsilon_{S,t} \\ F_{\ln(\varepsilon_{R,t})} &= c(1+\chi) T_{t}^{\chi+1} g_{t}^{-1} \varepsilon_{R,t}, \quad F_{\ln(\varepsilon_{S,t})} = c(1+\chi) T_{t}^{\chi} g_{t}^{-1} \frac{S_{t}}{R_{t}} \varepsilon_{S,t} \\ F_{ln(T_{t})} &= c(1+\chi) g_{t}^{-1} \left((1+\chi) T_{t}^{\chi+1} \varepsilon_{R,t} \quad \chi T_{t}^{\chi} \frac{S_{t}}{R_{t}} \varepsilon_{S,t} \right) \\ F_{\ln(g_{t})} &= c(1+\chi) T_{t}^{\chi} g_{t}^{-1} \left(T_{t} \varepsilon_{R,t} \quad \frac{S_{t}}{R_{t}} \varepsilon_{S,t} \right) \\ F_{\ln(g_{t+1})} &= \frac{1}{1+\rho} \psi^{w} \left(\Pi_{t+1}^{w} - 1 \right) (\Pi_{t+1}^{w})^{2} g_{t+1}, \end{split}$$

which at the steady state with zero wage inflation can be written as

$$\begin{split} F_{\ln(\Pi_t^w)} &= \psi^w, \ F_{\ln(\Pi_{t+1}^w)} = -\frac{\psi^w}{1+\rho} \\ F_{\ln(S_t)} &= c(1+\chi)T^{\chi}g^{-1}\frac{S}{R}\varepsilon_S = c(1+\chi)\frac{T^{\chi+1}}{g}\varepsilon_S \\ F_{\ln(R_t)} &= -c(1+\chi)T^{\chi}g^{-1}\frac{S_t}{R_t}\varepsilon_{S,t} = -c(1+\chi)\frac{T^{\chi+1}}{g}\varepsilon_S \\ F_{\ln(\varepsilon_{R,t})} &= -c(1+\chi)\frac{T^{\chi+1}}{g}\varepsilon_R \\ F_{\ln(\varepsilon_{S,t})} &= -c(1+\chi)\frac{T^{\chi+1}}{g}\varepsilon_S \\ F_{\ln(T_t)} &= -c(1+\chi)\frac{T^{\chi+1}}{g}((1+\chi)\varepsilon_R - \chi\varepsilon_S) \\ F_{\ln(g_t)} &= -c(1+\chi)\frac{T^{\chi+1}}{g}(\varepsilon_R - \varepsilon_S) > 0 \\ F_{\ln(g_{t+1})} &= 0 \end{split}$$

where we made use of the fact that $T = \frac{V}{N} = \frac{S}{R}$ in steady state, which we obtain from (H.2). We can see that the assumption above that $\frac{\psi^w}{2} (\Pi_t^w - 1)^2 \approx 0$ is correct in the sense that it drops out in our first-order approximation. We also find that in a zero inflation steady-state, there is no role for expectations of future employment growth in our first-order approximation.

Let the magenta terms above be collected as $\kappa \equiv c(1+\chi)\frac{T^{\chi+1}}{g}$. Then the first order approximation of $F(\cdot)$ around its steady state is given by⁵

$$0 = \psi^{w} \check{\Pi}_{t}^{w} \quad \frac{\psi^{w}}{1+\rho} \check{\Pi}_{t+1}^{w} + \kappa \varepsilon_{S} \left(\check{S}_{t} \quad \check{R}_{t} \right) + \kappa \left(\varepsilon_{S} \check{\varepsilon}_{S,t} \quad \varepsilon_{R} \check{\varepsilon}_{R,t} \right)$$
$$+ \kappa (\chi \varepsilon_{S} \quad (1+\chi) \varepsilon_{R}) \check{T}_{t} + \kappa (\varepsilon_{R} \quad \varepsilon_{S}) \check{g}_{t},$$

which we can rewrite as

$$\check{\Pi}_{t}^{w} = \underbrace{\frac{\kappa \varepsilon_{S}}{\psi^{w}} \left(\check{S}_{t} \quad \check{R}_{t}\right)}_{\text{tree Labor Market "Tightness" Terms}} \underbrace{\frac{\kappa (\varepsilon_{R} \quad \varepsilon_{S})}{\psi^{w}} \check{g}_{t}}_{\text{Three Labor Market "Tightness" Terms}} + \underbrace{\frac{\kappa (\varepsilon_{R} \quad \varepsilon_{S})}{\psi^{w}} \check{g}_{t}}_{\text{Employment Growth}} \underbrace{\frac{1}{1 + \rho} \check{\Pi}_{t+1}^{w}}_{\text{Expectations}} \tag{B.14}$$

Note that the law of motion for employment in (H.2) that the firm faces implies:

$$g_t = \begin{pmatrix} 1 & S_t \end{pmatrix} + R_t T_t \tag{B.15}$$

Log linearizing (B.15) yields:

$$\frac{1}{S}\check{g}_t = \check{R}_t + \check{T}_t \quad \check{S}_t,$$
$$\check{S}_t \quad \check{R}_t = \check{T}_t \quad \frac{1}{S}\check{g}_t \tag{B.16}$$

which leads to

Plugging (B.16) into the log-linear wage Phillips curve in (B.14) and assuming $g_{=1}$, we obtain

$$\check{\Pi}_{t}^{w} = \underbrace{\frac{\kappa\left(\begin{array}{cc}\varepsilon_{R} + \frac{1+S}{S}\varepsilon_{S}\right)}{\psi^{w}}\check{g}_{t}}_{\text{Employment Growth}} \check{g}_{t} + \underbrace{\frac{\kappa}{\psi^{w}}\left(\varepsilon_{R}\check{\varepsilon}_{R,t} - \varepsilon_{S}\check{\varepsilon}_{S,t}\right)}_{\text{Two Labor Market "Tightness" Terms}} + \underbrace{\frac{\kappa(1+\chi)(\varepsilon_{R} - \varepsilon_{S})}{\psi^{w}}\check{T}_{t}}_{\text{Expectations}} + \underbrace{\frac{1}{1+\rho}\check{\Pi}_{t+1}^{w}}_{\text{Expectations}}.$$
(B.17)

where $\kappa \equiv c(1 + \chi)T^{\chi+1}$. From (B.17), we observe that stronger monopsony, i.e., a lower $\varepsilon_R = \varepsilon_S$, flattens the wage Phillips curve, as documented in de la Barrera i Bardalet (2023).

⁵We let $\check{X}_t \equiv \ln X_t$ $\ln X$ for any X_t . If $X_t < 0$, then we let $\check{X}_t \equiv \frac{X_t - X}{X}$.

We summarize this in the following Proposition 2.

Proposition 2 The wage Phillips curve in (B.17) becomes flatter as the recruiting elasticity net of the separation elasticity, $\varepsilon_R = \varepsilon_S$, falls.

Further Simplification Plugging (B.15) into (B.14) yields:

$$\check{\Pi}_{t}^{w} = \frac{\kappa}{\psi^{w}} \begin{pmatrix} S(\varepsilon_{R} & \varepsilon_{S}) \left(\check{T}_{t} + \check{R}_{t} & \check{S}_{t}\right) & \varepsilon_{S}\left(\check{S}_{t} & \check{R}_{t}\right) + (\varepsilon_{R}\check{\varepsilon}_{R,t} & \varepsilon_{S}\check{\varepsilon}_{S,t}) + (\varepsilon_{R} + \chi(\varepsilon_{R} & \varepsilon_{S}))\check{T}_{t} \\ &+ \frac{1}{1+\rho}\check{\Pi}_{t+1}^{w},$$
(B.18)

which leads to⁶

$$\check{\Pi}_{t}^{w} = \frac{\kappa}{\psi^{w}} \left[\underbrace{\left(\begin{array}{cc} \varepsilon_{S} + S(\varepsilon_{R} & \varepsilon_{S}) \right)}_{>0} \left(\check{S}_{t} & \check{R}_{t} \right) + \underbrace{\left(\varepsilon_{R} + \left(\chi & S \right) \left(\varepsilon_{R} & \varepsilon_{S} \right) \right)}_{>0} \check{T}_{t} + \left(\varepsilon_{R} \check{\varepsilon}_{R,t} & \varepsilon_{S} \check{\varepsilon}_{S,t} \right) \right. \\ \left. + \frac{1}{1 + \rho} \check{\Pi}_{t+1}^{w}.$$
(B.19)

B.3.1 Reduced Form Log-Linear Wage Phillips Curve in Quits and Unemployment

Estimating this regression (E.2) in the data via OLS shows that the regression puts more weight on quits than on unemployment, as documented in Table 1. As explained in Section 2, the regression yields a surprising empirical result for the sign of the coefficient on unemployment: replacing vacancies with quits, the sign on unemployment flips, and becomes positive. In other words, holding quits constant, a higher unemployment rate is correlated with *higher* wage growth!

Intriguingly, our model's benchmark calibration actually captures this result: when $\chi = 1$, i.e., firms' vacancy costs are convex, we find a much larger coefficient on quits than on unemployment, where the coefficient on unemployment is relatively small and positive. In showing this, our strategy is to first express the above (B.19) into the following form:⁷

$$\check{\Pi}_{t}^{w} = \phi_{V}\check{V}_{t} + \phi_{U}\check{U}_{t-1} + \frac{1}{1+\rho}\check{\Pi}_{t+1}^{w}$$
(B.20)

⁶As $\chi = 1$ and S = 3.6%, $\chi > S$ at our steady state.

⁷(B.19) becomes equivalent to equation (20), as θ_t is a function of V_t and U_t .

for some ϕ_V and ϕ_U , which are complex collections of model parameters and steady-state values. And then we use the fact that quits, which can also be decomposed into deviations of vacancy and unemployment, can be viewed as an imperfect proxy for "true" tightness since higher tightness leads to a higher rate of quits. A higher unemployment (vacancy) rate lowers (raises) tightness, and thus reduces (raises) quits. If $\check{Q}_t \equiv g_{Q,V}\check{V}_t + g_{Q,U}\check{U}_{t-1}$ where $g_{Q,U} < 0$ is of magnitude large enough, then equation (B.20) becomes

$$\Pi_t^w = \underbrace{\frac{\phi_V}{g_{Q,V}}}_{\equiv \beta_Q > 0} \check{Q}_t + \underbrace{\left(\phi_U \quad \phi_V \frac{g_{Q,U}}{g_{Q,V}}\right)}_{\equiv \beta_U} \check{U}_{t-1} + \frac{1}{1+\rho} \check{\Pi}_{t+1}^w, \tag{B.21}$$

possibly yielding a positive β_U .

Derivation To begin to simplify the wage Phillips curve (B.19), we decompose all of the following right-hand-side variables, \check{Q}_t , \check{T}_t , \check{R}_t , $\check{\varepsilon}_{R,t}$, and $\check{\varepsilon}_{S,t}$ into vacancy and unemployment deviations. The tightness term, $T_t = \frac{V_t}{N_{t-1}}$, is simple: in log deviations from steady state, it becomes

$$\check{T}_t = \check{V}_t + \frac{U}{1 - U} \check{U}_{t-1}.$$

As for the rest, we will show that we can write the decompositions as follows:

1. $\check{R}_t \equiv g_{R,V}\check{V}_t + g_{R,U}\check{U}_{t-1}$

Derivation: Recall that the recruiting function is

$$R_t = g(\theta_t) \left(\phi_{E,t} \frac{1}{2} + \phi_{U,t} \left(\frac{\xi^{\gamma}}{1 + \xi^{\gamma}} \right) \right).$$

For practical purposes we define $C \equiv \frac{\xi^{\gamma}}{1+\xi^{\gamma}}$, which is increasing in the ratio of consumption for employed to unemployed workers. Then we obtain:

$$g_{R,V} = -\frac{\theta^2}{1+\theta^2},$$

and

$$g_{R,U} = \frac{\theta^2}{1+\theta^2} \cdot \frac{U(1-\lambda_{EE})}{\lambda_{EE}(1-U)+U} + \frac{0.5\phi_E}{0.5\phi_E + \mathcal{C}\phi_U} \cdot \frac{U}{1-U} \cdot \frac{\lambda_{EE}\phi_E-\lambda_{EE}-\phi_E}{\lambda_{EE}} + \frac{\mathcal{C}\phi_U}{0.5\phi_E + \mathcal{C}\phi_U} \cdot (1-\phi_U(1-\lambda_{EE})).$$

2. $\check{S}_t \equiv g_{S,V}\check{V}_t + g_{S,U}\check{U}_{t-1}$ and $\check{Q}_t \equiv g_{Q,V}\check{V}_t + g_{Q,U}\check{U}_{t-1}$

Derivation: Recall that the quit function $Q_t = S_t$ s is given by

$$Q_t = \begin{pmatrix} 1 & s \end{pmatrix} \left(\lambda_{EE} f(\theta_t) \frac{1}{2} + \lambda_{EU} \left(\frac{1}{1 + \xi^{\gamma}} \right) \right)$$
(B.22)

Then, we obtain:

$$g_{Q,V} = \frac{0.5\lambda_{EE}f}{0.5\lambda_{EE}f + \lambda_{EU}(1 - \mathcal{C})} \cdot \frac{1}{1 + \theta^2},$$

and

$$g_{Q,U} = \frac{0.5\lambda_{EE}f}{0.5\lambda_{EE}f + \lambda_{EU}(1-\mathcal{C})} \cdot \frac{1}{1+\theta^2} \cdot \frac{U(1-\lambda_{EE})}{\lambda_{EE}(1-U) + U}.$$

3. $\check{\varepsilon}_{R,t} = g_{\varepsilon_{R,U}}\check{U}_{t-1}$

Derivation: Note that in equilibrium, $\varepsilon_{R,t}$ is given by

$$\varepsilon_{R,t} = \frac{g(\theta_t)\gamma\left(\frac{\phi_{E,t}}{4} + \phi_{U,t}\frac{\xi^{\gamma}}{(1+\xi^{\gamma})^2}\right)}{g(\theta_t)\left(0.5\phi_{E,t} + \left(\frac{\xi^{\gamma}}{1+\xi^{\gamma}}\right)\phi_{U,t}\right)} = \frac{\gamma\left(\frac{\phi_{E,t}}{4} + \phi_{U,t}\mathcal{C}(1-\mathcal{C})\right)}{0.5\phi_{E,t} + \phi_{U,t}\mathcal{C}},$$

from which we obtain

$$g_{\varepsilon_{R,U}} = \left(\frac{0.25\phi_E}{0.25\phi_E + \mathcal{C}(1-\mathcal{C})\phi_U} - \frac{0.5\phi_E}{0.5\phi_E + \mathcal{C}\phi_U}\right) \frac{U}{1-U} \frac{\lambda_{EE}\phi_E - \lambda_{EE} - \phi_E}{\lambda_{EE}} + \left(\frac{\mathcal{C}(1-\mathcal{C})\phi_U}{0.25\phi_E + \mathcal{C}(1-\mathcal{C})\phi_U} - \frac{\mathcal{C}\phi_U}{0.5\phi_E + \mathcal{C}\phi_U}\right) (1-\phi_U(1-\lambda_{EE})).$$

4. $\check{\varepsilon}_{S,t} = g_{\varepsilon_{S,V}}\check{V}_t + g_{\varepsilon_{S,U}}\check{U}_{t-1}$

Derivation: Note that in equilibrium $\varepsilon_{S,t}$ is given by

$$\varepsilon_{S,t} = \frac{(1 \quad s)\gamma \left(f(\theta_t)\lambda_{EE}\frac{1}{4} + \mathcal{C}(1 \quad \mathcal{C})\lambda_{EU}\right)}{s + (1 \quad s) \left(0.5 \cdot \lambda_{EE}f(\theta_t) + (1 \quad \mathcal{C})\lambda_{EU}\right)},$$

from which we obtain

$$g_{\varepsilon_{S,V}} = \left(\frac{0.25\lambda_{EE}f}{0.25\lambda_{EE}f + \mathcal{C}(1-\mathcal{C})\lambda_{EU}} - \frac{0.5(1-s)\lambda_{EE}f}{s + (1-s)(0.5\lambda_{EE}f + (1-\mathcal{C})\lambda_{EU})}\right)\frac{1}{1+\theta^2},$$

$$g_{\varepsilon_{S,U}} = g_{\varepsilon_{S,V}} \cdot \frac{U(1 \quad \lambda_{EE})}{\lambda_{EE}(1 \quad U) + U}.$$

Decomposing Wage Growth into Vacancies and Unemployment Combining these results, we can plug in and rewrite the wage Phillips curve just in terms of vacancies and unemployment. Let $\Delta_1 \equiv \varepsilon_S + S(\varepsilon_R \quad \varepsilon_S)$ and let $\Lambda_1 \equiv \varepsilon_R + (\chi \quad S)(\varepsilon_R \quad \varepsilon_S)$. Then the wage Phillips curve (B.19) can be written as:

$$\begin{split} \check{\Pi}_{t}^{w} &= \underbrace{\frac{\kappa}{\psi^{w}} \left[\Lambda_{1} + \Delta_{1} \left(g_{S,V} \quad g_{R,V} \right) \quad \varepsilon_{S} g_{\varepsilon_{S},V} \right] \check{V}_{t}}_{\equiv \phi_{V} > 0} \\ &+ \underbrace{\frac{\kappa}{\psi^{w}} \left[\frac{U}{1 \quad U} \Lambda_{1} + \Delta_{1} \left(g_{S,U} \quad g_{R,U} \right) + \varepsilon_{R} g_{\varepsilon_{R},U} \quad \varepsilon_{S} g_{\varepsilon_{S},U} \right]}_{\equiv \phi_{U} < 0} \check{U}_{t-1} + \frac{1}{1 + \rho} \check{\Pi}_{t+1}^{w}. \end{split}$$

$$\end{split}$$

$$(B.23)$$

Under our calibration in Table 2, quantitatively (B.23) becomes

$$\check{\Pi}_{t}^{w} = 10^{-2} \times \left(1.83\check{V}_{t} \quad 0.3\check{U}_{t-1}\right) + \frac{1}{1+\rho}\check{\Pi}_{t+1}^{w}$$
(B.24)

Decomposing Wage Growth into Quits and Unemployment: First, note $\check{Q}_t \equiv g_{Q,V}\check{V}_t + g_{Q,U}\check{U}_{t-1}$ yields

$$\check{V}_t = \frac{1}{g_{Q,V}}\check{Q}_t \quad \frac{g_{Q,U}}{g_{Q,V}}\check{U}_{t-1},$$

which with (B.24) yields:

$$\Pi_{t}^{w} = \underbrace{\frac{\phi_{V}}{g_{Q,V}}}_{\equiv \beta_{Q} > 0} \check{Q}_{t} + \underbrace{\left(\phi_{U} \quad \phi_{V} \frac{g_{Q,U}}{g_{Q,V}}\right)}_{\equiv \beta_{U}} \check{U}_{t-1} + \frac{1}{1+\rho} \check{\Pi}_{t+1}^{w}$$

$$= 10^{-2} \times \left(2.46\check{Q}_{t} + 0.0916\check{U}_{t-1}\right) + \frac{1}{1+\rho} \check{\Pi}_{t+1}^{w}$$
(B.25)

at a monthly frequency, where β_Q dominates β_U in magnitude under our calibration, and β_U becomes positive. Thus, we prove equation (E.2). Equation (20) follows from log-linearizing equation (B.22), which describes a one-to-one relationship between \check{Q}_t and $\check{\theta}_t$.

A Simpler Wage Phillips Curve With No On-the-job Search Here, we argue that convex vacancy costs and on-the-job search combine to make vacancies more important in the wage

and

Phillips curve, i.e., $|\phi_V|$ is significantly bigger than $|\phi_U|$. When $\chi \approx 0$ and $\lambda_{EE} = 0$, we can show that $\check{V}_t \quad \check{U}_{t-1}$, i.e., $\left(\frac{\check{V}_t}{U_{t-1}}\right)$, becomes a sufficient statistic that explains wage growth.⁸ In doing this, we will assume that $s \simeq 0$ and $\mathcal{C} \equiv \frac{\xi^{\gamma}}{1+\xi^{\gamma}} \simeq 1$, both of which hold approximately under our calibration.

First, we demonstrate that as we eliminate on-the-job search and let $\lambda_{EE} \rightarrow 0$, the decomposition of the wage Phillips curve into \check{V}_t and \check{U}_{t-1} in (B.23) simplifies considerably. The first term \check{T}_t remains:

$$\lim_{\lambda_{EE}\to 0}\check{T}_t = \check{V}_t + \frac{U}{1-U}\check{U}_{t-1}.$$

As for the rest, we will show that we can write the decompositions as follows:

- 1. $\lim_{\lambda_{EE}\to 0} \check{R}_t = \frac{\theta^2}{1+\theta^2} \left(\check{V}_t \quad \check{U}_{t-1}\right)$ since $\phi_U \to 1$ (and $\phi_E \to 0$) as we shut down on-the-job search, i.e., $\lambda_{EE} \to 0$.
- 2. $\lim_{\lambda_{EE}\to 0} \check{S}_t = 0.$
- 3. $\lim_{\lambda_{EE}\to 0} \check{\varepsilon}_{R,t} = 0.$
- 4. $\lim_{\lambda_{EE}\to 0} \check{\varepsilon}_{S,t} = 0.$

leading to the simplified wage Phillips curve given by:

$$\lim_{\lambda_{EE}\to 0} \check{\Pi}_t^w = \frac{\kappa}{\psi^w} \left[\underbrace{(\varepsilon_S + S(\varepsilon_R - \varepsilon_S))}_{\equiv \Delta_1} \underbrace{\frac{\theta^2}{1 + \theta^2} \left(\underbrace{V_t}_{U_{t-1}}\right) + \underbrace{(\varepsilon_R + (\chi - S)(\varepsilon_R - \varepsilon_S))}_{\equiv \Lambda_1} \check{T}_t \right] + \frac{1}{1 + \rho} \check{\Pi}_{t+1}^w$$

Noting that:

$$\lim_{\lambda_{EE} \to 0} \theta_t = \frac{V_t}{U_{t-1}}$$

So we obtain:

$$\lim_{\lambda_{EE}\to 0} \check{\Pi}_t^w = \frac{\kappa}{\psi^w} \left[\underbrace{(\varepsilon_S + S(\varepsilon_R - \varepsilon_S))}_{\equiv \Delta_1} \frac{\theta^2}{1 + \theta^2} \check{\theta}_t + \underbrace{(\varepsilon_R + (\chi - S)(\varepsilon_R - \varepsilon_S))}_{\equiv \Lambda_1} \check{T}_t \right] + \frac{1}{1 + \rho} \check{\Pi}_{t+1}^w$$

As we further assume that $s \simeq 0$ and $C \simeq 1$, we can find the following results for the

⁸In other words, when $\chi \approx 0$ and $\lambda_{EE} = 0$ with very small s, $|\phi_V| \simeq |\phi_U|$.

steady-state values underlying Λ_1 and Δ_1 :

$$\lim_{\lambda_{EE} \to 0} S = s + (1 \quad s)\lambda_{EU}(1 \quad \mathcal{C}) = 0$$
$$\lim_{\lambda_{EE} \to 0} \varepsilon_R = \gamma(1 \quad \mathcal{C}) = 0$$
$$\lim_{\lambda_{EE} \to 0} \varepsilon_S = \frac{(1 \quad s)\gamma\left(\mathcal{C}(1 \quad \mathcal{C})\lambda_{EU}\right)}{s + (1 \quad s)\left(1 \quad \mathcal{C}\right)\lambda_{EU}} = -\gamma$$

which implies that $\Delta_1 = \gamma$ and $\Lambda_1 = \gamma \chi$. So with $\chi = 0$, our wage Phillips curve in terms of \check{V}_t and \check{U}_{t-1} simplifies to:

$$\lim_{\lambda_{EE}\to 0}\check{\Pi}_t^w = \frac{\kappa}{\psi^w} \gamma\left(\frac{\theta^2}{1+\theta^2}\right)\check{\theta}_t + \frac{1}{1+\rho}\check{\Pi}_{t+1}^w.$$

Therefore, the wage Phillips curve can be written entirely in terms of market tightness θ_t when there is no on-the-job search, as in Gagliardone and Gertler (2023).

In sum, as the exogenous separation rate $s \to 0$, the consumption ratio $\xi \to \infty$ or $\mathcal{C} \to 1$ (so unemployed workers always take jobs) and $\lambda_{EE} \to 0$, we have that $S \to 0$, $\varepsilon_R \to 0$, and $\varepsilon_S \to -\gamma$. Then there is complete weight on θ in the wage Phillips curve, and if $\lambda_{EE} = 0$, $\theta_t = \frac{V_t}{U_{t-1}}$. In contrast, in our setting where $\chi = 1$, which implies a convex vacancy cost, and $\lambda_{EE} > 0$, we see that $|\phi_V|$ is much higher than $|\phi_U|$ as seen in (B.24), and β_Q is much higher than β_U as seen in (B.25).

B.4 Euler Equation With Fixed Real Unemployment Benefits

This section shows how the assumptions in Section 4.2.1 can be made consistent with the standard Euler equation of the household, given appropriate assumptions on how the household reallocates consumption. Recall the goal in Section 4.2.1 was to modify the model so that the desirability of unemployment varied with the price level; here, we show one way to make that model consistent with the standard Euler equation (10) used throughout the main text.

Suppose that when unemployed, household members are guaranteed some quantity b of real consumption goods and receive no other income (e.g., some nominal unemployment benefit perfectly indexed to inflation). When employed, they receive a nominal wage W_t . The household takes b, market wages W_t , and the price level P_t as given, but can smooth all members consumption by choosing a proportional "top-up" each period, multiplying each

type of worker's income by $1 + \nu_t$. This yields consumption levels

$$C_t^u = b(1 + \nu_t)$$
$$C_t^e = \frac{W_t}{P_t}(1 + \nu_t),$$

and total consumption

$$C_t = U_t b(1 + \nu_t) + (1 \quad U_t) \frac{W_t}{P_t} (1 + \nu_t).$$
(B.26)

Making the top-up proportional and identical in both states u and e implies that as the household smooths consumption, it does not affect the relative attractiveness of unemployment and employment: the $1 + \nu_t$ terms cancel out separation and recruiting probabilities from unemployment $s_{ju}(W_{jt}|P_t)$ and $r_{uj}(W_{jt}|P_t)$ presented in Section 4.2.1.

Continuing on to derive the Euler equation, the household maximizes

$$\sum_{t=0}^{\infty} \left(\frac{1}{1+\rho}\right)^{t} \left(U_{t}\ln(C_{t}^{u}) + (1 \quad U_{t})\ln(C_{t}^{e})\right)$$
$$= \sum_{t=0}^{\infty} \left(\frac{1}{1+\rho}\right)^{t} \left(U_{t}\ln(b(1+\nu_{t})) + (1 \quad U_{t})\ln\left(\frac{W_{t}}{P_{t}}(1+\nu_{t})\right)\right)$$
$$= \sum_{t=0}^{\infty} \left(\frac{1}{1+\rho}\right)^{t} \left(U_{t}\ln(b) + (1 \quad U_{t})\ln\left(\frac{W_{t}}{P_{t}}\right) + \ln(1+\nu_{t})\right).$$

The household's budget constraint can be written as:

$$(1+\nu_t)\left(U_tb+(1 \quad U_t)\frac{W_t}{P_t}\right)+\frac{B_t}{P_t}=\frac{D_t}{P_t}+(1 \quad U_t)\frac{W_t}{P_t}+\frac{(1+i_{t-1,t})B_{t-1}}{P_t}.$$

The household's Lagrangian function is given by:

$$\mathcal{L} = \sum_{t=0}^{\infty} \left(\frac{1}{1+\rho} \right)^t \left(U_t \ln(b) + (1 \quad U_t) \ln\left(\frac{W_t}{P_t}\right) + \ln(1+\nu_t) + \lambda_t \left[(1+\nu_t) \left(U_t b + (1 \quad U_t) \frac{W_t}{P_t} \right) \quad \frac{B_t}{P_t} + \frac{D_t}{P_t} + \frac{(1+i_{t-1,t})B_{t-1}}{P_t} \right] \right).$$

The household's only choice variables are τ_t and B_t . The first order conditions are

$$\mathcal{L}_{\nu_t} = 0 : \frac{1}{1+\nu_t} = \lambda_t \left(U_t b + (1 \quad U_t) \frac{W_t}{P_t} \right),$$
$$\mathcal{L}_{B_t} = 0 : \frac{\lambda_t}{P_t} = \left(\frac{1}{1+\rho} \right) \lambda_{t+1} \frac{(1+i_{t,t+1})B_t}{P_{t+1}}.$$

Plugging in the expression for aggregate consumption in equation (B.26) into the first order condition on τ_t yields the standard Euler equation used in the main text:

$$C_t^{-1} = \frac{1}{1+\rho} \frac{1+i_{t,t+1}}{\pi_{t,t+1}} C_{t+1}^{-1}.$$

Appendix C Analytical Results for Pass-Through Across Different Classes of Models

This section analyzes the pass-through of prices to wages in response to a temporary decline in the endowment good X_t , assuming that monetary policy stabilizes the business cycle holding N_t fixed. We consider the following variations of the model which alter the labor block in Section 3. We work through the case of (i) a sticky-price, flexible-wage New Keynesian model where workers supply labor in a frictionless market, (ii) a flexible price, sticky-wage New Keynesian model where wages are set by unions as in Erceg et al. (2000); Galí et al. (2012); and (iii) our model in Section 3.

As in the paper, we assume throughout that consumption is a CES bundle of services Y_t , produced with labor, and goods X_t which households receive as an endowment (equivalently, perfectly competitive firms receive X_t and sell it for pure profit, rebating the proceeds to households as dividends). We have

$$C_{t} = \left(\alpha_{Y}^{\frac{1}{\eta}}Y_{t}^{\frac{\eta-1}{\eta}} + \alpha_{X}^{\frac{1}{\eta}}X_{t}^{\frac{\eta-1}{\eta}}\right)^{\frac{\eta}{\eta-1}}$$
(C.1)

and

$$P_{t} = \left(\alpha_{Y} P_{y,t}^{1 \eta} + \alpha_{X} P_{x,t}^{1 \eta}\right)^{\frac{1}{1 \eta}}$$

C.1 Sticky-Price, Flexible Wage New Keynesian Model

We assume here that $P_{y,t}$ is set subject to some nominal rigidities as in the baseline model in Section 3 (i.e. Rotemberg adjustment costs), but where firms hire labor in a standard spot market with flexible nominal wage W_t , so there is no unemployment. The household chooses paths for consumption and labor (and zero net supply nominal bonds) to maximize:

$$\sum_{t=0}^{\infty} \beta^t \left(\frac{\sigma}{\sigma - 1} C_t^{\frac{\sigma - 1}{\sigma}} - \frac{1}{1 + \frac{1}{\nu}} N_t^{1 + \frac{1}{\nu}} \right)$$

subject to the budget constraint,

$$C_t = \frac{D_t}{P_t} - \frac{B_t}{P_t} + \frac{(1 + i_{t-1,t})B_{t-1}}{P_t} + \frac{W_t}{P_t}N_t.$$

This yields the following intratemporal optimality condition:

$$N_t^{\frac{1}{\nu}} = W_t \frac{C_t^{\frac{1}{\sigma}}}{P_t}$$

So in our model with Neoclassical labor supply, the following decomposition must hold to first order:

$$\frac{1}{\nu}\check{N}_t = \check{W}_t \quad \frac{1}{\sigma}\underbrace{\left(\check{P}_t + \check{C}_t\right)}_{=\widecheck{P_tC_t}} + \frac{1}{\sigma}\underbrace{\sigma}\check{P}_t$$

So that when monetary policy fixes $\check{N}_t = 0$, we have

$$\frac{\widetilde{W_t}}{P_{y,t}} = \frac{1}{\sigma} \frac{\widetilde{P_tC_t}}{P_{y,t}} + \frac{\sigma}{\sigma} \frac{1}{\sigma} \frac{\widetilde{P_t}}{P_{y,t}}$$
(C.2)

Now we can write the two right hand side terms as functions of the shock X_t : first note that CES demand implies

$$\frac{P_t}{P_{y,t}} = \left(\frac{Y_t}{C_t}\right)^{\frac{1}{\eta}} \alpha_y^{\frac{1}{\eta}}.$$

Under our experiment where monetary policy stabilizes N_t , and hence Y_t , from (C.1) we have to first order that $\check{C}_t = \alpha_x^{\frac{1}{\eta}} \left(\frac{X}{C}\right)^{\frac{\eta-1}{\eta}} \check{X}_t$ and

$$\frac{\widetilde{P_t}}{P_{y,t}} = \frac{1}{\eta} \left(\alpha_x^{\frac{1}{\eta}} \left(\frac{X}{C} \right)^{\frac{\eta-1}{\eta}} \check{X}_t \right)$$
(C.3)

so that when X_t falls, the price of the aggregate consumption bundle in terms of the laborintensive good, $\frac{P_t}{P_{y,t}}$ goes up (i.e. we need more units of Y-good to buy one unit of C-good). We also have for aggregate spending,

$$\frac{\widecheck{P_tC_t}}{P_{y,t}} = \frac{\eta - 1}{\eta} \left(\alpha_x^{\frac{1}{\eta}} \left(\frac{X}{C} \right)^{\frac{\eta - 1}{\eta}} \check{X}_t \right)$$

So that aggregate nominal spending may either rise, or fall, depending on η . With Cobb-Douglas utility with $\eta = 1$, nominal spending is unchanged. Consider the effects of negative shock to X_t on the wage when monetary policy holds N_t fixed, and examine equation (C.2):

- We can see when $\eta = \sigma = 1$, the wage denoted in units of the service good or numeraire, i.e., $\frac{W_t}{P_{y,t}}$ remains unchanged.
- With Cobb-Douglas preferences, $\eta = 1$, we see from (C.3) that the relative price still rises, so everything depends on σ : if $\sigma > 1$, as is commonly assumed in macro applications, then there is positive pass-through from prices to wages.
- If σ = 1, η < 1 then there is positive pass-through from prices to wages. When it is hard to substitute away from X_t, and total expenditure rises.

Discussion: Even in a perfectly competitive labor market, workers' wages can respond to an increased cost of living even when their productivity is unaffected by the shock. The sign and magnitude of the response depends on the strength of income and substitution effects (governed by σ) and wealth effects (governed by η) stemming from a change in $P_{x,t}X_t$, where we obtain

$$\frac{P_{x,t}X_t}{P_{y,t}} = \underbrace{\frac{\eta}{\eta}}_{<0} \underbrace{\check{X}_t}_{<0} > 0 \tag{C.4}$$

when $\eta < 1$, so that households' non-labor income from endowment good X_t increases and so does their wealth, possibly lowering labor supply due to the wealth effect.

In specifications where $\sigma \ge 1$ and $\eta < 1$, the decline in X_t makes workers prefer leisure; thus if monetary policy is holding leisure (and labor) fixed, the wage must rise in equilibrium.

C.2 Flexible Price, Sticky Wage New Keynesian Model

We now consider the effect of a temporary fall in X_t when wages are sticky as in Erceg et al. (2000), again analyzing the shock under the assumption that monetary policy stabilizes aggregate labor output N_t . Specifically, we assume that households now supply multiple types of labor; unions set wages for each type to maximize household utility subject to facing CES demand for each type from a "labor packer" which packages each labor type $N_t(i)$ into aggregate labor $N_t = \left(\int_0^1 N_t(i)^{\frac{1+\nu}{\nu}} di\right)^{\frac{\nu}{1+\nu}}$ which is purchased at wage W_t by services firms—and in our setting, combined with X_t to form consumption C_t . Wages are sticky because unions only occasionally receive the chance to reset their wage.

Households now maximize the following: specializing to log utility with $\sigma = 1$,

$$\sum_{t=0}^{\infty} \beta^t \left(\ln C_t - \int_0^1 \frac{1}{1+\frac{1}{\nu}} N_t(i)^{1+\frac{1}{\nu}} di \right),$$

subject to the budget constraint, $C_t = \frac{D_t}{P_t} - \frac{B_t}{P_t} + \frac{(1+i_{t-1,t})B_{t-1}}{P_t} + \frac{W_t}{P_t}N_t$. Under these assumptions, we can derive the following standard wage Phillips curve (see e.g., Galí, 2011; Galí et al., 2012):

$$\check{\Pi}_t^{\check{w}} = \beta \mathbb{E}_t \{ \Pi_{t+1}^{\check{w}} \} + \lambda \left(\quad \widecheck{W}_t + \widecheck{P_t C_t} + \frac{1}{\nu} \check{N}_t \right).$$

for some constant $\lambda > 0$. Analyzing this case is only harder than the flexible wage case of Section C.1 because of the presence of the forward-looking term Π_{t+1}^w . To make progress, rewrite this in relative price terms:

$$\check{\Pi}_{t}^{\check{w}} = \beta \mathbb{E}_{t} \{ \Pi_{t+1}^{\check{w}} \} + \lambda \left(\frac{\widecheck{W_{t}}}{P_{y,t}} + \frac{\widecheck{P_{t}C_{t}}}{P_{y,t}} + \frac{1}{\nu} \check{N}_{t} \right).$$
(C.5)

Consider the household's budget constraint in equilibrium: using the fact that bonds are in zero net supply, and expanding the D_t term by denoting d_t the dividends potentially paid by services firms (zero, if prices are flexible and they are perfectly competitive), this is

$$P_t C_t = W_t N_t + P_{x,t} X_t + d_t$$

How will each term in this equation respond to an X_t shock? Rewriting, we have:

$$\frac{P_t C_t}{P_{y,t}} = \frac{W_t}{P_{y,t}} N_t + \frac{P_{x,t}}{P_{y,t}} X_t + \frac{d_t}{P_{y,t}}$$

Now recall that given fixed N_t CES demand yields:

$$\frac{\widecheck{P_tC_t}}{P_{y,t}} = \frac{\eta - 1}{\eta} \left(\alpha_x^{\frac{1}{\eta}} \left(\frac{X}{C} \right)^{\frac{\eta - 1}{\eta}} \check{X}_t \right)$$

If $\eta < 1$, then $\frac{P_0C_0}{P_{y,0}}$ rises in response to a negative X_0 shock and $\frac{P_tC_t}{P_{y,t}}$ is zero in other periods (t > 0) when there is no shock. From (C.4), we see the middle term, in its deviation from steady state, is zero when there is no shock. Thus, we obtain for all t > 0:

$$0 = \frac{WN}{PC} \frac{\widetilde{W_t}}{P_{y,t}} + \frac{d}{PC} \check{d}_t.$$

If there are no time-varying profits, e.g., if prices are flexible, then we have that $\check{d}_t = 0$ and thus $\underbrace{W_t}_{P_{y,t}} = 0$. As a result, the forward looking wage Phillips curve (C.5) implies $\Pi_t^{\check{w}} = 0$ for all t > 0, and the wage Phillips curve for the initial period greatly simplifies to

$$\pi_0^w = \lambda \left(\quad \widecheck{W}_0 + \widecheck{P_0 C_0} \right)$$

Given that wage inflation is defined as $\check{\Pi}_t^w = \check{W}_t \quad \check{W}_{t-1}$ with $\check{W}_{-1} = 0$, we can write

$$\check{W}_0 = \lambda (\quad \widecheck{W}_0 + \widecheck{P_0 C_0})$$

Divide by $P_{y,0}$ to apply our above results for $\frac{P_0C_0}{P_{y,0}}$ and find that when $\eta < 1$, the right hand side is positive for a negative X shock, and we thus have positive pass-through to wages.

Discussion: As discussed in above Section C.1, depending on the strength of income, substitution, and wealth effects (governed by η), wages can either rise or fall in response to the shock. Here for $\sigma = 1$, we again find that $\eta < 1$ implies pass-through from prices to wages in response to the X_t shock. The analysis with sticky wages is not that different from the flexible wage case.

C.3 Wage Posting Model with On-the-job Search and Nominal Rigidities

This section analyzes the baseline model in Section 3 to elaborate the conditions under which there is no pass-through from prices to wages. In that model, there is no pass-through from prices to wages: in response to an X_t shock, when monetary policy perfectly stabilizes N_t , it also perfectly stabilizes wage inflation. We demonstrate that this relies in part on the assumption that vacancy costs are denominated in labor: if vacancy costs are denominated in final goods, then headline inflation passes through into wages, even when monetary policy stabilizes the labor market. The title of this section reflects the fact that it does not matter for the analysis here whether prices or wages are sticky, so long as the presence of nominal rigidities allows monetary authorities to stabilize N_t .

To demonstrate the role of how adjustment costs are denominated, we generalize the firms problem slightly: let P_t^V denote the nominal price in which vacancy costs are denominated, and let P_t^{ψ} be the nominal price in which wage adjustment costs are denominated (which we will show will not matter). Then firm *j* maximizes present-discounted revenues, less costs (abstracting from price adjustment costs, which do not affect the wage Phillips curve), given by

$$\sum_{t=0}^{\infty} \left(\frac{1}{1+\rho}\right)^t \left(P_{y,t}^j Y_t^j - W_t^j N_t^j - c(V_t^j)^{1+\chi} (N_{t-1}^j)^{-\chi} P_t^V - \frac{\psi^w}{2} \left(\frac{W_t^j}{W_{t-1}^j} - 1\right)^2 N_t P_t^{\psi}\right),$$

subject to the law of motion for employment,

$$N_t^j = (1 \quad S_t(W_t^j))N_{t-1}^j + V_t^j R_t(W_t^j)$$

and some production and demand functions for Y_t^j . Combining the firm's first order conditions for V_t^j and W_t^j , and assuming a symmetric equilibrium, yields a nonlinear wage Phillips curve given by

$$\begin{split} \psi^{w} \left(\Pi_{t}^{w} - 1\right) \Pi_{t}^{w} P_{t}^{\psi} + W_{t} = & P_{t}^{V} c(1+\chi) \left(\frac{V_{t}}{N_{t-1}}\right)^{\chi} \left(\frac{V_{t}}{N_{t}} \varepsilon_{R,W_{t}} - \frac{N_{t-1}}{N_{t}} \frac{S_{t}(W_{t})}{R_{t}(W_{t})} \varepsilon_{S,W_{t}}\right) \\ &+ \frac{\psi^{w}}{1+\rho} \left(\Pi_{t+1}^{w} - 1\right) \Pi_{t+1}^{w} \frac{N_{t+1}}{N_{t}} P_{t+1}^{\psi}. \end{split}$$

Gather the labor market tightness terms in $Z_t \equiv c(1+\chi) \left(\frac{V_t}{N_t-1}\right)^{\chi} \left(\frac{V_t}{N_t}\varepsilon_{R,W_t} - \frac{N_{t-1}}{N_t}\frac{S_t(W_t)}{R_t(W_t)}\varepsilon_{S,W_t}\right)$

and log-linearize, defining $\pi_t^V \equiv \frac{P_t^V}{P_{t-1}^V}$, let $\check{\omega}_t \equiv \sum_{s=0}^t (\pi_t^w - \pi_t^V)$, we can obtain

$$\check{\pi}_t^w = \frac{ZP^V}{\psi^w P^\psi} \sum_{s=0}^\infty \left(\frac{1}{1+\rho}\right)^s \left(\check{Z}_{t+s} \quad \check{\omega}_{t+s}\right).$$

When monetary policy stabilizes employment, and $N_t = 0$, it does follow that $Z_t = 0$, as proved in Section C.3.1 so the wage Phillips curve reduces to:

$$\check{\pi}_t^w = \frac{ZP^V}{\psi^w P^\psi} \sum_{s=0}^\infty \left(\frac{1}{1+\rho}\right)^s (\check{\omega}_{t+s})$$

If the cost of posting vacancies is denominated in labor, so $P_t^V = W_t$, then $\check{\omega}_t = 0$ and monetary policy stabilizes wage growth as well as employment.

C.3.1 Showing That $\check{N}_t = 0$ implies $\check{Z}_t = 0$

To see this result, first note that holding $N_t = N$ constant implies that the number of searchers $S = \lambda_{EE}N + \lambda_{EU}(1 N)$ is constant; the shares appearing in the definitions of the separation and recruiting rates, $\phi_{E,t}$ and $\phi_{U,t}$ in equations (17) and (18), are thus also constant. This means the tightness term θ_t is constant so long as V_t is constant. If V_t and therefore θ_t are constant, then the separation rates and elasticities in Z_t are also held constant. Ergo, all we must do is show that V_t is constant.

To do so, write the law of motion for employment when $N_t = N_{t-1} = N$, plugging in for the separation and recruiting rates, to yield $V_t \cdot R_t = N \cdot S_t$ that can be written as:

$$V_t g\left(\frac{V_t}{\mathcal{S}}\right) \left(\phi_E \frac{1}{2} + \phi_U \left(\frac{\xi^{\gamma}}{1+\xi^{\gamma}}\right)\right) \qquad N f\left(\frac{V_t}{\mathcal{S}}\right) (1-s) \frac{\lambda_{EE}}{2} = N \left[s + (1-s) \left(\lambda_{EU} \left(\frac{1}{1+\xi^{\gamma}}\right)\right)\right]$$

Now using our definition for g, rewrite the left hand side in terms of f:

$$\mathcal{S}f\left(\frac{V_t}{\mathcal{S}}\right)\left(\phi_E\frac{1}{2} + \phi_U\left(\frac{\xi^{\gamma}}{1+\xi^{\gamma}}\right)\right) \quad Nf\left(\frac{V_t}{\mathcal{S}}\right)(1-s)\frac{\lambda_{EE}}{2} = N\left[s + (1-s)\left(\lambda_{EU}\left(\frac{1}{1+\xi^{\gamma}}\right)\right)\right]$$

leading to

$$f\left(\frac{V_t}{\mathcal{S}}\right) = \frac{N\left[s + (1 \quad s)\left(\lambda_{EU}\left(\frac{1}{1+\xi\gamma}\right)\right)\right]}{\mathcal{S}\left(\phi_E \frac{1}{2} + \phi_U\left(\frac{\xi\gamma}{1+\xi\gamma}\right)\right) \quad N(1 \quad s)\frac{\lambda_{EE}}{2}}$$

Thus, there is a unique solution for $V_t = V$ for a given N (the steady state value). So

we conclude that when monetary policy stabilizes N_t , V_t and θ_t are also stabilized, and Z_t is stabilized.

Appendix D Weight on Vacancies and Unemployment in the Baseline Models' Wage Phillips Curve

We start from the log-linearized wage Phillips curve (B.20), written in vacancy and unemployment rates:

$$\check{\Pi}_{t}^{w} = \phi_{V}\check{V}_{t} + \phi_{U}\check{U}_{t-1} + \frac{1}{1+\rho}\check{\Pi}_{t+1}^{w}$$
(D.1)

for some coefficients $\phi_V > 0$ and $\phi_U < 0$. In Appendix C.3, we prove that the absence of aggregate price inflation in the right hand side of the linearized wage Phillips curve (D.1) is stemming from the fact that the vacancy-creating cost is denominated in labor, not the final good. Based on our calibration in Table 2, we now illustrate how varying the probability of being allowed to search on the job, λ_{EE} , affects the predictions of the model about the relative importance of vacancies, as opposed to unemployment, in the wage Phillips curve.

Since both $\phi_V > 0$ and $\phi_U < 0$ are complex collections of model parameters and steadystate values, to consider their relative magnitudes, we proceed numerically, and specialize to particular parameter choices. As we see in (B.24), ϕ_V is much larger in magnitude than ϕ_U . This result turns out to stem both from the presence of on-the-job search ($\lambda_{EE} > 0$) and also from the convexity of vacancy costs ($\chi > 0$), as proven in Appendix B.3.1. Figure D.2 shows how the relative importance of vacancies in explaining wage growth, represented by the ratio of coefficients in (D.1), $\left|\frac{\phi_V}{\phi_U}\right|$, increases monotonically in on-the-job search intensity λ_{EE} under the benchmark calibration $\chi = 1$ and also when $\chi = 0$, or a linear cost of posting a vacancy which is commonly assumed in the search literature. The limit case where $\chi = 0$ and $\lambda_{EE} \rightarrow 0$ is of particular interest as a benchmark: as Appendix B.3.1 shows, at the limit where $\lambda_{EE} \rightarrow 0$ and $\chi \rightarrow 0$, $\left|\frac{\phi_V}{\phi_U}\right|$ converges to one and wage growth becomes solely a function of market tightness $\theta_t = \frac{V_t}{U_{t-1}}^9$ following the literature: see e.g., Gagliardone and Gertler (2023).

We acknowledge that simply pointing out the coefficient on V is larger than U does not technically imply that variations in U are less important in explaining wage growth: if U has a much higher variance than V, it can have a small coefficient while still playing a large role. To show more formally how rising $\left|\frac{\phi_V}{\phi_U}\right|$ diminishes the importance of unemployment in

⁹With
$$\lambda_{EE} = 0$$
, $\theta_t = \frac{V_t}{S_t} = \frac{V_t}{U_{t-1}}$.

Figure D.2: In Economies with More On-the-job Search, Vacancies Matter More in the Wage Phillips Curve



Notes: The red, starred line plots the effects of a change in on-the-job search intensity λ_{EE} , holding all other model parameters constant at their values in Table 2, on the ratio of the coefficients in equation (D.1): $\check{\Pi}_t^w = \phi_V \check{V}_t + \phi_U \check{U}_{t-1} + \frac{1}{1+\rho} \check{\Pi}_{t+1}^w$. The blue dotted line repeats the exercise but with $\chi = 0$, a linear cost of vacancy posting. The vertical line marks the value for λ_{EE} used in our benchmark calibration. The relative importance of vacancies in explaining wage inflation, compared with unemployment, increases with both λ_{EE} and χ .

explaining wage growth, consider the variance decomposition of wage growth in the model under the assumption that we can ignore the inflation expectations term:¹⁰

$$\operatorname{Var}\left(\check{\Pi}_{t}^{w}\right) \propto \left(\frac{\phi_{V}}{\phi_{U}}\right)^{2} \operatorname{Var}\left(\check{V}_{t}\right) + \operatorname{Var}\left(\check{U}_{t-1}\right) + \underbrace{2\frac{\phi_{V}}{\phi_{U}}}_{\leq 0} \operatorname{Cov}\left(\check{V}_{t},\check{U}_{t-1}\right) \tag{D.2}$$

Now consider Figure D.2, which increases λ_{EE} holding other parameters constant, raising $\left|\frac{\phi_V}{\phi_U}\right|$. Given that the covariance term Cov $(\check{V}_t, \check{U}_{t-1})$ in (D.2) is strongly negative (both empirically, and also in any reasonably calibrated model), the importance of unemployment in explaining wage growth falls monotonically as we increase the amount of on-the-job search,

¹⁰For example, assuming firms have constant inflation expectations, $\mathbf{E}_t \Pi_{t+1}^w = \Pi^w$, or "adaptive" expectations $\mathbf{E}_t \Pi_{t+1}^w = \Pi_t^w$. Alternatively, we might view (D.2) as an approximation when ρ is high, permitting us to ignore the many covariance cross-terms complicating the expression when solved forward.

and convexity of the vacancy costs, in the model.

Appendix EComparing Empirical and Model-Implied WagePhillips Curves in the Model Extension

In this section, we linearize the non-linear wage Phillips curve based on the household block with inflation-indexed unemployment benefits of Section 4.2 and re-derive the log-linearized wage Phillips curve in terms of observable labor market variables. Because many labor market variables are linear combinations of each other in the log-linearized version of the model, there are many ways to express our wage Phillips curve in terms of observable variables. We focus on two specifications: in the first specification, wages depend on the job vacancy rate, the unemployment rate, and a "real wage term". This is an interesting specification because we can investigate whether the coefficients on vacancies and unemployment are symmetric (equal in magnitude but oppositely signed) as would be the case in search models without on-the-job search, such as Gagliardone and Gertler (2023). The second specification writes wage growth as a function of quits (instead of vacancies), the unemployment rate, and a real wage term, allowing us to evaluate the role of quits in our model and compare our model to the motivating evidence in Section 2.

Our calibrated model captures three facts regarding the wage Phillips curve: (i) in a regression with vacancies and unemployment, the weight on vacancies is larger than the weight on unemployment;¹¹ (ii) in a regression with quits and unemployment, quits dominates and the coefficient on unemployment is near zero; and (iii) the "catch up" of nominal wages after a cost-of-living shock raises prices and pushes down the real wage is very weak, in line with Bernanke and Blanchard (2024). All these coefficients in the wage Phillips curve are untargeted moments, as the model was calibrated only to match labor market flows and the sensitivity of workers' mobility decisions to relative wages.

Linearized Wage Phillips Curve with Vacancies We begin by presenting the log-linearized wage Phillips curve, where nominal wage growth is a function of log deviations in the vacancy rate \check{V}_t , the unemployment rate \check{U}_t , and a real wage term $\check{\tilde{w}}$

¹¹A recent literature has focuses on the ratio of vacancies to unemployment as the forcing term in the Phillips curve, see (Barnichon and Shapiro, 2022; Benigno and Eggertsson, 2024). Andolfatto and Birinci (2022) argue that the appropriate level of labor market tightness should account for on-the-job searchers.

$$\check{\Pi}_{t}^{w} = \phi_{V}\check{V}_{t} + \phi_{U}\check{U}_{t-1} + \phi_{\tilde{w}}\tilde{\tilde{w}}_{t} + \frac{1}{1+\rho}\check{\Pi}_{t+1}^{w},$$
(E.1)

where $\tilde{w}_t = \sum_{s=0}^{t-1} \Pi_s^w \sum_{s=0}^t \Pi_s$ is a real wage term that takes into account the real wage last period and the realized price inflation in period t. Therefore $\phi_{\tilde{w}}$ represents direct passthrough of prices to wages; the coefficient $\phi_{\tilde{w}}$ is zero in our baseline model.¹² When we estimate the OLS regression, we truncate the real wage term to include wage growth and price inflation up to 12 quarters prior, i.e. s = t 12.

Panel A of Table E.1 reports the model-implied, structural coefficients and the estimated coefficients in regression (E.1). In both the model and the data, the coefficient on vacancies ϕ_V is positive, the coefficient on unemployment ϕ_U is negative, and the coefficient on real wages $\phi_{\bar{w}}$ is negative. Also, the magnitude of ϕ_V is larger than the magnitude of ϕ_U , suggesting that ratio of job openings to unemployed workers $\frac{V}{U}$ does not summarize the effect of labor market variables on wage growth. Instead, fluctuations in the job openings rate have bigger effects on wages than similar percent fluctuations in the unemployment rate. While these coefficients are the result of the implicit weights of vacancies and unemployment that determine the recruiting rate, separation rate, and their elasticities, there is an intuitive reason why vacancies matter more than unemployment in wage growth: unemployed workers are not the only searchers. Recall that labor market tightness is defined as $\theta = \frac{V}{S}$, where the mass of searchers S is defined as $S = U + \lambda_{EE}(1 - U)$. As such, fluctuations in U have less of a proportionate effect on tightness θ than do fluctuations in V, and consequently less of an effect on wages.¹³

Table E.1 also reports coefficients ϕ_V , ϕ_U , and $\phi_{\tilde{w}}$ for different values of χ , the convexity of vacancy costs, showing that the relative importance of V in affecting wage growth relative to U increases with the convexity of vacancy posting costs (recalling that per vacancy costs take the form of $c(\frac{V_t}{N_{t-1}})^{\chi}$). This is intuitive, as when χ is higher, each additional increment of vacancies raises the marginal cost of recruiting even more, incentivizing firms to raise wages more when firms are posting vacancies.¹⁴

The third row of Panel A of Table E.1 reports the model implied coefficients for our

¹²Derivations of (E.1) can be found in Supplementary Appendix I. Appendix B.3 derives a log-linearized wage Phillips curve in our baseline model of Section 3 without the real wage term \tilde{w} in equation (E.1).

¹³Appendix D explores how the size of on-the-job search probability λ_{EE} affects the relative sizes of coefficients ϕ_V and ϕ_U in equation (E.1) in our baseline model of Section 3 with $\phi_{\tilde{w}} = 0$.

¹⁴A popular alternative way to model turnover costs is to assume that hires, rather than vacancies, are costly: see (Pissarides, 2009; Blatter et al., 2012; Moscarini and Postel-Vinay, 2016b). However, we show in Supplementary Appendix H that adding hiring costs has minimal effect on our model-implied log-linearized wage Phillips curve.

extended model with inflation-indexed unemployment benefits, which includes a non-zero coefficient on the real wage term. This coefficient is negative, since lower real wages would imply catch-up wage growth, as high cost of living improves the relative desirability of unemployment, making recruiting and retention more difficult for firms. However, the model-implied coefficient is very small, implying that a 10% drop in the real wage would increase quarterly nominal wage growth by only 0.3%.

Table E.1: Structural Wage Phillips Curve Coefficients vs. OLS Coefficients

Panel A: Vacancies V_t and Unemployment U_{t-1}						
Source	ϕ_V	ϕ_U	$\phi_{ ilde{w}}$			
Baseline Model ($\chi = 1$)	1.84	-0.30	0			
Baseline Model ($\chi = 0$)	0.95	-0.63	0			
Real Unemployment Benefit Model ($\chi = 1$)	1.84	-0.30	030			
OLS using ECI 1990-Present	0.40***	-0.22*	019*			
	(0.12)	(0.12)	(.010)			

f and D . Out $\delta \otimes f$ and One motovine in Of	Panel B: C	Duits Q_t	and Unem	olovment	U_t	1
---	------------	-------------	----------	----------	-------	---

1 2	0 1	
eta_Q	β_U	$eta_{ ilde{w}}$
2.48	0.09	0
2.13	-0.11	0
2.48	0.09	.0426
1.11***	-0.04	021***
(0.16)	(0.07)	(.007)
	$ \begin{array}{c} \frac{\beta_Q}{2.48} \\ 2.13 \\ 2.48 \\ 1.11^{***} \\ (0.16) \end{array} $	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$

Standard errors in parentheses (Newey-West; 4 lags) *** p<0.01, ** p<0.05, * p<0.1

(1, 1, 1, 1, p<0.03, · p<0.1

Notes: These variations on the baseline model's wage Phillips curve are broadly in line with the OLS estimates from U.S. data, putting much more weight on vacancies or quits than unemployment. The table compares our OLS estimates with results from two calibrations: the baseline calibration with $\chi = 1$ (convex vacancy posting costs) and $\chi = 0$ (linear vacancy posting costs). See Table 2 for other parameter choices. The models' structural wage Phillips curves are converted into quarterly frequency.

The fourth and final rows of Panel A in Table E.1 report the estimated OLS coefficients for (E.1) using U.S. data from the second quarter of 1990 through the first quarter of 2024.¹⁵ Consistent with the model, the coefficient on V is larger in absolute value than the coefficient on U, though the difference is not quite as large as implied by the model. The coefficient

¹⁵Prior to 2001, the job openings rate comes from the Help-Wanted Index in Barnichon (2010) and Michaillat and Saez (2022). Beginning in 2001, the jobs opening rate comes from the Job Openings and Labor Turnover Survey (JOLTS).
on the catch-up term $\phi_{\tilde{w}}$ reports an elasticity of 0.019, indicating that a 10 percent decline in real wages would generate a 0.2 percent higher nominal wage growth in a quarter, which is similar to the magnitude implied by our extension with inflation-indexed unemployment insurance.

Linearized Wage Phillips Curve with Quits Returning to earlier evidence and motivation on quits and wage growth, we rewrite the wage Phillips curve so that wage growth depends on the quits rate. To write down a wage Phillips curve of a similar form as equation (1), recall that we define quits Q_t as all separations S_t less the exogenous separations s, i.e., $Q_t = S_t$ s, so quits in the model captures both voluntary job-to-job quits and voluntary quits from employment into unemployment. Since the quits rate itself is a function of vacancies, unemployment, and the real wage, the wage Phillips curve (E.1) can in turn be written in terms of quits \check{Q}_t , unemployment \check{U}_{t-1} , and real wage \check{w}_t as¹⁶

$$\check{\Pi}_{t}^{w} = \beta_{Q}\check{Q}_{t} + \beta_{U}\check{U}_{t-1} + \beta_{\tilde{w}}\tilde{\tilde{w}}_{t} + \frac{1}{1+\rho}\check{\Pi}_{t+1}^{w}$$
(E.2)

for positive $\beta_Q > 0$ and β_U of indeterminate sign which depends on the calibration (both coefficients are functions of parameters and steady state values). The exact value of β_U will depend on the relative weights on V and U in determining Π^w and Q in the log-linear model.

The possibility of a positive coefficient on the unemployment rate may seem surprising, since traditionally the coefficient on the unemployment rate in the wage Phillips curve is negative. However, this can arise because the quits rate already contains information on the unemployment rate: because workers' quit rate depends on their job finding probability $f(\theta_t)$, which itself depends on tightness θ_t and subsequently on the unemployment rate U_{t-1} , the extent that the unemployment rate affects wage growth may be entirely or more than entirely accounted for by its effect on wages through quits. Therefore after controlling for quits, the coefficient on the unemployment rate may be "wrong-signed". This happens to occur in the first and third rows of Panel B in Table E.1 when $\chi = 1$. Regardless of the sign of β_U , for reasonable calibrations we find that β_U is very close to zero in the model once also conditioning on the quits rate, which is consistent with the empirical estimates in the fourth row of Panel B, as well as in Table 1. This is in line with our explanation in Section 3.3.

Just as the inclusion of the quit rate can flip the sign of β_U , including the quits rate may

¹⁶For this part of the derivation, i.e., expressing \check{Q}_t in terms of \check{V}_t , \check{U}_{t-1} , and \check{w}_t , see Supplementary Appendix I; see Appendix B.3.1 for the same derivation in the baseline model: there, \check{Q}_t , will depend only on \check{V}_t and \check{U}_{t-1} .

also flip the sign of $\beta_{\bar{w}}$, at least in theory. If unemployment benefits are inflation indexed, then an increase in the cost of living will lower real wages, make recruiting and retaining workers more difficult for firms, and push nominal wages up. However, this effect will be partially reflected in the quit rate: when real wages fall and unemployment becomes relatively more desirable, quits will rise in response. Therefore, as was the case with unemployment, there is information about the level of the real wage embedded in the quits rate. As a consequence, once we account for the quits rate, the coefficient on the real wage term may become positive. As the fourth row of Panel B shows, the OLS regression with quits, unemployment, and the real wage term still produces a negative coefficient of -0.021, which is not predicted under our model calibration. However, the coefficient is still quantitatively small, consistent with very weak pass-through of cost-of-living shocks to wages.

In total, the wage Phillips curve predicted by our model matches three facts about the US wage Phillips curve since 1990: (i) in a regression with vacancies and unemployment, the weight on vacancies is larger than the weight on unemployment; (ii) in a regression with quits and unemployment, quits dominates and the coefficient on unemployment is near zero; and (iii) the catch up of wages after higher cost of living pushes down the real wage is very weak, which is in line with Bernanke and Blanchard (2024). These results, combined with the model matching microeconomic evidence on how wages are determined, lead us to conclude that our model is a good representation of wage determination and the wage Phillips curve for the United States.

Supplementary Appendix for Firm Wage Setting and On-the-Job Search Limit Wage-Price Spirals

JUSTIN BLOESCH SEUNG JOO LEE JACOB P. WEBER

Appendix F Cost-of-Living Shocks and Forward-Looking Workers

This section relaxes the assumption that workers are completely myopic. While relaxing this assumption increases the complexity of the model, we show here that it barely matters for the dynamics: the response of wage inflation and other key labor market variables to a cost-of-living shock is nearly identical, assuming the monetary authority stabilizes output as well as inflation.

To understand why, note that the only part of the model that changes is the firm's problem, and more specifically, the first order condition for wages. Even relaxing myopia, it continues to be the case that stabilizing output Y_t (and hence N_t) stabilizes the firm's first order condition for wages and hence wage inflation. However, this is in spite of the fact that this first order condition becomes much more complicated because separation and recruiting rates at time t now depend not just on the wage at time t, but also on the wage promised for all T > t. The reason for this is that now when a worker considers joining a firm, they know they may remain there for more than one period. They thus care not just about the wage offered today, but also about the wage promised in the future.

Technically speaking, this makes the firm's problem *time-inconsistent*: when making a plan for wages, it is optimal to promise to offer a high wage in the future in order to raise the recruiting rate today "for free." But when the time arrives to pay the higher wage, the firm will be tempted to renege. This shows up in the firm's problem as a first order condition for wages which is non-stationary, as the first period is special, which introduces additional difficulty both in solving and discussing the model.

To make progress, and keep the model with forward looking workers as similar as possible to that in the main text to ease comparison, we again study the effect of an unanticipated "MIT shock" to the cost of living that becomes known at some time t = 0. At that date, we

assume that firms are paying wages that they committed to at some point in the distant past, $t = -\infty$ so that at t = -1 the economy has converged to a long run steady state which we will describe below.

In letting firms respond to this MIT shock, we solve for the responses that the firms would have found optimal had they known about the shock back in the infinite past. More formally, we assume firms knew at $t = -\infty$ that there was some chance p of the shock occurring at time t, and solve for the economy's response to the shock under that optimal plan as $p \to 0$, so that the shock is entirely unanticipated.¹⁷

As in the main text, we will solve for a symmetric, perfect-foresight equilibrium where all firms set the same wage. Unlike in the main text, we will solve for the solution to the nonlinear models (with and without myopia) to better compare their behavior and make the point that the assumption of myopic workers is largely-innocuous simplifying assumption.

F.1 Firm's Problem With Dynamic Separation and Recruiting Rates

Recall that the only choice facing workers occurs when they are offered a chance to take a job. With probability $\lambda_{EE} f(\theta_t)$, workers meet alternative jobs, and with probability λ_{EU} they are allowed to consider quitting into unemployment. When workers meet alternate jobs, or consider the unemployment state, they draw idiosyncratic utilities ι from a type-1 extreme value distribution with scale parameter γ . They then choose the state which yields the greatest utility: for example, let \mathcal{V}_{jt} be the value of being employed at firm j and $\overline{\mathcal{V}}_t$ be the value of being employed at the other matched firm. Then a worker at firm j who matches with another firm of value $\overline{\mathcal{V}}_t$ maximizes:

$$\max\{\mathcal{V}_{jt}+\iota_{jt},\bar{\mathcal{V}}_t+\iota_{kt}\}.$$

Given this, the expected value of having a choice between two value functions next period is $\mathcal{V}^{E}(\mathcal{V}^{a}, \mathcal{V}^{b}) = \frac{1}{\gamma} \ln \left(\exp \left(\gamma \mathcal{V}^{a} \right) + \exp \left(\gamma \mathcal{V}^{b} \right) \right)$. So the worker's Bellman equation at firm j is thus the following: letting \mathcal{V}_{t}^{u} be the value of the unemployed state at time t, and assuming all firms other than j have identical value \bar{V}_{t} (which is without loss of generality, given that

¹⁷In the context of time-inconsistent optimal monetary policy, this is equivalent to the "timeless approach" to computing optimal policy under commitment.

we will search for a symmetric equilibrium later),

$$\begin{split} \mathcal{V}_{jt} &= \ln\left(\frac{\tau_t W_{jt}}{P_t}\right) + \beta s \mathcal{V}_{t+1}^u \\ &+ \beta (1-s) \left[\lambda_{EE} f(\theta_{t+1}) \mathcal{V}^E(\mathcal{V}_{j,t+1}, \bar{\mathcal{V}}_{t+1}) + \lambda_{EU} \mathcal{V}^E(\mathcal{V}_{j,t+1}, \mathcal{V}_{t+1}^u) + (1-\lambda_{EE} f(\theta_{t+1}) - \lambda_{EU}) \mathcal{V}_{j,t+1}\right] \\ &\equiv \mathcal{F}_t(\mathcal{V}_{j,t+1}) \end{split}$$

where we write this function \mathcal{F}_t as time-varying because of its dependence on the real wage w_t , tax rate 1 τ_t , tightness θ_{t+1} , and $\overline{\mathcal{V}}_{t+1}$. We also assume workers discount the future at rate β , which need not equal $\frac{1}{1+\rho}$. This allows us to easily nest the myopic worker case where $\frac{1}{1+\rho} > \beta = 0$. In practice, we solve a version of model where workers discount the future with $\beta = .96$ which corresponds to an annual discount rate of $\beta^{12} \approx .60$, consistent with experimental evidence; see Michaillat and Saez (2021) for a discussion and summary of this evidence. So we can plug in the following:

$$\mathcal{V}^{E}(\mathcal{V}_{j,t+1},\bar{\mathcal{V}}_{t+1}) = \frac{1}{\gamma} \ln\left(\exp\left(\gamma \mathcal{V}_{j,t+1}\right) + \exp\left(\gamma \bar{\mathcal{V}}_{t+1}\right)\right)$$
$$\mathcal{V}^{E}(\mathcal{V}_{j,t+1},\mathcal{V}_{t+1}^{u}) = \frac{1}{\gamma} \ln\left(\exp\left(\gamma \mathcal{V}_{j,t+1}\right) + \exp\left(\gamma \mathcal{V}_{t+1}^{u}\right)\right)$$

To see that \mathcal{V}_{jt} depends on the whole path of nominal wages, $\{W_{jt}\}_{t=0}^{\infty}$, we evaluate the following derivative: for $S \ge 1$,

$$\frac{d\mathcal{V}_{jt}}{dW_{j,t+\$}} = \frac{d}{dW_{j,t+\$}} \mathcal{F}_t \left(\mathcal{F}_{t+1} \left(\dots \mathcal{F}_{t+\$-1} (\mathcal{V}_{j,t+\$}) \dots \right) \right)$$
$$= \mathcal{F}'_t (\mathcal{V}_{j,t+1}) \times \mathcal{F}'_{t+1} (\mathcal{V}_{j,t+2}) \times \dots \times \mathcal{F}'_{t+\$-1} (\mathcal{V}_{j,t+\$}) \frac{d\mathcal{V}_{j,t+\$}}{dW_{j,t+\$}}$$

where

$$\mathcal{F}'_{t}(\mathcal{V}_{j,t+1}) = \beta \begin{pmatrix} 1 & s \end{pmatrix} \left[\frac{\lambda_{EE} f(\theta_{t+1})}{1 + \exp\left(\gamma(\bar{\mathcal{V}}_{t+1} - \mathcal{V}_{j,t+1})\right)} + \frac{\lambda_{EU}}{1 + \exp\left(\gamma(\mathcal{V}_{t+1}^{u} - \mathcal{V}_{j,t+1})\right)} + 1 - \lambda_{EE} f(\theta_{t+1}) - \lambda_{EU} \right]$$

and since the final term $\frac{dV_{j,t+\$}}{dW_{j,t+\$}} = \frac{1}{W_{j,t+\$}} > 0$, we can see that $\frac{dV_{j,t}}{dW_{j,t+\$}} > 0$. Promising a higher wage in the future makes a job offer at firm j more attractive, and helps recruit workers today.

In a symmetric equilibrium with $\bar{\mathcal{V}}_t = \mathcal{V}_{jt}$ always, and where the household fixes $\bar{\mathcal{V}}_t - \mathcal{V}_t^u$

at some value $\ln \xi$ (by varying the consumption of the unemployed),¹⁸ this simplifies greatly to

$$\mathcal{F}'_t(\mathcal{V}_{j,t+1} = \bar{\mathcal{V}}_{t+1}) = \beta(1 \quad s) \underbrace{\left[\frac{\lambda_{EE}f(\theta_{t+1})}{2} + \frac{\lambda_{EU}}{1+\xi^{-\gamma}} + 1 \quad \lambda_{EE}f(\theta_{t+1}) \quad \lambda_{EU}\right]}_{\equiv P(\theta_{t+1}) < 1}$$

Thus in a symmetric equilibrium, and for $S \ge 1$ we can write down these derivatives as:

$$\frac{d\mathcal{V}_{jt}}{dW_{j,t+\mathfrak{S}}} = \frac{\left(\beta(1-s)\right)^{\mathfrak{S}}\prod_{k=1}^{\mathfrak{S}}P(\theta_{t+k})}{W_{j,t+\mathfrak{S}}},$$

and for S = 0:

$$\frac{d\mathcal{V}_{jt}}{dW_{j,t}} = \frac{1}{W_{j,t}}.$$

So we obtain

$$P(\theta_{t+1}) = \frac{\lambda_{EE} f(\theta_{t+1})}{2} + \frac{\lambda_{EU}}{1 + \xi^{-\gamma}} + 1 \quad \lambda_{EU}$$

which is decreasing in θ_{t+1} . It can be understood as follows: as future market tightness θ_{t+1} increases, it is easier for workers to switch firms in which they work in the future, so a future wage increase leads to less increase in the current value \mathcal{V}_{jt} at period t.

Now that we know the value function \mathcal{V}_{jt} depends on the whole path of wages, we rewrite our recruiting and separation rates as follows:

$$R_{t}(\mathcal{V}_{jt}) \equiv g(\theta_{t}) \left[\phi_{E,t} \left(\frac{\exp(\gamma(\mathcal{V}_{jt} \quad \bar{\mathcal{V}}_{t}))}{1 + \exp(\gamma(\mathcal{V}_{jt} \quad \bar{\mathcal{V}}_{t}))} \right) + \phi_{U,t} \left(\frac{\exp(\gamma(\mathcal{V}_{jt} \quad \mathcal{V}_{t}^{u}))}{1 + \exp(\gamma(\mathcal{V}_{jt} \quad \mathcal{V}_{t}^{u}))} \right) \right]$$
$$S_{t}(\mathcal{V}_{jt}) \equiv s + (1 \quad s) \left[\lambda_{EE} f(\theta_{t}) \left(\frac{\exp(\gamma(\mathcal{V}_{jt} \quad \bar{\mathcal{V}}_{t}))}{1 + \exp(\gamma(\mathcal{V}_{jt} \quad \bar{\mathcal{V}}_{t}))} \right) + \lambda_{EU} \left(\frac{\exp(\gamma(\mathcal{V}_{jt} \quad \mathcal{V}_{t}^{u}))}{1 + \exp(\gamma(\mathcal{V}_{jt} \quad \mathcal{V}_{t}^{u}))} \right) \right]$$

Where we write that R_t and S_t are time varying because of changes labor market conditions (tightness θ_t and employed/unemployed searcher shares $\phi_{E,t}$ and $\phi_{U,t}$) and competition from both other firms and unemployment ($\bar{\mathcal{V}}_t$ and \mathcal{V}_t^u).

Note that this impacts only the FOCs for vacancies and wages, and the law of motion for employment. We can no longer derive a nice nonlinear wage Phillips Curve (though our price Phillips curve is, happily, unchanged). Putting changes in red, the firm's problem is

¹⁸With log utility and myopia ($\beta = 0$), we have $\bar{\mathcal{V}}_t = \ln C_t^e = \ln C_t^e = \ln \xi$ implying $\frac{C_t^e}{C_t^u} = \xi$ as before.

now

$$\max_{\substack{\{P_{y,t}^{j}\},\{N_{jt}\}\\\{W_{jt}\},\{V_{j,t}\}}} \sum_{t=0}^{\infty} \left(\frac{1}{1+\rho}\right)^{t} \left(P_{y,t}^{j}Y_{t}^{j} - W_{jt}N_{jt} - c\left(\frac{V_{j,t}}{N_{j,t-1}}\right)^{\chi}V_{j,t}W_{t} - \frac{\psi}{2}\left(\frac{P_{y,t}^{j}}{P_{y,t-1}^{j}} - 1\right)^{2}Y_{t}^{j}P_{y,t}^{j} \\
\frac{\psi^{w}}{2}\left(\frac{W_{jt}}{W_{j,t-1}} - 1\right)^{2}W_{jt}N_{jt}\right)$$

subject to

$$N_{jt} = (1 \quad S_t(\boldsymbol{\mathcal{V}}_{jt})) N_{j,t-1} + R_t(\boldsymbol{\mathcal{V}}_{jt}) V_{j,t}.$$

Output is produced with labor with the linear production: $Y_t^j = A_t^j N_{jt}$, and Dixit-Stiglitz demand, so $\frac{Y_t^j}{Y_t} = \begin{pmatrix} \frac{P_{y,t}^j}{P_{y,t}} \end{pmatrix}^{\epsilon}$, hence $N_{jt} = \begin{pmatrix} \frac{P_{y,t}^j}{P_{y,t}} \end{pmatrix}^{\epsilon} \frac{Y_t}{A_t^j}$ with $\epsilon > 1$.

Here we can see why only the first order condition with respect to wages will change: the only new component is that the derivatives of the recruiting and separation rates R_t and S_t will be different. Note that their levels will be unchanged: in a symmetric equilibrium, R_t and S_t will have the same functional forms as before, in the main text: formally in a symmetric equilibrium with $\mathcal{V}_{jt} = \bar{\mathcal{V}}_t$, and assuming households set $\bar{\mathcal{V}}_t = \ln \xi$ by adjusting the consumption of unemployed households, i.e., adjusting τ_t , we have:

$$R_{t} \equiv g(\theta_{t}) \left[\phi_{E,t} \left(\frac{1}{2} \right) + \phi_{U,t} \left(\frac{\xi^{\gamma}}{1 + \xi^{\gamma}} \right) \right]$$
$$S_{t} \equiv s + (1 \quad s) \left[\lambda_{EE} f(\theta_{t}) \left(\frac{1}{2} \right) + \lambda_{EU} \left(\frac{\xi^{-\gamma}}{1 + \xi^{-\gamma}} \right) \right].$$

Note that we continue to assume that the representative household imposes taxes and transfers to satisfy a standard Euler equation: while this may no longer necessarily be optimal, we make this assumption to facilitate direct comparison with the model in the main text.

Equilibrium and Steady State We focus on one particular equilibrium where $P_{y,t}^j = P_{y,t}$, $V_{j,t} = V_t$, $W_{jt} = W_t$, $A_t^j = A_t \forall j$. All equations in our model in the main body, and the firms problem, are unchanged except for the first order condition for the wage, W_t : defining

for convenience

$$\Delta_t \equiv N_t + \psi^w (\Pi_t^w \quad 1) \Pi_t^w N_t \quad \frac{1}{1+\rho} \psi^w (\Pi_{t+1}^w \quad 1) (\Pi_{t+1}^w)^2 N_{t+1},$$

$$\Lambda_t \equiv \frac{\lambda_t}{P_{y,t}} (V_t R'_t \quad S'_t N_{t-1})$$

where λ_t is the co-state variable associated with the law of motion for N_t , we can write the FOC for wage W_t as

for
$$t = 0$$
: $\frac{W_0}{P_{y,0}}\Delta_0 = \Lambda_0$
for $t \ge 1$: $\frac{W_t}{P_{y,t}}\Delta_t = \Lambda_t + \beta(1-s)\frac{P(\theta_t)}{\Pi_{y,t}}\left(\frac{W_{t-1}}{P_{y,t-1}^y}\Delta_{t-1}\right)$

In the long run, assuming that firms committed to a wage plan arbitrarily far in the past $(t = \infty)$, we should converge to a steady state satisfying the FOC for W_t with $t \ge 1$, i.e. not the equation in the initial period. Thus we look for a solution which solves the following: in a zero-inflation steady state, and noting that $P(\theta_t) < 1$, the wage satisfies:

$$\frac{W}{P^y}\Delta = \frac{\Lambda}{1 - \beta(1 - s)P(\theta)}$$

Note that if $\beta = 0$, and workers are myopic, we recover the expression for the real wage in steady state for the myopic worker model in the main body. Given this equation, and all the other equations in the model, we can again solve numerically for a zero-inflation steady state of the model (assuming the monetary authority targets $\Pi_t = 1$).

This makes it easy to see the time inconsistency in the firm's optimal wage plan: if we plug in the long run steady state here for the t = 0 constraint, and consider what firms choose for the real wage given Λ, Δ, θ , we see that the long run steady state does not satisfy the FOC at t = 0: the wage is too high by a factor of $1 \quad \beta(1 \quad s)P(\theta)$. In short, given the chance to re-optimize, firms choose a lower wage than the one committed to in the infinite past, because that future commitment once helped with contemporary recruitment.

Convergence to the Long Run Steady State Assume an economy as described above without aggregate risk and where firms made their wage plans an infinitely long time ago, so that the economy is now in the long run steady state described above. When firms first set their wage plan, they initially choose to promise a wage below the long run value, and later

would want to renege, but this is not permitted. Figure F.3 plots the transition to this long run steady state if we allow firms to choose new wage plans. All impulse responses are shown as percent deviations from the long-run steady state, so that we can confirming the intuition described above for the wage. Re-optimization acts like an expansionary shock: firms immediately lower nominal wages, but increase in size by recruiting through promising higher nominal wages in the future and by posting more vacancies. This causes market tightness to rise in the aggregate, which puts upward pressure on firms' marginal costs. The net effect is a modest amount of inflation as these costs are passed on to consumers, which the monetary authority responds to by raising real interest rates.

Response to an MIT Cost-of-Living Shock X_t When firms commit to their wage plan at some time $t = \infty$, we assume that firms know that an X_t MIT shock might hit the economy at date t = 0, but that the probability of that MIT shock is effectively zero (in the limit). However, we can characterize what their wage plan looks like in the case that the shock hits: this is plotted in Figure F.4. That wage plan respects the FONC for wages for $t \ge 1$, not t = 0, written above.

Replicating the experiment in the main text, where the central bank perfectly stabilizes $N_t = N$ in response to the shock, yields identical impulse response functions for both models. Because of this, we instead plot the slightly more interesting case where the monetary authority follows an active Taylor rule that imperfectly stabilizes output, and hence domestic employment: $1 + i_t = (1 + \rho) \left(\frac{\Pi_t}{\Pi}\right)^{\phi_{\Pi}} \left(\frac{Y_t}{Y}\right)^{\phi_Y}$, where here $\phi_Y = \phi_{\Pi} = 2$. The responses in both models remain very similar, though they are no longer identical: as the monetary authority responds to the inflationary shock by raising real interest rates, the price of domestic output falls and domestic consumption Y_t remains basically flat. Wage inflation remains extremely modest, but positive—and is slightly more positive on impact for the "myopic" model.

We conclude by noting that the assumption of myopic workers is a largely-innocuous simplifying assumption which is not critical to obtaining the results in the main text.



Nonlinear Dynamic Model Response to a One-Time Firm Reoptimization Allowed at t = 0

Figure F.3: The effects of allowing firms to reoptimize and choose new paths for wages and all other choice variables, which they then commit to following forever. All impulse responses are shown as percent deviations from the long run steady state.



Nonlinear Model Response to an MIT X_t Shock: Myopic vs. Forward-Looking Workers

Figure F.4: The effects of a negative shock to X_t assuming the central bank follows an active Taylor rule: $1 + i_t = (1 + \rho) \left(\frac{\Pi_t}{\Pi}\right)^{\phi_{\Pi}} \left(\frac{Y_t}{Y}\right)^{\phi_Y}$, where here $\phi_Y = \phi_{\Pi} = 2$. The responses in both models are very similar: the monetary authority responds to the inflationary shock by raising real interest rates. The price of domestic output falls and domestic consumption Y_t remains basically flat. Wage inflation remains extremely modest, but positive—and is slightly more positive on impact for the "myopic" model. All impulse responses are shown as percent deviations from the long run steady state.

Appendix G Constant Relative Risk Aversion (CRRA) Utility

In this section, we deviate from our log-preference assumption and assume instead that the per-period utility function is given by $\frac{C_t^{1-\sigma}}{1-\sigma}$ featuring σ as relative risk aversion and the inverse of elasticity of intertemporal substitution. A worker working at firm j, receiving wage W_{jt} and facing τ_t as tax rate, will consume $C_t^e = \tau_t \frac{W_{jt}}{P_t}$. Under the same consumption sharing rule $\frac{C_t^e}{C_t^u} = \xi$ within a household, an unemployed person consumes $C_t^u = \frac{\tau_t}{\xi} \frac{W_t}{P_t}$, where $\overline{W_t}$ is the average wage of employed workers as defined in Section 3.2.

Now, the probability that a worker chooses firm j paying W_{jt} relative to market wage \overline{W}_t is given by

$$r_{mj}\left(\bar{W}_{t}, W_{jt}|P_{t}\right) = \frac{e^{\frac{\gamma}{1-\sigma}\left(\frac{\tau_{t}W_{jt}}{P_{t}}\right)^{1-\sigma}}}{e^{\frac{\gamma}{1-\sigma}\left(\frac{\tau_{t}W_{jt}}{P_{t}}\right)^{1-\sigma}} + e^{\frac{\gamma}{1-\sigma}\left(\frac{\tau_{t}\bar{W}_{t}}{P_{t}}\right)^{1-\sigma}}} = \frac{1}{1 + e^{\frac{\gamma}{1-\sigma}\left(\frac{\tau_{t}}{P_{t}}\right)^{(1-\sigma)}\left(\bar{W}_{t}^{1-\sigma} - W_{jt}^{1-\sigma}\right)^{(1-\sigma)}}}$$

where subscript m denotes market, the recruiting rate r_{mj} depends P_t explicitly. The probability that a unemployed worker chooses firm j paying W_{jt} is

$$r_{uj}\left(\bar{W}_{t}, W_{jt}|P_{t}\right) = \frac{e^{\frac{\gamma}{1-\sigma}\left(\frac{\tau_{t}W_{jt}}{P_{t}}\right)^{1-\sigma}}}{e^{\frac{\gamma}{1-\sigma}\left(\frac{\tau_{t}W_{jt}}{P_{t}}\right)^{1-\sigma}} + e^{\frac{\gamma}{1-\sigma}\left(\frac{\tau_{t}\bar{W}_{t}}{\xi P_{t}}\right)^{1-\sigma}}} = \frac{1}{1+e^{\frac{\gamma}{1-\sigma}\left(\frac{\tau_{t}}{P_{t}}\right)^{(1-\sigma)}\left(\left(\frac{\bar{W}_{t}}{\xi}\right)^{1-\sigma} W_{jt}^{1-\sigma}\right)}}$$

which again depends directly on P_t . Under the symmetric equilibrium with $W_{jt} = \bar{W}_t$, a rise in P_t raises the recruiting rate r_{uj} from the unemployed, when $\sigma > 1$, giving an incentive for firms to post higher wages. Then, the recruiting function $R(W_{jt}|P_t)$ is given by

$$R_{t}(W_{jt}|P_{t}) = g(\theta_{t}) \left[\phi_{E,t} \frac{1}{1 + e^{\frac{\gamma}{1-\sigma} \left(\frac{\tau_{t}}{P_{t}}\right)^{(1-\sigma)} \left(\bar{W}_{t}^{1-\sigma} - W_{jt}^{1-\sigma}\right)}} + \phi_{U,t} \frac{1}{1 + e^{\frac{\gamma}{1-\sigma} \left(\frac{\tau_{t}}{P_{t}}\right)^{(1-\sigma)} \left(\left(\frac{\bar{W}_{t}}{\xi}\right)^{1-\sigma} - W_{jt}^{1-\sigma}\right)}} \right]$$

Then we have that $R'(w_{jt})W_{jt}$ is given by

$$\begin{aligned} R'_{t}(W_{jt})W_{jt} = g(\theta_{t})\phi_{E,t} \left(1 + e^{\frac{\gamma}{1-\sigma}\left(\frac{\tau_{t}}{P_{t}}\right)^{(1-\sigma)}\left(\bar{W}_{t}^{1-\sigma} - W_{jt}^{1-\sigma}\right)}\right)^{-2} e^{\frac{\gamma}{1-\sigma}\left(\frac{\tau_{t}}{P_{t}}\right)^{(1-\sigma)}\left(\bar{W}_{t}^{1-\sigma} - W_{jt}^{1-\sigma}\right)}\gamma \left(\frac{\tau_{t}W_{jt}}{P_{t}}\right)^{1-\sigma} \\ + g(\theta_{t})\phi_{U,t} \left(1 + e^{\frac{\gamma}{1-\sigma}\left(\frac{\tau_{t}}{P_{t}}\right)^{(1-\sigma)}\left(\left(\frac{\bar{W}_{t}}{\xi}\right)^{1-\sigma} - W_{jt}^{1-\sigma}\right)}\right)^{-2} e^{\frac{\gamma}{1-\sigma}\left(\frac{\tau_{t}}{P_{t}}\right)^{(1-\sigma)}\left(\left(\frac{\bar{W}_{t}}{\xi}\right)^{1-\sigma} - W_{jt}^{1-\sigma}\right)}\gamma \left(\frac{\tau_{t}W_{jt}}{P_{t}}\right)^{1-\sigma} \end{aligned}$$

which increases in P_t in the symmetric equilibrium when $\sigma > 1$.¹⁹ As the recruiting (and separation) elasticities ε_{R,W_t} and ε_{S,W_t} are increasing in P_t , a cost-of-living shock can provide an incentive for firms to raise wages in response in this case.

As special case, if s = 0, $\lambda_{EU} = 0$, then $\phi_{E,t} = 1$ and we obtain

$$\frac{\partial \varepsilon_{R,W_t}}{\partial P_t} \frac{P_t}{\varepsilon_{R,W_t}} = \frac{\partial \varepsilon_{S,W_t}}{\partial P_t} \frac{P_t}{\varepsilon_{S,W_t}} = \sigma \quad 1 > 0.$$

Euler Equation In this case, the household's consumption Euler equation takes a slightly different form. First, their preference is given by

$$\sum_{t=0}^{\infty} \left(\frac{1}{1+\rho}\right)^t \left[U_t \frac{(C_t^u)^{1-\sigma}}{1-\sigma} + (1-U_t) \frac{(C_t^e)^{1-\sigma}}{1-\sigma} \right].$$
 (G.1)

We keep assuming that the household is constrained by fairness considerations to choose $\frac{C_t^e}{C_t^u} = \xi$.

First, from the aggregate consumption, we obtain

$$C_t = (1 \quad U_t)C_t^e + U_tC_t^u = ((1 \quad U_t)\xi + U_t)C_t^u$$

leading to

$$C_t^u = \frac{C_t}{(1 \quad U_t)\xi + U_t}$$

Now the household's per-period utility in (G.1) can be written as

$$U_{t} \frac{(C_{t}^{u})^{1-\sigma}}{1-\sigma} + (1-U_{t}) \frac{(C_{t}^{e})^{1-\sigma}}{1-\sigma} = \frac{(C_{t}^{u})^{1-\sigma}}{1-\sigma} \begin{bmatrix} U_{t} + (1-U_{t}) \begin{pmatrix} \underbrace{C_{t}^{e}}{C_{t}^{u}} \\ \underbrace{-\xi} \end{pmatrix}^{1-\sigma} \end{bmatrix}$$
$$= \frac{(C_{t})^{1-\sigma}}{1-\sigma} \cdot \frac{U_{t} + (1-U_{t})\xi^{1-\sigma}}{[U_{t} + (1-U_{t})\xi]^{1-\sigma}}$$

Thus the household effectively maximizes:

$$\sum_{t=0}^{\infty} \left(\frac{1}{1+\rho}\right)^t \left[\frac{(C_t)^{1-\sigma}}{1-\sigma} \cdot \frac{U_t + (1-U_t)\xi^{1-\sigma}}{\left[U_t + (1-U_t)\xi\right]^{1-\sigma}}\right]$$

¹⁹We can prove this property similarly for the separation function $S_t(W_{jt})$.

subject to

$$C_{t} = (1 \quad U_{t})\frac{W_{t}}{P_{t}} + NWI_{t} \quad B_{t} + (1 + r_{t-1,t})B_{t-1}$$

by choosing real bonds B_t and consumption C_t , where income including real non-wage income NWI_t (dividends paid out by firms) and real wage income $\frac{W_t}{P_t}$ is taken as given by the household, $r_{t-1,t}$ is the real rate between t = 1 and t, and B_t are real bonds in zero net supply. Optimization requires that the household's choices obey the following consumption Euler equation:

$$\frac{C_t^{\sigma}}{P_t} \cdot \underbrace{\frac{U_t + (1 \quad U_t)\xi^{1 \quad \sigma}}{\left[U_t + (1 \quad U_t)\xi\right]^{1 \quad \sigma}}}_{\equiv f(U_t)} = \frac{1}{1 + \rho} (1 + i_{t,t+1}) \frac{C_{t+1}^{\sigma}}{P_{t+1}} \cdot \underbrace{\frac{U_{t+1} + (1 \quad U_{t+1})\xi^{1 \quad \sigma}}{\left[U_{t+1} + (1 \quad U_{t+1})\xi\right]^{1 \quad \sigma}}}_{\equiv f(U_{t+1})} \quad (G.2)$$

Note that (G.2) becomes a standard Euler equation

$$\frac{C_t^{\sigma}}{P_t} = \frac{1}{1+\rho} (1+i_{t,t+1}) \frac{C_{t+1}^{\sigma}}{P_{t+1}}$$
(G.3)

when the labor market is stabilized by monetary policy, i.e., $U_t = U_{t+1} = \overline{U}$ for $\forall t$. As in Section 4.1.1, here we assume that the monetary policy stabilizes labor market, i.e., (G.2) and (G.3) are both satisfied.



Figure G.5: Impulse Response to a 10% Negative Shock to Supply of Endowment Good

Figure G.5 illustrates that (i) when $\sigma > 1$, wage inflation can rise in response to a pure cost-of-living shock, as it raises the recruiting and separation elasticities and firms are thus willing to offer higher wages in equilibrium; (ii) but still the magnitude of a rise in wage

growth under our preferred calibration is small: in response to around 2% price increase, a rise in wage growth is less than 0.1%.

Monetary Policy Equation (G.3) when the labor market is stabilized can be written as

$$i_{t,t+1} = (1+\rho) \cdot \frac{P_{t+1}}{P_t} \cdot \left(\frac{C_{t+1}}{C_t}\right)^{\sigma} = (1+\rho) \cdot \frac{P_{t+1}C_{t+1}}{P_tC_t} \cdot \left(\frac{C_{t+1}}{C_t}\right)^{\sigma-1}.$$
 (G.4)

In our Dixit-Stiglitz structure with $\eta = 1$, we have constant expenditure shares on endowment good X and service good Y, i.e., $\alpha_Y P_t C_t = P_{y,t} Y_t$ for $\forall t$. With $N_t = Y_t = \overline{N}$ for $\forall t$, it implies

$$\frac{P_{t+1}C_{t+1}}{P_tC_t} = \frac{P_{Y,t+1}}{P_{Y,t}}$$

which if plugged into (G.4) leads to

$$i_{t,t+1} = (1+\rho) \cdot \underbrace{\frac{P_{Y,t+1}}{P_{Y,t}}}_{\equiv \Pi_{Y,t+1}} \cdot \left(\frac{C_{t+1}}{C_t}\right)^{\sigma-1}.$$

If period t is the timing of a cost-of-living shock, we have $\frac{C_{t+1}}{C_t} > 1$. If $\sigma > 1$ (i.e., elasticity of intertemporal substitution of households is less than 1), interest rate $i_{t,t+1}$ needs to rise from ρ since otherwise, households' demand for service goods at period t becomes higher than $\overline{Y} = \overline{N}$, destabilizing labor market at t. Also, since wage W_{t+1} rises due to higher price P_t raising the recruiting and separation elasticities under $\sigma > 1$, raising the marginal cost for firms, service firms raise their prices, i.e., $\Pi_{Y,t+1} > 1$. Both terms raise $i_{t,t+1}$ from ρ .

Log-Utility Case In the previous log-preference case (i.e., $\sigma = 1$), we had the following Euler equation:

$$\frac{1}{P_t C_t} = \frac{1}{1+\rho} (1+i_{t,t+1}) \frac{1}{P_{t+1} C_{t+1}}.$$
(G.5)

With interest rate pegging $i_t = \rho$ for $\forall t$, we equalize intertemporal consumption expenditure, i.e., $P_tC_t = P_{t+1}C_{t+1}$ and it equalizes intertemporal Y_t -consumption expenditure, i.e., $P_{Y,t}Y_t = P_{Y,t+1}Y_{t+1}$. Since wage stays at the steady state if labor market is stabilized under $\sigma = 1$, price $P_{Y,t}$ does not change, and labor market becomes actually stabilized.

Appendix H With Hiring Costs

Now, in addition to the direct vacancy-creating costs in the firm optimization (5), we assume that an intermediate firm pays a hiring cost which is convex in the number of new employees hired in each period. In this environment, the firm j maximizes

$$\max_{\substack{\{P_{y,t}^{j}\},\{N_{t}^{j}\}\\\{W_{t}^{j}\},\{V_{t}^{j}\}}} \sum_{t=0}^{\infty} \left(\frac{1}{1+\rho}\right)^{t} \left(P_{y,t}^{j}Y_{t}^{j} - W_{t}^{j}N_{t}^{j} - c_{v}\left(\frac{V_{t}^{j}}{N_{t-1}^{j}}\right)^{\chi_{v}}V_{t}^{j}W_{t} - \underbrace{c_{h}\left(\frac{H_{t}^{j}}{N_{t-1}^{j}}\right)^{\chi_{h}}H_{t}^{j}W_{t}}_{\text{Hiring cost}} \\
\underbrace{\frac{\psi}{2}\left(\frac{P_{y,t}^{j}}{P_{y,t-1}^{j}} - 1\right)^{2}Y_{t}^{j}P_{y,t}^{j} - \frac{\psi^{w}}{2}\left(\frac{W_{t}^{j}}{W_{t-1}^{j}} - 1\right)^{2}W_{t}^{j}N_{t}^{j}\right)}_{(\text{H.1})}$$

subject to

$$N_{t}^{j} = (1 \quad S_{t}(W_{t}^{j}))N_{t-1}^{j} + \underbrace{R_{t}(W_{t}^{j})V_{t}^{j}}_{\equiv H_{t}^{j}}.$$
 (H.2)

We will have that physical output is produced with labor with the linear production: $Y_t^j = A_t^j N_t^j$. The Lagrangian then can be written as:

where due to the Dixit-Stiglitz structure, the labor demand N_t^j is given by $N_t^j = \begin{pmatrix} P_{y,t}^j \\ \overline{P_{y,t}} \end{pmatrix}^{\epsilon} \frac{Y_t}{A_t^j}$.

First order conditions We write the first order conditions under the symmetric equilibrium, where $N_t^j = N_t$, $W_t^j = W_t$, $V_t^j = V_t$, and $\lambda_t^j = \lambda_t$. The first order condition with V_t^j

is given by

$$c_{v}(1+\chi_{V})\left(\frac{V_{t}}{N_{t-1}}\right)^{\chi_{V}}W_{t} \quad c_{h}(1+\chi_{h})\left(\frac{V_{t}}{N_{t-1}}\right)^{\chi_{h}}(R_{t}(W_{t}))^{1+\chi_{h}}W_{t} + \lambda_{t}R_{t}(W_{t}) = 0,$$
(H.3)

which leads to

$$\lambda_{t} = c_{v} \frac{(1 + \chi_{V}) \left(\frac{V_{t}}{N_{t-1}}\right)^{\chi_{V}}}{R_{t}(W_{t})} W_{t} + c_{h}(1 + \chi_{h}) \left(\frac{H_{t}}{N_{t-1}}\right)^{\chi_{h}} W_{t}, \tag{H.4}$$

where λ_t can be interpreted as a shadow value of a worker. It increases with c_h , a shifter in the hiring cost function.

The first order condition with W_t^j is given by

$$\frac{\psi^{w}}{2} \begin{pmatrix} W_{t} \\ W_{t-1} \end{pmatrix}^{2} + \psi^{w} \begin{pmatrix} W_{t} \\ W_{t-1} \end{pmatrix} \frac{W_{t}}{W_{t-1}} + 1 = \lambda_{t} \begin{pmatrix} R'_{t}(W_{t}) \frac{V_{t}}{N_{t}} & \frac{N_{t-1}}{N_{t}} S'_{t}(W_{t}) \end{pmatrix} \\ c_{h}(1+\chi_{h}) R'_{t}(W_{t}) \begin{pmatrix} H_{t} \\ N_{t-1} \end{pmatrix}^{\chi_{h}} \frac{V_{t}}{N_{t}} W_{t} \\ + \frac{1}{1+\rho} \psi^{w} \begin{pmatrix} W_{t+1} \\ W_{t} \end{pmatrix} \frac{1}{W_{t}^{2}} W_{t+1} \frac{N_{t+1}}{N_{t}}.$$
(H.5)

Combining equations (H.4) and (H.5), we obtain

$$\frac{\psi^{w}}{2} \left(\frac{W_{t}}{W_{t-1}} - 1\right)^{2} + \psi^{w} \left(\Pi_{t}^{w} - 1\right) \Pi_{t}^{w} + 1$$

$$= \left(\underbrace{W_{t}c_{v} \frac{\left(1 + \chi_{V}\right) \left(\frac{V_{t}}{N_{t-1}}\right)^{\chi_{V}}}{R_{t}(W_{t})} + W_{t}c_{h}(1 + \chi_{h}) \left(\frac{H_{t}}{N_{t-1}}\right)^{\chi_{h}}}_{=\lambda_{t}}\right) \left(R_{t}'(W_{t}) \frac{V_{t}}{N_{t}} - \frac{N_{t-1}}{N_{t}}S_{t}'(W_{t})\right)$$

$$c_{h}(1 + \chi_{h})R_{t}'(W_{t}) \left(\frac{H_{t}}{N_{t-1}}\right)^{\chi_{h}} \frac{V_{t}}{N_{t}}W_{t}$$

$$+ \frac{1}{1 + \rho}\psi^{w} \left(\Pi_{t+1}^{w} - 1\right) \frac{W_{t+1}}{W_{t}^{2}}W_{t+1} \frac{N_{t+1}}{N_{t}}.$$
(H.6)

Equation (H.6) can be rewritten as

$$\frac{\psi^{w}}{2} (\Pi_{t}^{w} \quad 1)^{2} + \psi^{w} (\Pi_{t}^{w} \quad 1) \Pi_{t}^{w} + 1 = W_{t} c_{v} \frac{(1 + \chi_{V}) \left(\frac{V_{t}}{N_{t-1}}\right)^{\chi_{V}}}{R_{t}(W_{t})} \left(R_{t}'(W_{t}) \frac{V_{t}}{N_{t}} \quad \frac{N_{t-1}}{N_{t}} S_{t}'(W_{t})\right) + W_{t} c_{h} (1 + \chi_{h}) \left(\frac{H_{t}}{N_{t-1}}\right)^{\chi_{h}} \left(-\frac{N_{t-1}}{N_{t}} S_{t}'(W_{t})\right) + \frac{1}{1 + \rho} \psi^{w} (\Pi_{t+1}^{w} \quad 1) (\Pi_{t+1}^{w})^{2} \frac{N_{t+1}}{N_{t}},$$

which lead to the following wage Phillips curve with hiring costs:

$$\frac{\psi^{w}}{2} (\Pi_{t}^{w} - 1)^{2} + \psi^{w} (\Pi_{t}^{w} - 1) \Pi_{t}^{w} + 1 = c_{v} (1 + \chi_{V}) \left(\frac{V_{t}}{N_{t-1}}\right)^{\chi_{V}} \left(\varepsilon_{R,W_{t}} \frac{V_{t}}{N_{t}} - \frac{N_{t-1}}{N_{t}} \frac{S_{t}(W_{t})}{R_{t}(W_{t})} \varepsilon_{S,W_{t}}\right) \\ + \underbrace{c_{h} (1 + \chi_{h}) \left(\frac{H_{t}}{N_{t-1}}\right)^{\chi_{h}} S_{t}(W_{t}) \frac{N_{t-1}}{N_{t}} (-\varepsilon_{S,W_{t}})}_{\text{New term}} \\ + \underbrace{\frac{1}{1 + \rho} \psi^{w} (\Pi_{t+1}^{w} - 1) (\Pi_{t+1}^{w})^{2} \frac{N_{t+1}}{N_{t}}}_{(H.7)}.$$

Two special cases In the case of convex vacancy costs, i.e., $\chi_v > 0$, and linear hiring costs, i.e., $\chi_h = 0$, the wage Phillips curve becomes

$$\frac{\psi^{w}}{2} (\Pi_{t}^{w} - 1)^{2} + \psi^{w} (\Pi_{t}^{w} - 1) \Pi_{t}^{w} + 1 = c_{v} (1 + \chi_{V}) \left(\frac{V_{t}}{N_{t-1}}\right)^{\chi_{V}} \left(\varepsilon_{R,W_{t}} \frac{V_{t}}{N_{t}} - \frac{N_{t-1}}{N_{t}} \frac{S_{t}(W_{t})}{R_{t}(W_{t})} \varepsilon_{S,W_{t}}\right) + c_{h} S_{t}(W_{t}) \frac{N_{t-1}}{N_{t}} (-\varepsilon_{S,W_{t}}) + \frac{1}{1 + \rho} \psi^{w} (\Pi_{t+1}^{w} - 1) (\Pi_{t+1}^{w})^{2} \frac{N_{t+1}}{N_{t}}.$$

If instead we have linear vacancy costs, i.e., $\chi_v = 0$ and convex hiring costs, i.e., $\chi_h > 0$, the wage Phillips curve is given by

$$\frac{\psi^{w}}{2} (\Pi_{t}^{w} - 1)^{2} + \psi^{w} (\Pi_{t}^{w} - 1) \Pi_{t}^{w} + 1 = c_{v} \left(\varepsilon_{R,W_{t}} \frac{V_{t}}{N_{t}} - \frac{N_{t-1}}{N_{t}} \frac{S_{t}(W_{t})}{R_{t}(W_{t})} \varepsilon_{S,W_{t}} \right) \\ + c_{h} (1 + \chi_{h}) \left(\frac{H_{t}}{N_{t-1}} \right)^{\chi_{h}} S_{t}(W_{t}) g_{t}^{-1} (-\varepsilon_{S,W_{t}}) \\ + \frac{1}{1 + \rho} \psi^{w} (\Pi_{t+1}^{w} - 1) (\Pi_{t+1}^{w})^{2} g_{t+1},$$

where we define $g_t \equiv \frac{N_t}{N_{t-1}}$ as employment growth.

No vacancy cost For simplicity, let us assume $c_v \to 0$, $c_h > 0$. With $B_t \equiv \frac{H_t}{N_{t-1}} = R_t T_t$ where $T_t \equiv \frac{V_t}{N_{t-1}}$ as defined in Appendix B.3, we can express equation (H.7) as

$$0 = F\left(\ln(\Pi_t^w), \ln(\Pi_{t+1}^w), \ln g_t, \ln g_{t+1}, \ln B_t, \ln \varepsilon_{S,W_t}, \ln S_t\right),$$

where

$$\begin{split} F_{\ln(\Pi_{t}^{w})} &= \psi^{w} \Pi_{t}^{w} \left(2(\Pi_{t}^{w} \quad 1) + \Pi_{t}^{w} \right) = \psi^{w} \\ F_{\ln(\Pi_{t+1}^{w})} &= \quad \frac{\psi^{w} g_{t+1}}{1+\rho} \left(\Pi_{t+1}^{w} (\Pi_{t+1}^{w})^{2} + (\Pi_{t+1}^{w} \quad 1) 2(\Pi_{t+1}^{w})^{2} \right) = \quad \frac{\psi^{w}}{1+\rho} \\ F_{\ln g_{t}} &= \quad c_{h} (1+\chi_{h}) B_{t}^{\chi_{h}} \cdot S_{t} \cdot (\quad g_{t}^{-1}) (\quad \varepsilon_{S,W_{t}}) = c_{h} (1+\chi_{h}) B^{\chi_{h}} \cdot S \cdot (\quad \varepsilon_{S}) \equiv \kappa_{h} > 0 \\ F_{\ln(g_{t+1})} &= \quad \frac{1}{1+\rho} \psi^{w} \left(\Pi_{t+1}^{w} \quad 1 \right) (\Pi_{t+1}^{w})^{2} g_{t+1} = 0 \\ F_{ln(B_{t})} &= \quad c_{h} (1+\chi_{h}) \chi_{h} \cdot B_{t}^{\chi_{h}} \cdot S_{t} \cdot g_{t}^{-1} (\quad \varepsilon_{S,W_{t}}) = \quad \chi_{h} \kappa_{h} \\ F_{\ln(\epsilon_{S,W_{t}})} &= c_{h} (1+\chi_{h}) B_{t}^{\chi_{h}} g_{t}^{-1} \epsilon_{S,W_{t}} = \quad \kappa_{h}, \quad F_{\ln(S_{t})} = c_{h} (1+\chi_{h}) B_{t}^{\chi_{h}} g_{t}^{-1} \epsilon_{S,W_{t}} = \quad \kappa_{h}. \end{split}$$

at the steady state. Therefore, up to a first order, we obtain

$$0 = \psi^{w} \check{\Pi}_{t}^{w} \quad \frac{\psi^{w}}{1+\rho} \check{\Pi}_{t+1}^{w} + \kappa_{h} \begin{pmatrix} \check{g}_{t} & \chi_{h} \check{B}_{t} & \check{\varepsilon}_{S,W_{t}} & \check{S}_{t} \end{pmatrix},$$

leading to the following linearized wage Phillips curve:

$$\check{\Pi}_{t}^{w} = \frac{1}{1+\rho} \check{\Pi}_{t+1}^{w} + \underbrace{\frac{\kappa_{h}}{\psi^{w}}}_{>0} \left(\check{S}_{t} + \check{\varepsilon}_{S,W_{t}} + \chi_{h}\check{B}_{t} \quad \check{g}_{t}\right). \tag{H.8}$$

Equation (H.8) is straightforward to interpret: higher separation \check{S}_t and more negative separation elasticity,²⁰ i.e., $\check{\varepsilon}_{S,W_t} > 0$, raise wage growth. With $\chi_h > 0$, i.e., convex hiring costs, higher \check{B}_t implies higher marginal costs of new hires, thereby incentivizing a firm to raise wages so that it does not want to lose its current employees. Finally, higher \check{g}_t

$$\check{\varepsilon}_{S,W_t} = \frac{\varepsilon_{S,W_t} \quad \varepsilon_S}{\varepsilon_S},$$

which implies that $\check{\varepsilon}_{S,W_t} > 0$ when ε_{S,W_t} is more negative than its steady state level ε_S .

²⁰In log-linearizing the non-linear wage Phillips curve (H.7), we use the following definition for negative ε_{S,W_t} :

(employment growth) means there is less incentive of firms to raise wages.

More Simplification To rearrange (H.8) so that it is represented in vacancy and unemployment, we use the followings we derived in Appendix **B**.3:

$$\check{S}_{t} = g_{S,V}\check{V}_{t} + g_{S,U}\check{U}_{t-1}
\check{\varepsilon}_{S,W_{t}} = g_{\varepsilon_{S},V}\check{V}_{t} + g_{\varepsilon_{S},U}\check{U}_{t-1},$$
(H.9)

and

$$\check{B}_{t} = \check{R}_{t} + \check{T}_{t} = \left(g_{R,V} + 1\right)\check{V}_{t} + \left(g_{R,U} + \frac{U}{1-U}\right)\check{U}_{t-1},\tag{H.10}$$

with

$$\check{g}_{t} = S(\check{R}_{t} + \check{T}_{t} \quad \check{S}_{t})
= S(g_{R,V} + 1 \quad g_{S,V})\check{V}_{t} + S\left(g_{R,U} + \frac{U}{1 \quad U} \quad g_{S,U}\right)\check{U}_{t-1}.$$
(H.11)

Based on equations (H.9), (H.10), and (H.11), equation (H.8) can be written as

$$\check{\Pi}_{t}^{w} = \frac{1}{1+\rho} \check{\Pi}_{t+1}^{w} + \phi_{V}^{H} \check{V}_{t} + \phi_{U}^{H} \check{U}_{t-1}, \tag{H.12}$$

where

$$\phi_V^H = \frac{\kappa_h}{\psi^w} \left[(g_{S,V} + g_{\varepsilon_S,V} + \chi_h (g_{R,V} + 1) \quad S(g_{R,V} + 1 \quad g_{S,V}) \right]$$

and

$$\phi_U^H = \frac{\kappa_h}{\psi^w} \left[g_{S,U} + g_{\varepsilon_S,U} + \chi_h \left(g_{R,U} + \frac{U}{1 - U} \right) \quad S \left(g_{R,U} + \frac{U}{1 - U} - g_{S,U} \right) \right].$$

Based on $\check{Q}_t = g_{Q,V}\check{V}_t + g_{Q,U}\check{U}_{t-1}$, equation (H.12) can be again re-written as

$$\check{\Pi}_{t}^{w} = \frac{1}{1+\rho}\check{\Pi}_{t+1}^{w} + \underbrace{\frac{\phi_{V}^{H}}{g_{Q,V}}}_{\equiv\beta_{Q}^{H}}\check{V}_{t} + \underbrace{\left(\phi_{U}^{H} - \phi_{V}^{H}\frac{g_{Q,U}}{g_{Q,V}}\right)}_{\equiv\beta_{U}^{H}}\check{U}_{t-1}$$
(H.13)

which is a function of quits \check{Q}_t and unemployment \check{U}_{t-1} . Under the current steady state levels unchanged, with parameters $c_h = 10$ and $\chi_h = 1$, we obtain

$$\check{\Pi}_{t}^{w} = \frac{1}{1+\rho} \check{\Pi}_{t+1}^{w} + 10^{-3} \left(\check{V}_{t} + 6.8 \times 10^{-2} \check{U}_{t-1}\right),$$

where we see $\phi_V^H = 6.8 \times 10^{-5} > 0$ and

$$\frac{\phi_V}{\phi_U} = 15.151.$$

The reason we have a positive coefficient $\phi_U > 0$ on unemployment is easy to understand: higher \check{U}_{t-1} (equivalently lower \check{N}_{t-1}) raises a marginal cost of new hire, inducing firms to raise wages to reduce turnover. This channel is absent under $\chi_h = 0$, in which case the wage Phillips curve becomes

$$\check{\Pi}_{t}^{w} = \frac{1}{1+\rho} \check{\Pi}_{t+1}^{w} + 10^{-3} \left(5.8 \check{V}_{t} - 1.4 \times 10^{-2} \check{U}_{t-1} \right),$$

which depends negatively on \check{U}_{t-1} . Still $\frac{|\phi_V|}{|\phi_U|} = 4.14$.

Quits and Unemployment In terms of quits and unemployment, with $\chi_h = 1$, at a monthly frequency we obtain

$$\check{\Pi}_t^w = \frac{1}{1+\rho} \check{\Pi}_{t+1}^w + 10^{-3} \left(1.9 \times \check{V}_t + 0.29 \check{U}_{t-1} \right),$$

where both $\beta_Q^H > 0$ and $\beta_U^H > 0$, and $\frac{\beta_Q^H}{\beta_U^H} = 6.57$.

Special case: no on-the-job search When $\lambda_{EE} \to 0$, we already know that $\check{S}_t \to 0$, $\check{\varepsilon}_{S,W_t} \to 0$, and

$$\check{R}_t \rightarrow \frac{\theta^2}{1+\theta^2}\check{\theta}_t = \frac{\theta^2}{1+\theta^2}(\check{V}_t \quad \check{U}_{t-1}).$$
(H.14)

We also know that

$$\check{B}_{t} = \check{R}_{t} + \check{T}_{t} = \underbrace{\check{R}_{t} + \check{V}_{t}}_{=\check{H}_{t}} + \frac{U}{1 - U} \check{U}_{t-1}$$
(H.15)

where

$$\check{H}_{t} = \check{R}_{t} + \check{V}_{t} = \frac{1}{1+\theta^{2}}\check{V}_{t} + \frac{\theta^{2}}{1+\theta^{2}}\check{U}_{t-1}.$$
(H.16)

Finally

$$\check{g}_t = S\left(\check{R}_t + \check{T}_t \quad \check{S}_t\right) \to S\check{B}_t = S\left(\check{H}_t + \frac{U}{1-U}\check{U}_{t-1}\right).$$
(H.17)

With equations (H.14), (H.15), (H.16), and (H.17), our linearized wage Phillips curve

(H.8) can be written as

$$\check{\Pi}_{t}^{w} = \frac{1}{1+\rho}\check{\Pi}_{t+1}^{w} + \underbrace{\frac{\kappa_{h}}{\psi^{w}}}_{>0} \left(\underbrace{\check{S}_{t}}_{\rightarrow 0} + \underbrace{\check{\varepsilon}_{S,W_{t}}}_{\rightarrow 0} + \chi_{h}\check{B}_{t} \quad \check{g}_{t} \right) \\
= \frac{1}{1+\rho}\check{\Pi}_{t+1}^{w} + \frac{\kappa_{h}}{\psi^{w}}(\chi_{h} \quad S)\check{B}_{t},$$
(H.18)

where the right hand side is written in \check{B}_t . When $\chi_h = 1$, since $\chi_h > S$ at the steady state, higher \check{B}_t raises wage growth since it increases the marginal cost of hiring a new worker due to the assumed convexity. In contrast, with $\chi_h = 0$, higher \check{B}_t actually reduces wage growth: under linear hiring costs, marginal costs of new hires become constant, and higher \check{B}_t means too many new hires, thereby inducing firms to reduce wages.

Appendix ILog-Linearized Wage Phillips Curve with Inflation-Indexed Unemployment Insurance b

In this section, we derive log-linearized wage Phillips curve in our model extension with fixed b in Section 4.2. We start from equation (B.19):

$$\begin{split} \check{\Pi}_{t}^{w} = & \frac{\kappa}{\psi^{w}} \left[\underbrace{\left(\underbrace{\varepsilon_{S} + S(\varepsilon_{R} \quad \varepsilon_{S})}_{>0} \right)}_{>0} \left(\check{S}_{t} \quad \check{R}_{t} \right) + \underbrace{\left(\varepsilon_{R} + \left(\chi \quad S\right)\left(\varepsilon_{R} \quad \varepsilon_{S}\right)\right)}_{>0} \check{T}_{t} + \left(\varepsilon_{R}\check{\varepsilon}_{R,t} \quad \varepsilon_{S}\check{\varepsilon}_{S,t}\right) \right] \\ & + \frac{1}{1+\rho} \check{\Pi}_{t+1}^{w}. \end{split}$$

Functional Forms with Real b Now we have

$$R_t = g(\theta_t) \left(\phi_{E,t} \frac{1}{2} + \phi_{U,t} \left(\frac{\tilde{w}_t^{\gamma}}{\tilde{w}_t^{\gamma} + b^{\gamma}} \right) \right).$$

where $\tilde{w}_t \equiv \frac{W_t}{P_t}$ is real wage, i.e., wage denominated in price aggregator P_t . At the steady state, we assume

$$\frac{\tilde{w}^{\gamma}}{\tilde{w}^{\gamma} + b^{\gamma}} = \frac{\xi^{\gamma}}{1 + \xi^{\gamma}} \equiv \mathcal{C}$$

Likewise, we obtain

$$S_{t} = s + (1 \quad s) \left(\lambda_{EE} f(\theta_{t}) \frac{1}{2} + \lambda_{EU} \left(\frac{b^{\gamma}}{\tilde{w}_{t}^{\gamma} + b^{\gamma}} \right) \right)$$
$$\varepsilon_{R,t} = \frac{\gamma \left(\frac{\phi_{E,t}}{4} + \phi_{U,t} \cdot \frac{b^{\gamma} \tilde{w}_{t}^{\gamma}}{\left[\tilde{w}_{t}^{\gamma} + b^{\gamma} \right]^{2}} \right)}{0.5 \phi_{E,t} + \left(\frac{\tilde{w}_{t}^{\gamma}}{\tilde{w}_{t}^{\gamma} + b^{\gamma}} \right) \phi_{U,t}}$$
$$\varepsilon_{S,t} = \frac{(1 \quad s)\gamma \left(f(\theta_{t}) \lambda_{EE} \frac{1}{4} + \lambda_{EU} \cdot \frac{b^{\gamma} \tilde{w}_{t}^{\gamma}}{\left[\tilde{w}_{t}^{\gamma} + b^{\gamma} \right]^{2}} \right)}{s + (1 \quad s) \left(0.5 \cdot \lambda_{EE} f(\theta_{t}) + \lambda_{EU} \cdot \frac{b^{\gamma}}{\tilde{w}_{t}^{\gamma} + b^{\gamma}} \right)}$$

Derivation To begin to simplify the wage Phillips curve with fixed b (B.19), we decompose all of the following right-hand-side variables, \check{S}_t , \check{T}_t , \check{R}_t , $\check{\varepsilon}_{R,t}$, and $\check{\varepsilon}_{S,t}$ into vacancy, unemployment, and the real wage \tilde{w}_t deviations. The tightness term, $T_t = \frac{V_t}{N_{t-1}}$, is the same: in log deviations from steady state, it becomes

$$\check{T}_t = \check{V}_t + \frac{U}{1 - U} \check{U}_{t-1}.$$

For the rest, we can write the decomposition as follows:

1. $\check{R}_t \equiv g_{R,V}\check{V}_t + g_{R,U}\check{U}_{t-1} + g_{R,w}\check{\tilde{w}}_t$

Derivation: Recall that the recruiting function is

$$R_t = g(\theta_t) \left(\phi_{E,t} \frac{1}{2} + \phi_{U,t} \left(\frac{\tilde{w}_t^{\gamma}}{\tilde{w}_t^{\gamma} + b^{\gamma}} \right) \right).$$

For practical purposes we define $C \equiv \frac{\xi^{\gamma}}{1+\xi^{\gamma}}$, which is increasing in the ratio of consumption for employed to unemployed workers. Then we obtain:

$$g_{R,V} = -\frac{\theta^2}{1+\theta^2},$$

and

$$g_{R,U} = \frac{\theta^2}{1+\theta^2} \cdot \frac{U(1-\lambda_{EE})}{\lambda_{EE}(1-U)+U} + \frac{0.5\phi_E}{0.5\phi_E + \mathcal{C}\phi_U} \cdot \frac{U}{1-U} \cdot \frac{\lambda_{EE}\phi_E - \lambda_{EE} - \phi_E}{\lambda_{EE}} + \frac{\mathcal{C}\phi_U}{0.5\phi_E + \mathcal{C}\phi_U} \cdot (1-\phi_U(1-\lambda_{EE})),$$

and

$$g_{R,w} = \frac{\mathcal{C}\phi_U}{0.5\phi_E + \mathcal{C}\phi_U} \cdot \gamma(1 \quad \mathcal{C}) > 0.$$

Note that $|g_{R,w}|$ is decreasing in λ_{EE} .

2. $\check{S}_t \equiv g_{S,V}\check{V}_t + g_{S,U}\check{U}_{t-1} + g_{S,w}\check{\tilde{w}}_t$

Derivation: Recall that the separation function S_t is given by

$$S_t = s + \begin{pmatrix} 1 & s \end{pmatrix} \left(\lambda_{EE} f(\theta_t) \frac{1}{2} + \lambda_{EU} \left(\frac{b^{\gamma}}{\tilde{w}_t^{\gamma} + b^{\gamma}} \right) \right)$$

Then, we obtain:

$$g_{S,V} = \frac{(1 \quad s)0.5\lambda_{EE}f}{s + (1 \quad s)(0.5\lambda_{EE}f + \lambda_{EU}(1 \quad \mathcal{C}))} \cdot \frac{1}{1 + \theta^2},$$

and

$$g_{S,U} = \frac{(1 \quad s)0.5\lambda_{EE}f}{s + (1 \quad s)\left(0.5\lambda_{EE}f + \lambda_{EU}(1 \quad \mathcal{C})\right)} \cdot \frac{1}{1 + \theta^2} \cdot \frac{U(1 \quad \lambda_{EE})}{\lambda_{EE}(1 \quad U) + U},$$

and

$$g_{S,w} = \frac{\gamma(1 \quad s)\lambda_{EU}\mathcal{C}(1 \quad \mathcal{C})}{s + (1 \quad s)\left(0.5\lambda_{EE}f + \lambda_{EU}(1 \quad \mathcal{C})\right)} < 0.$$

Note that $|g_{S,w}|$ is decreasing in λ_{EE} , given θ .

3. $\check{\varepsilon}_{R,t} = g_{\varepsilon_{R,U}} \check{U}_{t-1} + g_{\varepsilon_{R,w}} \check{\tilde{w}}_{t}$

Derivation: Note that in equilibrium, $\varepsilon_{R,t}$ is given by

$$\varepsilon_{R,t} = \frac{\gamma \left(\frac{\phi_{E,t}}{4} + \phi_{U,t} \cdot \frac{b^{\gamma} \tilde{w}_{t}^{\gamma}}{\left[\tilde{w}_{t}^{\gamma} + b^{\gamma}\right]^{2}}\right)}{\left(0.5\phi_{E,t} + \left(\frac{\tilde{w}_{t}^{\gamma}}{\tilde{w}_{t}^{\gamma} + b^{\gamma}}\right)\phi_{U,t}\right)}$$

from which we obtain

$$g_{\varepsilon_{R,U}} = \begin{pmatrix} 0.25\phi_E & 0.5\phi_E \\ 0.25\phi_E + \mathcal{C}(1-\mathcal{C})\phi_U & 0.5\phi_E + \mathcal{C}\phi_U \end{pmatrix} \frac{U}{1-U} \frac{\lambda_{EE}\phi_E & \lambda_{EE} & \phi_E}{\lambda_{EE}} \\ + \left(\frac{\mathcal{C}(1-\mathcal{C})\phi_U}{0.25\phi_E + \mathcal{C}(1-\mathcal{C})\phi_U} & \frac{\mathcal{C}\phi_U}{0.5\phi_E + \mathcal{C}\phi_U} \right) (1-\phi_U(1-\lambda_{EE})).$$

and

$$g_{\varepsilon_{R},w} = \left[\frac{\phi_{U}\mathcal{C}(1 \ \mathcal{C})}{0.25\phi_{E} + \phi_{U}\mathcal{C}(1 \ \mathcal{C})}\gamma(2\mathcal{C} \ 1) + \frac{\mathcal{C}\phi_{U}}{0.5\phi_{E} + \mathcal{C}\phi_{U}}\gamma(1 \ \mathcal{C})\right] < 0$$

Note that $|g_{\varepsilon_R,w}|$ is decreasing in λ_{EE} .

4. $\check{\varepsilon}_{S,t} = g_{\varepsilon_{S,V}}\check{V}_t + g_{\varepsilon_{S,U}}\check{U}_{t-1} + g_{\varepsilon_S,w}\check{\tilde{w}}_t$

Derivation: Note that in equilibrium $\varepsilon_{S,t}$ is given by

$$\varepsilon_{S,t} = \frac{(1 \quad s)\gamma \left(f(\theta_t)\lambda_{EE}\frac{1}{4} + \lambda_{EU} \cdot \frac{b^{\gamma}\tilde{w}_t^{\gamma}}{\left[\tilde{w}_t^{\gamma} + b^{\gamma}\right]^2} \right)}{s + (1 \quad s) \left(0.5 \cdot \lambda_{EE}f(\theta_t) + \lambda_{EU} \cdot \frac{b^{\gamma}}{\tilde{w}_t^{\gamma} + b^{\gamma}} \right)}$$

from which we obtain

$$g_{\varepsilon_{S,V}} = \left(\frac{0.25\lambda_{EE}f}{0.25\lambda_{EE}f + \mathcal{C}(1-\mathcal{C})\lambda_{EU}} - \frac{0.5(1-s)\lambda_{EE}f}{s + (1-s)(0.5\lambda_{EE}f + (1-\mathcal{C})\lambda_{EU})}\right)\frac{1}{1+\theta^2},$$

and

$$g_{\varepsilon_{S,U}} = g_{\varepsilon_{S,V}} \cdot \frac{U(1 \quad \lambda_{EE})}{\lambda_{EE}(1 \quad U) + U}$$

and

$$g_{\varepsilon_{s,w}} = \gamma \left[\mathcal{C} \cdot \frac{(1 \ s)\lambda_{EU}(1 \ \mathcal{C})}{s + (1 \ s)(0.5\lambda_{EE}f + \lambda_{EU}(1 \ \mathcal{C}))} \quad (2\mathcal{C} \quad 1) \frac{\lambda_{EU}\mathcal{C}(1 \ \mathcal{C})}{0.25\lambda_{EE}f + \lambda_{EU}\mathcal{C}(1 \ \mathcal{C})} \right]$$

Note that $g_{\varepsilon_S,w}$ is a complex function of λ_{EE} . Since $\mathcal{C} \simeq 1$ and $s \simeq 0$ under our calibration, we have $2\mathcal{C} = 1 \simeq 1$ and:

$$g_{\varepsilon_{s},w} \simeq \gamma \left[\frac{\lambda_{EU}(1 \ C)}{0.5\lambda_{EE}f + \lambda_{EU}(1 \ C)} \quad \frac{\lambda_{EU}(1 \ C)}{0.25\lambda_{EE}f + \lambda_{EU}(1 \ C)} \right]$$

$$= \gamma \left(\frac{0.25\lambda_{EE}f}{0.25\lambda_{EE}f + \lambda_{EU}(1 \ C)} \right) \cdot \left(\frac{\lambda_{EU}(1 \ C)}{0.5\lambda_{EE}f + \lambda_{EU}(1 \ C)} \right)$$

$$\simeq \gamma \lambda_{EU}(1 \ C) \frac{0.25\lambda_{EE}f^{2}}{0.125\lambda_{EE}^{2}f^{2} + \lambda_{EU}(1 \ C) \cdot 0.75\lambda_{EE}f + \frac{\lambda_{EU}^{2}(1 \ C)^{2}}{20}}{20}$$

$$\simeq \gamma \lambda_{EU}(1 \ C) \frac{0.25 \cdot f}{0.125\lambda_{EE}f^{2} + \lambda_{EU}(1 \ C) \cdot 0.75f}$$

Therefore, $|g_{\varepsilon_S,w}|$ is (approximately) decreasing in λ_{EE} .

Decomposing Wage Growth into Vacancies and Unemployment Combining these results, we can rewrite the wage Phillips curve just in terms of gaps in vacancies, unemployment, and real wage. Let

$$\Delta_1 \equiv \varepsilon_S + S(\varepsilon_R \quad \varepsilon_S)$$

and

$$\Lambda_1 \equiv \varepsilon_R + (\chi \quad S)(\varepsilon_R \quad \varepsilon_S)$$

as in Appendix B.2.

Then the wage Phillips curve (B.19) can be written as:

$$\begin{split} \check{\Pi}_{t}^{w} &= \frac{\kappa}{\underline{\psi}^{w}} \begin{bmatrix} \Lambda_{1} + \Delta_{1} \left(g_{S,V} \quad g_{R,V}\right) & \varepsilon_{S} g_{\varepsilon_{S},V} \end{bmatrix} \check{V}_{t} \\ &= \overline{\phi_{V} > 0} \\ &+ \underbrace{\frac{\kappa}{\underline{\psi}^{w}} \begin{bmatrix} \frac{U}{1 \quad U} \Lambda_{1} + \Delta_{1} \left(g_{S,U} \quad g_{R,U}\right) + \varepsilon_{R} g_{\varepsilon_{R},U} & \varepsilon_{S} g_{\varepsilon_{S},U} \end{bmatrix}}_{\equiv \phi_{U} < 0} \\ &+ \underbrace{\frac{\kappa}{\underline{\psi}^{w}} \begin{bmatrix} \Delta_{1} \left(\underbrace{g_{S,w}}_{<0} & \underbrace{g_{R,w}}_{>0}\right) + \underbrace{\varepsilon_{R}}_{>0} & \underbrace{g_{\varepsilon_{R},w}}_{<0} & \underbrace{\varepsilon_{S}}_{<0} & \underbrace{g_{\varepsilon_{S},w}}_{<0} \end{bmatrix}}_{\equiv \phi_{w} < 0} \\ \underbrace{\check{w}_{t}}_{\equiv \phi_{w} < 0} \end{split}$$
(I.1)

and $|\phi_w|$ is decreasing in on-the-job search probability λ_{EE} , given steady-state θ . Interpretation is very simple: given \check{V}_t and \check{U}_{t-1} , a cost of living shock lowers real wage $\check{\tilde{w}}_t$, and with $\phi_w < 0$ raises wage inflation $\check{\Pi}_t^w$. Therefore, there is pass-through from price to wage as we explain in Section 4.2.

I.1 Wage Phillips Curve in Quits, Unemployment, and Real Wage

First, we log-linearize the quit function Q_t as $\check{Q}_t = g_{Q,V}\check{V}_t + g_{Q,U}\check{U}_{t-1} + g_{Q,w}\check{\tilde{w}}_t$. Recall that the quit function $Q_t = S_t$ s is given by

$$Q_t = \begin{pmatrix} 1 & s \end{pmatrix} \left(\lambda_{EE} f(\theta_t) \frac{1}{2} + \lambda_{EU} \left(\frac{b^{\gamma}}{\tilde{w}_t^{\gamma} + b^{\gamma}} \right) \right)$$

Then, we obtain:

$$g_{Q,V} = \frac{0.5\lambda_{EE}f}{0.5\lambda_{EE}f + \lambda_{EU}(1-\mathcal{C})} \cdot \frac{1}{1+\theta^2} > 0,$$

and

$$g_{Q,U} = \frac{0.5\lambda_{EE}f}{0.5\lambda_{EE}f + \lambda_{EU}(1-\mathcal{C})} \cdot \frac{1}{1+\theta^2} \cdot \frac{U(1-\lambda_{EE})}{\lambda_{EE}(1-U) + U} < 0, \quad g_{Q,w} = \frac{\gamma\lambda_{EU}\mathcal{C}(1-\mathcal{C})}{0.5\lambda_{EE}f + \lambda_{EU}(1-\mathcal{C})} < 0$$

Rearranging in terms of

$$\check{V}_t = \frac{\check{Q}_t \quad g_{Q,U}\check{U}_{t-1} \quad g_{Q,w}\check{\tilde{w}}_t}{g_{Q,V}},$$

the above equation (I.1) becomes

$$\begin{split} \check{\Pi}_{t}^{w} &= \phi_{V}\check{V}_{t} + \phi_{U}\check{U}_{t-1} + \phi_{w}\check{\tilde{w}}_{t} + \frac{1}{1+\rho}\check{\Pi}_{t+1}^{w} \\ &= \phi_{V}\left(\frac{\check{Q}_{t} - g_{Q,U}\check{U}_{t-1} - g_{Q,w}\check{\tilde{w}}_{t}}{g_{Q,V}}\right) + \phi_{U}\check{U}_{t-1} + \phi_{w}\check{\tilde{w}}_{t} + \frac{1}{1+\rho}\check{\Pi}_{t+1}^{w} \\ &= \underbrace{\frac{\phi_{V}}{g_{Q,V}}\check{Q}_{t}}_{>0} + \left(\underbrace{\frac{g_{Q,U}}{g_{Q,V}}\phi_{V} + \underbrace{\phi_{U}}_{<0}}_{>0}\right)\check{U}_{t-1} + \left(\underbrace{\frac{g_{Q,w}}{g_{Q,V}}\phi_{V} + \underbrace{\phi_{w}}_{<0}}_{>0}\right)\check{\tilde{w}}_{t} + \frac{1}{1+\rho}\check{\Pi}_{t+1}^{w}. \end{split}$$
(1.2)

We should then expect that the "catch-up" (see e.g., Bernanke and Blanchard (2024)) term $\check{\tilde{w}}_t$ should be mitigated for the quits equation (I.2) than for the equation with vacancies (I.1). The reason is that some effect of this catch-up is absorbed by an endogenous response in quits, i.e., quits \check{Q}_t will rise with a cost of living shock, given \check{V}_t , \check{U}_{t-1} , and $\check{\tilde{w}}_t$. If those effects are strong enough, the coefficient on $\check{\tilde{w}}_t$ in equation (I.2) can be positive.

It turns out that the coefficient on $\check{\tilde{w}}_t$ in equation (I.2) becomes positive under our calibration. At monthly frequency, we obtain

$$\begin{split} \check{\Pi}_{t}^{w} = & 0.0183\check{V}_{t} \quad 0.003\check{U}_{t-1} \underbrace{0.011}_{<0} \check{\tilde{w}}_{t} + \frac{1}{1+\rho} \check{\Pi}_{t+1}^{w} \\ = & 0.0246\check{Q}_{t} \quad 0.0009\check{U}_{t-1} \underbrace{+0.0142}_{>0} \check{\tilde{w}}_{t} + \frac{1}{1+\rho} \check{\Pi}_{t+1}^{w}, \end{split}$$

where still quits remains as the strongest driver of wage growth.